## University of Wisconsin Milwaukee UWM Digital Commons

Theses and Dissertations

August 2013

# The Boundedness of Hausdorff Operators on Function Spaces

Xiaoying Lin University of Wisconsin-Milwaukee

Follow this and additional works at: https://dc.uwm.edu/etd Part of the Mathematics Commons

#### **Recommended** Citation

Lin, Xiaoying, "The Boundedness of Hausdorff Operators on Function Spaces" (2013). *Theses and Dissertations*. 256. https://dc.uwm.edu/etd/256

This Dissertation is brought to you for free and open access by UWM Digital Commons. It has been accepted for inclusion in Theses and Dissertations by an authorized administrator of UWM Digital Commons. For more information, please contact open-access@uwm.edu.

## THE BOUNDEDNESS OF HAUSDORFF OPERATORS ON FUNCTION SPACES

by

Xiaoying Lin

A Dissertation Submitted in Partial Fulfillment of the Requirements for the Degree of

> Doctor of Philosophy in Mathematics

> > $\operatorname{at}$

The University of Wisconsin–Milwaukee August 2013

### ABSTRACT THE BOUNDEDNESS OF HAUSDORFF OPERATORS ON FUNCTION SPACES

by

Xiaoying Lin

The University of Wisconsin–Milwaukee, 2013 Under the Supervision of Professor Dr. Dashan Fan

For a fixed kernel function  $\Phi$ , the one dimensional Hausdorff operator is defined in the integral form by

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(t)}{t} f(\frac{x}{t}) dt.$$

By the Minkowski inequality, it is easy to check that the Hausdorff operator is bounded on the Lebesgue spaces  $L^p$  when  $p \ge 1$ , with some size condition assumed on the kernel functions  $\Phi$ . However, people discovered that the above boundedness property is quite different on the Hardy space  $H^p$  when 0 . To establish theboundedness on the Hardy space for <math>0 , some smoothness must be assumed $on the kernel functions <math>\Phi$ .

In this thesis, we first study the boundedness of  $h_{\Phi}$  on the Hardy space  $H^1$ , and on the local Hardy space  $h^1(\mathbb{R})$ . Our work shows that for  $\Phi(t) \ge 0$ , the Hausdorff operator  $h_{\Phi}$  is bounded on the Hardy space  $H^1$  if and only if  $\Phi$  is a Lebesgue integrable function; and  $h_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R})$  if and only if the functions  $\Phi(t)\chi_{(1,\infty)}(t)$  and  $\Phi(t)\chi_{(0,1)(t)}\log(\frac{1}{t})$  are Lebesgue integrable. These results solve an open question posed by the Israeli mathematician Liflyand. We also establish an  $H^1(\mathbb{R}) \to H^{1,\infty}(\mathbb{R})$  boundedness theorem for  $h_{\Phi}$ . As applications, we obtain many decent properties for the Hardy operator and the *k*th order Hardy operators. For instance, we know that the Hardy operator  $\mathcal{H}$  is bounded from  $H^1(\mathbb{R}) \to H^{1,\infty}(\mathbb{R})$ , bounded on the atomic space  $H^1_A(\mathbb{R}_+)$ , but it is not bounded on both  $H^1(\mathbb{R})$  and the local Hardy space  $h^1(\mathbb{R})$ .

We also extend part of these results to the high dimensional Hausdorff operators. Here, we study two high dimensional extentions on the Hausdorff operator  $h_{\Phi}$ :

$$\tilde{H}_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^{n-\beta}} f(\frac{x}{|y|}) \, dy, \quad n \ge \beta \ge 0,$$

and

$$H_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(\frac{x}{|y|})}{|y|^{n-\beta}} f(y) \, dy, \quad n \ge \beta \ge 0,$$

where  $\Phi$  is a local integrable function.

For  $0 , we obtain a sufficient condition for the <math>H^p$  boundedness for the Hausdorff operator in the one dimensional case. This theorem needs less smoothness on the kernel  $\Phi$  than any other theorems in the literature. Since there is no result involving the boundedness on  $H^p(\mathbb{R}^n)$  in the literature for the high dimensional Hausdorff operators, if  $0 and <math>n \ge 2$ , it is interesting to study such problems in the high dimensional spaces. We establish several sufficient conditions by using a duality argument.

Additionally, we study boundedness of Hausdorff operators on some Herz type spaces, and some bilinear Hausdorff operators and fractional Hausdorff operators.

©Copyright by Xiaoying Lin, 2013 All Rights Reserved

## CONTENTS

1	Intr	oduction	1
<b>2</b>	Preliminaries 1		16
	2.1	Lebesgue and Hardy Spaces	16
		2.1.1 Lorentz Spaces	16
		2.1.2 Hardy Spaces	17
	2.2	Duality	18
		2.2.1 Lipschitz Spaces	18
		2.2.2 BMO	19
	2.3	Alternate Characterizations of $H^p$	19
		2.3.1 Atomic Characterization	19
		2.3.2 Hilbert and Riesz Transform	21
	2.4	Interpolation	23
		2.4.1 Stein-Weiss Analytic Interpolation	23
		2.4.2 Marcinkiewisz Interpolation	25
	2.5	Herz Type Spaces	26
3	One	Dimensional Hausdorff Operators: The Case $0$	<b>28</b>
4	One	Dimensional Hausdorff Operators: The Case $p = 1$	33
_	4.1	Introduction	33
		4.1.1 The Hardy Operator and its Generalizations	33
		4.1.2 A Lemma in Local Hardy Spaces	34
	4.2	Necessary and Sufficient Conditions for $H^1$ Boundedness	36
	4.3	Boundedness of the Hardy Operator Near $H^1$	41
	4.4	Generalizations of the Hardy Operator $(0 $	45
<b>5</b>	Hig	h Dimensional Hausdorff Operators	51
	5.1	$H^1(\mathbb{R}^n) \to H^1(\mathbb{R}^n)$ Boundedness of $\tilde{H}_{\Phi}$	53
	5.2	Boundedness on Lebesgue spaces	55
	5.3	Boundedness on $H^p(\mathbb{R}^n)$	58
		5.3.1 $H^p \to L^q$ Boundedness of $H_{\Phi,\beta}$	58
		5.3.2 $H^p \to H^q$ Boundedness of $H_{\Phi,\beta}$	61
6	Additional Spaces and Operators		75
Ū	6.1	Motivation	75
	6.2	Boundedness of Hausdorff Operator on $HK_q^{\alpha,p}(\mathbb{R}^n)$	79
	6.3	$H\dot{K}^{\alpha,p}_q(\mathbb{R}) \to H^r(\mathbb{R})$ Boundedness	83
	6.4	$\operatorname{Bilinear Hausdorff Operators}  \ldots  \ldots  \ldots  \ldots  \ldots  \ldots  \ldots  \ldots  \ldots  $	87
References			90

#### ACKNOWLEDGMENTS

As I complete my thesis, I would like to take the opportunity to express my gratitude to my advisor, Dr. Fan, and also to Dr. Sun, for their unselfish help in my academic career during my graduate study. None of my achievements would have been possible without their help. I would like to thank all the members of my committee for their attention and support, especially Dr. Volkmer, who has provided many helpful suggestions for clarifying my results. I would also like to thank Dr. ChunJie Zhang, who helped to introduce me to the topic of this thesis.

I also want to take the chance to thank my friend and classmate, America , for her help in proofreading this thesis.

Without the people I have mentioned, and others that I have doubtless omitted, I would never have gotten as far as I have. I will repay their kindness by passing it on to my own students, fostering my students' development as mathematicians and as people, as others have fostered mine.

## CHAPTER 1

#### INTRODUCTION

One of the core problems in harmonic analysis is to study the boundedness of an operator T on some function/distribution spaces

$$\left\|Tf\right\|_{Y} \leqslant \left\|f\right\|_{X},$$

where X and Y are two function/distribution spaces with norms or quasi norms  $\|\cdot\|_X$ ,  $\|\cdot\|_Y$  respectively. This question arises from many natural problems in mathematics and the sciences. To illustrate the importance of this problem, we look at the following two examples.

**Example 1.1.** Any  $L^1$  function f on  $[-\pi,\pi]$  has its Fourier series

$$f(x) \sim \sum_{k} c_k e^{ikx}.$$

This means a signal f might be built up from many simple sine and cosine waves with different wave lengths and amplitude. Unfortunately,

$$f(x) = \sum_{k} c_k e^{ikx}$$

is not always true pointwise, even when f is a continuous function(see [31]). So we need to modify the information using a filter  $\{m(\epsilon k)\}$  (called the summation of the Fourier series):

$$T_{m,\epsilon}f(x) = \sum_{k} c_k m(\epsilon k) e^{ikx}, \quad \epsilon > 0,$$

where m is a suitably nice function and m(0) = 1.  $T_m$  is called a multiplier (operator) with symbol m. This summation greatly improves the convergence of the Fourier series

2

in the sense that

$$\lim_{\epsilon \to 0+} \sum_{k} c_k m(\epsilon, k) e^{ikx} = f(x)$$

uniformly for a continuous function f if one chooses a suitable function m. For instance, from [31], we know that  $T_{m,\epsilon}(f)$  is the Abel summation if  $m(\sqrt{\epsilon}k) = e^{-\epsilon|k|^2}$ , and the Riesz summation if  $m(\epsilon k) = (1 - \epsilon |k|)_+$ .

Furthermore, let F be some function/distribution space with quasi-norm  $\|\cdot\|_F$ , we are interested in the global convergence

$$\lim_{\epsilon \to 0+} \|T_{m,\epsilon}g - g\|_F = 0, \text{ for } g \in F.$$

We suppose that the class S of Schwartz functions is dense in F, and that for  $f \in S$ ,

 $\lim_{\epsilon \to 0+} \left\| T_{m,\epsilon} f - f \right\|_F = 0.$ 

Then it is easy to check that the boundedness

$$|T_{m,\epsilon}g||_F \leq ||g||_F$$
, for  $\epsilon$  sufficiently small

implies that for all  $g \in F$ ,

$$\lim_{\epsilon \to 0+} \|T_{m,\epsilon}g - g\|_F = 0.$$

To see this fact, for given  $g \in F$ , and any  $\delta > 0$ , choose  $f \in S$  such that

$$\|f - g\|_F < \delta.$$

$$\begin{split} \lim_{\epsilon \to 0+} \|T_{m,\epsilon}g - g\|_F &\leq \lim_{\epsilon \to 0+} \left( \|f - g\|_F + \|T_{m,\epsilon}(f - g)\|_F + \|T_{m,\epsilon}f - f\|_F \right) \\ &\leq \delta + \lim_{\epsilon \to 0+} \|T_{m,\epsilon}f - f\|_F \\ &\leq \delta. \end{split}$$

Thus the convergence problem  $T_{m,\epsilon}f \to f$  in the space F is reduced to the boundedness

of  $T_{m,\epsilon}$  on the space F.

**Example 1.2.** The solution u(t, x) of the Cauchy problem of the Schrödinger equation

$$\begin{cases} i\partial_t u - \Delta u = 0, \quad (t, x) \in \mathbb{R}_+ \times \mathbb{R}^n \\ u(0, x) = u_0(x) \end{cases}$$

is formally given by  $u(t,x) = (e^{-it\Delta}u_0)(x)$ , where  $u(t,x) = (e^{-it\Delta}u_0)$  is defined through its Fourier transform by

$$(\widehat{e^{-it\Delta}u_0})(\xi) = e^{it|\xi|^2}\widehat{u_0}(\xi).$$

Let X and Y be two function spaces. To study the regularity of the solution, we need to estimate

$$\|e^{-it\Delta}(u_0-v_0)\|_Y = \|e^{-it\Delta}u_0-e^{-it\Delta}v_0\|_Y \leq \|u_0-v_0\|_X,$$

since the operator  $e^{-it\Delta}$  is linear. Again we face the boundedness inequality

$$\left\|e^{-it\Delta}f\right\|_{Y} \leqslant \|f\|_{X} \,.$$

These examples begin to show the importance of demonstrating the boundedness of operators on function spaces. In this thesis, we mainly study the boundedness of the Hausdorff operators on the Lebesgue spaces and on the Hardy spaces. We begin by recalling the one dimensional Hausdorff operator, defined in the integral form by

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(t)}{t} f(\frac{x}{t}) dt, \qquad (1.1)$$

where, for simplicity, we initially define  $h_{\Phi}$  on the Schwartz space S. Clearly,  $h_{\Phi}$  is a linear operator. This integral operator is deeply rooted in the study of 1-dimensional Fourier analysis. Particularly, it is closely related to the summability of the classical Fourier series (see [13]).

The definition of  $h_{\Phi}$  is based on the dilation structure of the Euclidean space. Two important geometric transformations on the Euclidean space  $\mathbb{R}^n$  are translation and dilation. The translation  $L_y$  is a linear operator defined by  $L_y(f)(x) = f(x+y)$ , so its integral form

$$\int \Phi(y) L_{-y}(f)(x) \, dy = \int \Phi(y) f(x-y) \, dy$$
 (1.2)

is the convolution operator  $\Phi * f(x)$  with kernel  $\Phi$ , and its  $L^1$  norm satisfies

$$\|\Phi * f\|_{L^1} = \|\Phi\|_{L^1} \|f\|_{L^1}, \quad \text{if} \quad \Phi, f \ge 0.$$

The dilation  $D_t$  is a linear operator defined by  $D_t(f)(x) = f(tx)$ , for t > 0. The integral

$$\int \Phi(t) D_{t^{-1}}(f)(x) \, dt = \int \Phi(t) f(\frac{x}{t}) \, dt \tag{1.3}$$

corresponds to the Hausdorff operator

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(t)}{t} D_{t^{-1}}(f)(x) \, dt.$$
(1.4)

Note that  $\Phi(t)$  in equation (1.3) is replaced by  $\frac{\Phi(t)}{t}$  in equation (1.4). This normalizes the operator, so that

$$\|h_{\Phi}(f)\|_{L^{1}} = \|\Phi\|_{L^{1}} \|f\|_{L^{1}}, \text{ if } \Phi, f \ge 0.$$

Many important operators in real and complex analysis are special cases of the Hausdorff operator, by taking suitable choice of  $\Phi$ . These operators include, among many others:

(1) the Hardy operator

$$\mathcal{H}f(x) = \frac{1}{x} \int_0^x f(t) \, dt,$$

the average of antiderivative of a function f in the Fundamental Theorem of Calculus, is obtained by choosing  $\Phi(t) = \frac{\chi_{(1,\infty)}(t)}{t}$ ;

(2) the adjoint Hardy operator

$$\mathcal{H}^*f(x) = \int_x^\infty \frac{f(t)}{t} \, dt,$$

which is obtained by choosing  $\Phi(t) = \chi_{(0,1)}(t)$ ;

(3) the Cesàro operator

$$\mathcal{C}_{\alpha}(f)(x) = \alpha \int_0^1 \frac{(1-t)^{\alpha-1}}{t} f(\frac{x}{t}) dt,$$

which is obtained by choosing  $\Phi(t) = \chi_{(0,1)}(t)(1-t)^{\alpha-1}$ ;

(4) the Hardy-Littlewood-Pólya operator

$$P(f)(x) = \int_0^\infty \frac{f(t)}{\max\{t, x\}} \, dt = Hf(x) + H^*f(x).$$

Also, the Riemann-Liouville fractional derivatives

$$\tilde{D}_{\beta}(f)(x) = \int_{0}^{x} (x-t)^{\beta-1} f(t) \, dt, \quad x > 0$$

can be derived from the fractional Hausdorff operator. In fact

$$\tilde{D}_{\beta}(f)(x) = x^{\beta-1} \int_0^x \left(1 - \frac{t}{x}\right)^{\beta-1} f(t) dt$$
$$\cong x^{\beta} \int_1^\infty \frac{(1 - \frac{1}{t})^{\beta-1}}{t^2} f(\frac{x}{t}) dt.$$

Thus,

$$\tilde{D}_{\beta}(f)(x) \cong x^{\beta}h_{\Phi}(f)(x)$$

with

$$\Phi(t) = \frac{(1 - \frac{1}{t})^{\beta - 1}}{t} \chi_{(1,\infty)}(t).$$

The Hausdorff operator has received extensive study in recent years, particularly its boundedness on the Lebesgue space  $L^p$  and the Hardy space  $H^p$  (see [9, 14, 18, 28, 30]).

Consider a quasi-normed space X with quasi-norm  $\|\cdot\|_X.$  We say that X satisfies

the norm scaling (N-S) property if there exists a number  $\sigma = \sigma(X)$  such that

$$\left\|f(\frac{\cdot}{t})\right\|_{X} = t^{\sigma} \, \|f\|_{X}$$

for any  $f \in X, t > 0$ . If X satisfies the N-S property, then by Minkowski's inequality

$$\begin{aligned} \|h_{\Phi}f\|_{X} &\leq \int_{0}^{\infty} \frac{|\Phi(t)|}{t} \left\| f(\frac{\cdot}{t}) \right\|_{X} dt \\ &= \left( \int_{0}^{\infty} \frac{|\Phi(t)|}{t} t^{\sigma} dt \right) \|f\|_{X} \end{aligned}$$

This shows that  $h_{\Phi}$  is bounded on the space X if

$$\left(\int_0^\infty \frac{|\Phi(t)|}{t} t^\sigma dt\right) < \infty.$$
(1.5)

Based on these observations, our main interests in the Hausdorff operator are:

- Determine when the condition given in equation (1.5) is sharp, that is, for which spaces X satisfying the N-S property the condition in equation (1.5) is necessary as well as sufficient.
- (2) Study the boundedness of  $h_{\Phi}: X \to X$  if X does not satisfy the N-S property.

The Lebesgue spaces  $L^p$  when  $p \ge 1$  satisfy the N-S property. Thus, by the above argument, we have

$$\|h_{\Phi}f\|_{L^{p}} \leq C_{\Phi} \|f\|_{L^{p}},$$

where  $C_{\Phi}$  is the constant

$$C_{\Phi} = \left( \int_0^\infty \frac{|\Phi(t)|}{t} t^{\frac{1}{p}} dt \right).$$

It is known that the Hausdorff operator is bounded on the Lebesgue spaces  $L^p$  when  $p \ge 1$  if and only if  $C_{\Phi} < \infty$ , provided  $\Phi(t)$  is a non-negative valued function. The real Hardy space  $H^1$  is also a space with the N-S property. Again, by a scaling argument together with the Minkowski inequality, we obtain

$$\|h_{\Phi}f\|_{H^1} \leq \|f\|_{H^1}$$

if  $\Phi(t)$  is a Lebesgue integrable function on  $(0, \infty)$ . The Israeli mathematician Liflyand posed the following question.

Question 1 (Liflyand [13]). Determine the sharpness of the condition  $\Phi \in L^1(0, \infty)$ to give  $h_{\Phi}$  bounded on  $H^1(\mathbb{R})$ .

In this thesis, we will solve this problem by showing that if  $\Phi(t)$  is a non-negative valued function, then

$$\|h_{\Phi}f\|_{H^1} \leq \|f\|_{H^1}$$

if and only if  $\Phi(t)$  is a Lebesgue integrable function on  $(0, \infty)$ . We will also solve the same problem on the local Hardy space  $h^1$ . As applications, we obtain some interesting results for the Hardy operator on the spaces near  $H^1$ .

The real Hardy space  $H^p$  is not a normed space when 0 . It is known that $the above boundedness property on <math>L^p$  or  $H^1$  is quite different on the Hardy space  $H^p$  when  $0 . To establish the boundedness of <math>h_{\Phi}$  on the real Hardy space  $H^p$ for all  $0 , it seems that any reasonable size condition on <math>\Phi$  is not sufficient; some smoothness condition must be included. This phenomeonon was discovered by Kinjin [9] who required a smoothness condition on  $\hat{\Phi}$ , the Fourier transforms of  $\Phi$ . It was further explored by Liflyand and Miyachi in [14], who required a smoothness condition on  $\Phi$ , and who further found a bounded function  $\Phi$  supported in a compact set  $E \subset (0, \infty)$ , such that the operator  $h_{\Phi}$  is not bounded on  $H^p(\mathbb{R})$  for any 0 .

The original work of Kanjin,Liflyand and Miyachi was motivated by the Cesàro operator

$$\mathcal{C}_{\alpha}(f)(x) = \alpha \int_0^1 \frac{(1-t)^{\alpha-1}}{t} f(\frac{x}{t}) dt, \quad \alpha > 0.$$

Kanjin showed in ([9]) that  $C_{\alpha}$  is bounded on  $H^p(\mathbb{R})$  provided  $\frac{2}{2\alpha+1} , and that this result actually is an application of the following:$ 

**Theorem A** ([9]). Let 0 , <math>M = [1/p - 1/2] + 1, and

$$A_{\Phi,p} = \int_0^\infty t^{-1+1/p} |\Phi(t)| \, dt.$$

Suppose  $A_{\Phi,1} + A_{\Phi,2} < \infty$ , and  $\widehat{\Phi} \in C^{2M}(\mathbb{R})$  with

$$\sup_{\xi \in \mathbb{R}} |\xi|^M \left( \left| \widehat{\Phi}^{(M)}(\xi) \right| + \left| \widehat{\Phi}^{(2M)}(\xi) \right| \right) < \infty.$$

Then  $h_{\Phi}$  is a bounded operator on  $H^p(\mathbb{R})$ .

The proof of Theorem A is based on the Taibleson-Weiss atomic-molecular characterization of the Hardy space (see [28]). Kanjin proved that if  $\Phi$  satisfies the conditions of Theorem A, then the Hausdorff operator  $h_{\Phi}$  maps an atom to an  $H^p$ molecule. Later on, using a different method, which includes a modified atomic decomposition of  $H^p$ , Miyachi improved Kanjin's result by showing that the Cesàro operator  $C_{\alpha}$  is bounded on  $H^p(\mathbb{R})$  for any  $\alpha > 0$  and all p > 0 (see [23]). In [14] Liflyand and Miyachi further extended the method used by Miyachi in [23] to study the  $H^p(\mathbb{R})$  boundedness of the Hausdorff operator  $h_{\Phi}$ . First, they observed that the following Theorem B is a direct corollary of Theorem A.

**Theorem B** ([14]). Let 0 and <math>M = [1/p - 1/2] + 1. If  $\Phi \in C^M$  and its support is a compact set in  $(0, \infty)$ , then the Hausdorff operator  $h_{\Phi}$  is bounded on  $H^p(\mathbb{R})$ .

Liflyand and Miyachi then generalized the main result in [23] and obtained the following two theorems.

**Theorem C** ([14]). Let 0 , <math>M = [1/p - 1/2] + 1 and let  $\varepsilon$  be a positive number. If  $\Phi \in C^M(0, \infty)$  satisfies

$$\left|\Phi^{(k)}(t)\right| \leq \min\{t^{\varepsilon}, t^{-\varepsilon}\}t^{-1/p-k}$$

for k = 0, 1, ..., M, then  $h_{\Phi}$  is bounded on  $H^p(\mathbb{R})$ .

**Theorem D** ([14]). Let 0 , <math>M = [1/p - 1/2] + 1 and let  $\varepsilon$  and a be positive numbers. Suppose that  $\Phi$  is a function on  $(0, \infty)$  such that the support of  $\Phi$  is a compact subset of  $(0, \infty)$ . If  $\Phi$  is of class  $C^M$  on  $(0, a) \cup (a, \infty)$  and satisfies

$$\left|\Phi^{(k)}(t)\right| \leq |t-a|^{\varepsilon-1-k} \text{ for } k=0,1,...,M_{\varepsilon}$$

then  $h_{\Phi}$  is bounded on  $H^p(\mathbb{R})$ .

In this thesis, we will establish another sufficient condition for  $H^p$  boundedness of  $h_{\Phi}$  with less smoothness on  $\Phi$  than that in Theorems B to D.

Next, we observe that all methods used to show the boundedness of  $h_{\Phi}$  on  $H^p(\mathbb{R})$ in the above theorems fail in the high dimensional case, and we notice that there is no result involving the boundedness on  $H^p(\mathbb{R}^n)$  in literature for the high dimensional Hausdorff operators, if  $0 and <math>n \ge 2$ . Thus we are particularly interested in studying such problems in the high dimensional spaces.

In the one-dimensional case, when x > 0, by a changing of variables,

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(\frac{x}{t})}{t} f(t) dt$$

This suggests us to study two different extensions of Hausdorff operator in the high dimensional space:

$$\tilde{H}_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^{n-\beta}} f(\frac{x}{|y|}) \, dy, \quad n \ge \beta \ge 0,$$

and

$$H_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(\frac{x}{|y|})}{|y|^{n-\beta}} f(y) \, dy, \quad n \ge \beta \ge 0$$

where  $\Phi$  is a locally integrable function. We denote

$$\tilde{H}_{\Phi,0} = \tilde{H}_{\Phi}$$
 and  $H_{\Phi,0} = H_{\Phi}$ .

We note that operators of the form  $H_{\Phi}$  include the high dimensional Hardy oper-

ator

$$\frac{1}{|x|^n}\int_{|y|\leqslant |x|}f(y)dy$$

and the adjoint Hardy operator

$$\int_{|y| \ge |x|} \frac{f(y)}{|y|^n} dy$$

In [10, 12, 15, 18], Liflyand and Mòricz addressed the following theorem.

**Theorem E** (Liflyand and Mòricz).  $\tilde{H}_{\Phi}$  is bounded on  $H^1(\mathbb{R}^n)$ ,  $n \ge 2$ , if  $\Phi \in L^1(\mathbb{R}^n)$ .

Again, we face the following question.

Question 2 (Liflyand [13]). Determine the sharpness of the condition on  $\Phi$  given in Theorem E. That is, determine whether  $\tilde{H}_{\Phi}$  is bounded on  $H^1(\mathbb{R}^n), n \ge 2$ , if and only if  $\Phi \in L^1(\mathbb{R}^n)$ .

Also, we notice that there is no research in the literature addressing  $H^p(\mathbb{R}^n)$ boundedness of either the operator  $\tilde{H}_{\Phi}$  or the operator  $H_{\Phi}$ . All methods of treating the one dimensional operator used in Theorems A to D fail to establish a similar theorem on the high dimensional case. This raises the following question.

Question 3. Establish some  $H^p(\mathbb{R}^n)$  boundedness theorems for the operators  $\hat{H}_{\Phi}$  or  $H_{\Phi}$  for  $n \ge 2$  and 0 .

As usual, the notation,  $A \leq B$  means that there is a constant C > 0 that is independent of all essential variables such that  $A \leq CB$ . Similarly, we use the notation  $A \cong B$  if there exist positive constants C and c, independent of all essential variables, such that

$$cB \leq A \leq CB$$
.

The structure of this thesis is as follows.

Chapter 2 contains some preliminary knowledge. This summary contains no new results, but several facts which are required for later chapters.

Chapter 3 presents some results on  $H^p(\mathbb{R})$  for 0 . In particular, Chapter 3 improves on the smoothness conditions described in Theorems B to D.

Chapter 4 includes several results on  $H^1(\mathbb{R})$  and related spaces. It includes an answer to Question 1.

Chapter 5 concerns  $H^p(\mathbb{R}^n)$ ,  $0 , <math>n \geq 2$ . This chapter addresses Questions 2 and 3.

Chapter 6 explores extending the techniques used for the previous results to study Hausdorff operators on the related Herz-type Hardy spaces, and also to study bilinear Hausdorff operators.

The following are the new theorems that we obtained.

In Chapter 3, we prove a boundedness result for Hausdorff operators on onedimensional Hardy spaces:

**Theorem 3.1.** Let  $\Phi$  be a Lebesgue integrable function. Denote  $\phi(t) = \frac{\Phi(\frac{1}{t})}{t}$  for t > 0, and  $\phi(t) = 0$  for  $t \leq 0$ . Assume  $0 and <math>\alpha = \frac{1}{p} - 1$ . If  $\phi \in \Lambda_{\alpha}$  then

$$||h_{\Phi}(f)||_{H^{p,\infty}(\mathbb{R})} \le ||f||_{H^{p}(\mathbb{R})}.$$

If there exists a small  $\epsilon > 0$ , such that  $\alpha - \epsilon > 0$  and  $\phi \in \Lambda_{\alpha + \epsilon} \cap \Lambda_{\alpha - \epsilon}$ , then

$$\|h_{\Phi}(f)\|_{H^{p}(\mathbb{R})} \leq \|f\|_{H^{p}(\mathbb{R})}.$$

In Chapter 4, we study the boundedness of Hausdorff operators on the Hardy space  $H^1(\mathbb{R})$ , obtaining the following main theorems:

**Theorem 4.2.** Let  $\Phi$  be a nonnegative valued locally integrable function.

(1)  $h_{\Phi}$  is bounded on  $H^1(\mathbb{R})$  if and only if  $\Phi \in L^1(0, \infty)$ .

(2)  $h_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R})$  if and only if

$$\int_{1}^{\infty} \Phi(t) dt + \int_{0}^{1} \Phi(t) \left(1 + \log(\frac{1}{t})\right) dt < \infty.$$

**Theorem 4.3.** Let  $\phi(t) = t^{-1}\Phi_0(t)$ . Suppose  $\phi \in BMO(\mathbb{R})$ . Then  $h_{\Phi}$  extends to a bounded operator from  $H^1(\mathbb{R})$  to  $H^{1,\infty}(\mathbb{R})$ .

Corollary 4.6. The Hardy operator  $\mathcal{H}$  has the following properties

- (1)  $\mathcal{H}$  is not bounded on  $H^1(\mathbb{R})$ .
- (2)  $\mathcal{H}$  is not bounded on  $h^1(\mathbb{R})$ .
- (3)  $\mathcal{H}$  is bounded from  $H^1(\mathbb{R})$  to  $H^{1,\infty}(\mathbb{R})$ .
- (4)  $\mathcal{H}$  is bounded on  $H^1_A(0,\infty)$ .

We also apply our results to some generalizations of the Hardy operator.

In Chapter 5, we begin by generalizing Theorem 3.1 to obtain

**Theorem 5.1.** Let  $\Phi$  be a nonnegative valued locally integrable function.

- (1)  $\tilde{H}_{\Phi}$  is bounded on  $H^1(\mathbb{R}^n)$  if and only if  $\Phi \in L^1(\mathbb{R}^n)$ .
- (2)  $\tilde{H}_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R}^n)$  if and only if

$$\int_{|y| \ge 1} \Phi(y) \, dy + \int_{|y| \le 1} \Phi(y) \left( 1 + \log\left(\frac{1}{|y|}\right) \right) \, dy < \infty.$$

We next show some boundedness results for the power-weight Lebesgue spaces, particularly,

**Theorem 5.4.** Let  $1 \leq p, q < \infty$ ,  $0 < \beta < n$ ,  $\gamma > \beta p - n$  and

$$\frac{1}{p} - \frac{\beta}{n+\gamma} = \frac{1}{q}$$

In addition, let

$$C_{p,\epsilon} = \int_0^\infty |\Phi(t)|^{\frac{p}{p-1}} t^{\frac{\gamma-\beta p+n-p+1}{p-1}+\epsilon} dt \text{ for } p > 1, \qquad C_{1,\epsilon} = \||\cdot|^{n-\beta+\gamma+\epsilon} \Phi(\cdot)\|_{L^\infty}.$$

For any  $p \ge 1$ , if, for arbitrarily small positive  $\epsilon$ ,  $C_{p,\pm\epsilon} < \infty$ , then

$$\|H_{\Phi,\beta}(f)(x)\|_{L^q(|x|^{\gamma}dx)} \le \|f\|_{L^p(|x|^{\gamma}dx)}.$$

Finally, we show sufficient conditions for boundedness of operators on high-dimensional Hardy spaces.

**Lemma 5.5.** Let  $0 \leq \beta < n$  and

$$\psi(y) = \frac{\Phi(1/|y|)}{|y|^{n-\beta}}.$$

Assume  $0 and <math>\alpha = n\left(\frac{1}{p} - 1\right)$ . If  $\psi \in \Lambda_{\alpha}$ , then

$$\|H_{\Phi,\beta}(f)\|_{L^{q,\infty}(\mathbb{R}^n)} \le \|f\|_{H^p(\mathbb{R}^n)}$$

where q satisfies

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

**Theorem 5.6.** Let  $\beta, p, \alpha, q, \psi$  be as in Lemma 5.5. If for some  $\epsilon > 0$  small enough that  $\alpha - \epsilon > 0$ ,  $\psi \in \Lambda_{\alpha + \epsilon} \cap \Lambda_{\alpha - \epsilon}$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^q(\mathbb{R}^n)} \le \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

**Theorem 5.11.** Suppose  $0 , <math>0 < \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform of  $\Phi$  and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M = n(\frac{1}{p}-1)$ , suppose that  $\widehat{\Phi}$  is a function in  $C^{2M+n}(\mathbb{R}^n)$  with compact support in the set  $\mathbb{R}^n \setminus \{0\}$ . Then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^{q,\infty}(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

**Theorem 5.13.** Suppose  $0 , <math>0 < \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform

of  $\Phi$  and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M > n(\frac{1}{p}-1)$ , suppose that  $\widehat{\Phi}$  is a function in  $C^{2M+n}(\mathbb{R}^n)$  with compact support in the set  $\mathbb{R}^n \setminus \{0\}$ . Then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^q(\mathbb{R}^n)} \leq \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

In Chapter 6, we establish a size condition for boundedness of Hausdorff operators on the Herz-type Hardy spaces:

**Theorem 6.1.** Let  $0 , and <math>n(1 - \frac{1}{q}) \le \alpha < \infty$ .

(1) For 0 , let

$$C_{p,\sigma} = \int_{\mathbb{R}^n} \frac{|\Phi(y)|}{|y|^n} |y|^{\alpha} |y|^{\frac{n}{q}} (1 + \log|y|)^{\sigma} dy.$$

If, for some  $\sigma > \frac{1-p}{p}$ ,  $C_p := C_{p,\sigma} < \infty$ , then

$$\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} \leq \|f\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})}.$$

(2) For p = 1, let

$$C_{1} = \int_{\mathbb{R}^{n}} \frac{|\Phi(y)|}{|y|^{n}} |y|^{\alpha} |y|^{\frac{n}{q}} dy.$$

If  $C_1 < \infty$ , then

$$\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,1}_q(\mathbb{R}^n)} \leqslant \|f\|_{H\dot{K}^{\alpha,1}_q(\mathbb{R}^n)}.$$

We further provide a condition for a modified Hausdorff operator to map the Herztype Hardy spaces into a weak Hardy space. Here, the modified Hausdorff operator  $\tilde{h}_{\Phi}$  is defined by

$$\widetilde{h}_{\Phi}(f)(x) = \int_{\mathbb{R}} \frac{\Phi(t)}{t} f\left(\frac{x}{t}\right) dt$$

and we let

$$\phi(v) = \begin{cases} \frac{\Phi\left(\frac{1}{v}\right)}{v} & \text{if } v \neq 0\\ \lim_{v \to 0} \frac{\Phi\left(\frac{1}{v}\right)}{v} & \text{if } v = 0. \end{cases}$$

**Theorem 6.3.** Let  $0 , and <math>(1 - \frac{1}{q}) \le \alpha < \infty$ . For

$$r = \frac{1}{\alpha + \frac{1}{q}}, \quad N = \left[\alpha + \frac{1}{q} - 1\right] = \left[\frac{1}{r} - 1\right], \quad \frac{1}{q'} + \frac{1}{q} = 1.$$

Let

$$\gamma = \frac{1}{r} - 1 - N = \frac{1}{r} - 1 - \left[\frac{1}{r} - 1\right].$$

If  $\phi \in C^N$  and

$$\int_{-\rho}^{\rho} \left| \phi^{(N)}(t) - \phi^{(N)}(0) \right|^{q'} dt \leqslant \rho^{1 + \gamma q'}$$

uniformly for  $\rho > 0$ , then we have

$$\left\|\widetilde{h}_{\Phi}(f)\right\|_{H^{r,\infty}} \leqslant \|f\|_{H\dot{K}^{\alpha,p}_{q}}.$$

Finally, we prove some preliminary boundedness results for bilinear Hausdorff operators on one-dimensional Lebesgue spaces.

**Theorem 6.5.** Let  $m, k = 1, 2, \ldots$  For any  $p, p_1, p_2, r, p' \ge 1$  satisfying

$$\frac{1}{p_1} + \frac{1}{p_2} = \frac{1}{p}, \qquad \frac{1}{r} = \frac{m}{p_1} + \frac{k}{p_2}, \qquad \frac{1}{p} + \frac{1}{p'} = 1,$$

if

$$\left(\int_{-\infty}^{\infty} \left|\frac{\Phi\left(\frac{1}{t}\right)}{t}\right|^{p'} dt\right)^{\frac{1}{p'}} < \infty$$

then

$$\|H_{\Phi,m,k}(f,g)\|_{L^{r,\infty}} \leqslant \|f\|_{L^{p_1}} \|g\|_{L^{p_2}}.$$

#### CHAPTER 2

#### PRELIMINARIES

#### 2.1 LEBESGUE AND HARDY SPACES

#### 2.1.1 Lorentz Spaces

Let dx be the Lebesgue measure on  $\mathbb{R}^n$ . The Lebesgue space  $L^p$ , 0 , is theset of all measurable functions <math>f satisfying

$$\|f\|_{L^p} = \left(\int_{\mathbb{R}^n} |f(x)|^p dx\right)^{\frac{1}{p}} < \infty.$$

The power weight Lebesgue space  $L^p(|x|^{\gamma} dx)$ ,  $1 \leq p \leq \infty$ , is the set of all measurable functions f satisfying

$$||f||_{L^p(|x|^{\gamma}dx)} = \left(\int_{\mathbb{R}^n} |f(x)|^p |x|^{\gamma} dx\right)^{\frac{1}{p}} < \infty.$$

For a measurable function f, we define the set  $E_f(t)$ , for any t > 0, by

$$E_f(t) = \{x : |f(x)| \ge t\}.$$

The distribution function of f is  $\mu_f(t) = |E_f(t)|$ , the Lebesgue measure of  $E_f(t)$ . We define the weak Lebesgue space by

$$L^{p,\infty} = \{f : \|f\|_{L^{p,\infty}}\} < \infty,$$

where

$$||f||_{L^{p,\infty}}^p = \sup_{t>0} (t^p(\mu_f(t))).$$

When  $0 and <math>0 < q < \infty$ , we define the norm (or quasi-norm)

$$\|f\|_{L^{p,q}} = p^{\frac{1}{q}} \left( \int_0^\infty t^q |\mu_f(t)|^{\frac{q}{p}} \frac{dt}{t} \right)^{\frac{1}{q}}.$$

The Lorentz space  $L^{p,q}$  is the set of all measurable function f satisfying

$$L^{p,q} = \{f : \|f\|_{L^{p,q}} < \infty\}.$$

The Lebesgue space  $L^p$  and its weak version  $L^{p,\infty}$  are special cases of the Lorentz space  $L^{p,q}$ . (It is easy to see that  $L^{p,p} = L^p$ .)

#### 2.1.2 Hardy Spaces

When  $0 , the dual space <math>(L^p)'$  of  $L^p$  contains only the zero function, which makes the structure of  $L^p$  difficult to study in this case. Instead, we study a closely related space, the Hardy space  $H^p$ .

Let  $S(\mathbb{R}^n)$  be the set of Schwartz functions and  $\Psi \in S(\mathbb{R}^n)$  satisfying

$$\int_{\mathbb{R}^n} \Psi(y) \, dy = 1,$$

and denote  $\Psi_s(y) = \frac{1}{s^n} \Psi(\frac{y}{s})$ 

The Hardy space  $H^p(\mathbb{R}^n)$  is the space of all distributions f satisfying

$$\|f\|_{H^p(\mathbb{R}^n)} := \left\| \sup_{0 < s < \infty} |\Psi_s * f| \right\|_{L^p(\mathbb{R}^n)} < \infty$$

We similarly define the weak Hardy space  $H^{p,\infty}$  to be the set of distributions f satisfying

$$\|f\|_{H^{p,\infty}(\mathbb{R}^n)} := \left\| \sup_{0 < s < \infty} |\Psi_s * f| \right\|_{L^{p,\infty}(\mathbb{R}^n)} < \infty$$

and more generally the Hardy-Lorentz space  $H^{p,q}$  the set of distributions f satisfying

$$\|f\|_{H^{p,q}(\mathbb{R}^n)} := \left\|\sup_{0 < s < \infty} |\Psi_s * f|\right\|_{L^{p,q}(\mathbb{R}^n)} < \infty.$$

The local Hardy space  $h^p(\mathbb{R}^n)$  is the space of all distributions f satisfying

$$\|f\|_{h^{p}(\mathbb{R}^{n})} := \left\|\sup_{0 < s < 1} |\Psi_{s} * f|\right\|_{L^{p}(\mathbb{R}^{n})} < \infty$$

The definitions above are independent (up to equivalence of norms) of the choices of

the function  $\Psi$ . From the definition, it is easy to see that

$$\|f\|_{h^p} \leq \|f\|_{H^p}$$
,

so that we have an embedding

$$H^p \subset h^p.$$

We note that  $||f||_{H^p}$  and  $||f||_{h^p}$  are norms when  $p \ge 1$ , and they are quasi-norms when 0 . When <math>p > 1, it is well known that  $H^p = L^p$ .

2.2 DUALITY

One motivation for studying  $H^p$  rather than  $L^p$  when  $0 is that the dual of <math>L^p$ is degenerate in this case, while the dual of  $H^p$  is well-developed. The dual of  $H^p$  is an appropriately chosen Companato space; for a good treatment of this space and its duality with  $H^p$ , see for instance [21]. However, in computation, it is easier to pass to a related space, the Lipschitz space  $\Lambda_{\alpha}$ , introduced below.

#### 2.2.1 Lipschitz Spaces

For  $\alpha \ge 0$  we define the Lipschitz space  $\Lambda_{\alpha}$  as follows.

- For  $\alpha = 0, \Lambda_0 = L^{\infty}, ||f||_{\Lambda_0} = ||f||_{L^{\infty}}.$
- For  $0 < \alpha < 1$ ,  $\Lambda_{\alpha} = \{f : \exists c > 0 \text{ s.t. } \forall x, y, |f(x) f(y)| \leq c |x y|^{\alpha}\},$  $\|f\|_{\Lambda_{\alpha}} = \inf \{c : \forall x, y, |f(x) - f(y)| \leq c |x - y|^{\alpha}\}.$
- For  $\alpha \ge 1$  write  $\alpha = k + \beta$ ,  $k \in \mathbb{N}$ ,  $0 \le \beta < 1$ .
  - $$\begin{split} \Lambda_{\alpha} &= \{ f : \forall \kappa \text{ multiindex s.t. } |\kappa| = k, f^{(\kappa)} \in \Lambda_{\beta} \}, \\ \|f\|_{\Lambda_{\alpha}} &= \max_{|\kappa| = k} \|f^{(\kappa)}\|_{\Lambda_{\beta}}. \end{split}$$

Intuitively, we can think of  $f \in \Lambda_{\alpha}$  as meaning that f is " $\alpha$  times" differentiable with bounded derivative. (This certainly makes sense when  $\alpha \in \mathbb{N}$ .) This intuition suggests that for  $\alpha < \beta$ ,  $\Lambda_{\beta} \subseteq \Lambda_{\alpha}$ ; but this is not true in general. We do, however, have the following inclusions.

- If  $\alpha < \gamma < \beta$ , then  $\Lambda_{\alpha} \cap \Lambda_{\beta} \subseteq \Lambda_{\gamma}$ .
- If  $\alpha < \beta$  and K is compact, then  $\Lambda_{\beta}(K) \subseteq \Lambda_{\alpha}(K)$ .

We don't have to look very far to find an exception to the expected inclusion. For example,  $f(x) = x^2 \in \Lambda_2(\mathbb{R})$  but not in  $\Lambda_1(\mathbb{R})$ .

When  $\alpha = n(\frac{1}{p} - 1)$ , for any  $f \in H^p(\mathbb{R}^n)$ ,  $g \in \Lambda_\alpha(\mathbb{R}^n)$ , an easy computation shows the pairing inequality

$$|\langle f, g \rangle| = \|f\|_{H^p} \|g\|_{\Lambda_{\alpha}}.$$
(2.1)

#### 2.2.2 BMO

The dual space of  $H^1(\mathbb{R}^n)$  is BMO( $\mathbb{R}^n$ ), the space of bounded mean oscillation. Here, we recall that BMO( $\mathbb{R}^n$ ) is the space of all locally integrable functions f satisfying

$$||f||_{BMO} := \sup \frac{1}{|B|} \int_{B} \left| f(x) - \frac{1}{|B|} \int_{B} f(t) dt \right| dx < \infty,$$

where the sup is taken over all balls B in  $\mathbb{R}^n$ . It is known that the space  $L^{\infty}$  is a proper subspace of BMO and

$$\log |x| \in BMO \setminus L^{\infty}.$$

#### 2.3 Alternate Characterizations of $H^p$

The Hardy spaces have several equivalent characterizations. In this thesis, we will invoke the atomic decomposition and the Hilbert/Riesz transform characterization.

#### 2.3.1 Atomic Characterization

Let  $0 , the integer part of <math>[n(\frac{1}{p}-1)]$ . We say that a function  $a(x) \in L^q(\mathbb{R}^n)$  is a (p,q,s) atom with the center at  $x_0$ , if it satisfies the following conditions:

Support Condition

$$\operatorname{supp}(a) \subset B(x_0, \rho), \ \rho > 0,$$

Cancellation Condition

$$\int_{\mathbb{R}^n} y^k a(y) \, dy = 0 \text{ for all multi-indices } k \text{ such that } |k| \leq s,$$

Size Condition

$$\|a\|_{L^q} \leqslant \rho^{(\frac{n}{q} - \frac{n}{p})}.$$

A function b is called a small (p,q) block if b satisfies the support and size conditions with  $\rho < 1$ . A function B is called a big (p,q) block if B satisfies the support and size conditions with  $\rho \ge 1$ .

A well-known theorem by Coifman [3] says that any  $f \in H^p(\mathbb{R})$  has an atomic decomposition

$$f = \sum \lambda_j a_j,$$

where  $\{\lambda_j\} \in \ell^p$ , and that

$$\|f\|_{H^p}^p \cong \inf\left\{\sum |\lambda_j|^p : f = \sum \lambda_j a_j, \ a_j \text{ are } (p,q,s) \text{ atoms}\right\}.$$

It is also well known

$$\|a\|_{L^p} \leqslant \|a\|_{H^p} \leqslant 1$$

uniformly for all (p, q, s) atoms if  $s > n(\frac{1}{p} - 1)$ .

The space  $h^p$  has an similar decomposition that was discovered by Goldberg in [6]. Namely, any  $f \in h^p(\mathbb{R})$  has a decomposition

$$f = \sum \lambda_j a_j,$$

where

$$||f||_{h^p}^p \cong \inf\left\{\sum |\lambda_j|^p : f = \sum \lambda_j a_j\right\}$$

and each  $a_j$  is a (p, q, s) atom or a big (p, q) block. A simple computation (or see [6]) shows

$$\|B\|_{h^p} \leqslant 1$$

uniformly for all big (p,q) blocks B. We establish a boundedness result for some small blocks as Lemma 4.1.

2.3.2 Hilbert and Riesz Transform

When n = 1, another important characterization of  $H^p(\mathbb{R})$  is the one involving the Hilbert transform

$$\mathcal{R}f(x) = p.v.\frac{1}{\pi} \int_{\mathbb{R}} \frac{f(y)}{x-y} \, dy.$$

The Hilbert transform is a very important operator in analysis. From [27] we know that a non-identity linear transform T commutes with dilation, translation and reflection if and only if T is the Hilbert transform, up to a constant multiple. It is known that the  $H^p$  space can be characterized by the Hilbert transform in the sense that, for all  $f \in H^p \cap L^2$ ,

$$||f||_{H^p} \cong ||\mathcal{R}f||_{L^p} + ||f||_{L^p}.$$

By checking the Fourier transform, it is easy to see that

$$\widehat{\mathcal{R}f(\xi)} = i \operatorname{sgn}(\xi)\widehat{f}(\xi),$$

so  $-\mathcal{R}^2 = \text{Id}$  is the identity operator. The relationship between the Hilbert transform and the Hausdorff operator was studied in [17] by Liflyand and Móricz . Particularly, they obtained the identity

$$\mathcal{R}(h_{\Phi}f)(x) = h_{\Phi}(\mathcal{R}f)(x).$$

When  $n \ge 2$ , one can similarly characterize the space  $H^p$  by using the Riesz

transform. For  $j \in \{1, 2, ..., n\}$ , the *j*th Riesz transform  $\mathcal{R}_j$  is defined by

$$\mathcal{R}_j(f)(x) = p.v. \int_{\mathbb{R}^n} c_n \frac{y_j}{|y|^{n+1}} f(x-y) \, dy,$$

where the constant  $c_n$  is given by

$$c_n = \Gamma\left(\frac{n+1}{2}\right) / \pi^{\frac{n+1}{2}}$$

.

The Fourier transform of the kernel of  $\mathcal{R}_j$  is called the symbol of  $\mathcal{R}_j$ . From [25], we know that the symbol of  $\mathcal{R}_j$  is  $i\xi_j/|\xi|$ . From this, we can easily see that

$$\sum_{j=1}^n \mathcal{R}_j^2 = -\mathrm{Id},$$

where Id is the identity map. We also denote

$$\mathcal{R}_0(f)(x) = f(x).$$

For an integer  $L \ge 0$ , and a multi-index  $J = \{j_1, \ldots, j_L\} \in \{0, 1, 2, \ldots, n\}^L$ , let  $\mathcal{R}_J(f)$ denote the generalized Riesz transform  $\mathcal{R}_J(f) = \mathcal{R}_{j_1} \ldots \mathcal{R}_{j_L} f$ . It is known that for Lsuch that  $p > \frac{n-1}{n-1+L}$  and all  $f \in L^2 \cap H^p(\mathbb{R}^n)$ ,

$$\sum_{J} \|\mathcal{R}_{J}(f)\|_{L^{p}(\mathbb{R}^{n})} \cong \|f\|_{H^{p}(\mathbb{R}^{n})},$$

where the sum is taken over all  $J \in \{0, 1, 2, ..., n\}^{L}$ . It follows easily that Riesz transforms are bounded on  $H^{p}$  for all 0 .

We have an analogous result for the weak Hardy spaces:

$$\sum_{J} \|\mathcal{R}_{J}(f)\|_{L^{p,\infty}(\mathbb{R}^{n})} \cong \|f\|_{H^{p,\infty}(\mathbb{R}^{n})}$$

Particularly, for n = 1, we have

$$\|f\|_{H^{p,\infty}} \cong \|\mathcal{R}f\|_{L^{p,\infty}} + \|f\|_{L^{p,\infty}}$$

for  $f \in H^{p,\infty} \cap L^2$ .

One major tool we will use to generate boundedness results is interpolation, which allows us to obtain strong boundedness from weak boundedness. In particular, we will use two interpolation theorems: Stein-Weiss analytic interpolation [27] and Marcinkiewisz interpolation.

2.4.1 Stein-Weiss Analytic Interpolation

Let  $(M, \mathcal{A}, \mu)$  and  $(N, \mathcal{B}, \nu)$  be two measure spaces and D(M) and D(N) be the sets of all simple functions on  $(M, \mathcal{A}, \mu)$  and  $(N, \mathcal{B}, \nu)$  respectively. Define the set  $\Delta$  to be the strip

$$\Delta = \{ z \in \mathbb{C} : 0 \leq \operatorname{Re} z \leq 1 \}$$
$$\mathring{\Delta} = \{ z \in \mathbb{C} : 0 < \operatorname{Re} z < 1 \}.$$

**Definition 2.1.** Let  $T_z$  be a linear operator  $T_z : D(M) \to D(N)$  for each  $z \in \Delta$ . If for any  $f \in D(M), g \in D(N)$ ,

$$h(z) = \int_{N} T_z(f)g \, dv \tag{2.2}$$

is analytic on  $\mathring{\Delta}$  and continuous on  $\Delta$ , and there exists a constant  $a < \pi$ , such that

$$e^{-a|y|}\log|h(z)|, \quad z = x + iy$$

has an upper bound on  $\Delta$ , then we call the family of the operators  $\{T_z\}$  admissible.

**Theorem F** (Stein-Weiss). Let  $\{T_z\}$  be an admissible family and z = x + iy. If

$$\|T_{iy}(f)\|_{L^{q_0,\infty}(N)} \leq M_0(y) \|f\|_{L^{p_0}(M)}$$
$$\|T_{1+iy}(f)\|_{L^{q_1,\infty}(N)} \leq M_1(y) \|f\|_{L^{p_1}(M)}.$$

for all  $f \in D(M)$ , and if there exists  $b < \pi$  for which

$$\sup_{y \in \mathbb{R}} e^{-b|y|} \log M_j(y) < \infty, \quad j \in \{1, 2\},$$
(2.3)

then for  $t \in (0, 1)$ , and  $p_t, q_t$  be defined by

$$\frac{1}{p_t} = \frac{1-t}{p_0} + \frac{t}{p_1}, \quad \frac{1}{q_t} = \frac{1-t}{q_0} + \frac{t}{q_1}$$

we have

$$||T_t(f)||_{L^{q_t}(N)} \leq M_t ||f||_{L^{p_t}(M)}$$

for all  $f \in D(M)$ , where  $M_t$  is a bounded constant that can be computed using the three circle theorem (a theorem in Complex Analysis).

We will be using Stein-Weiss analytic interpolation to prove results about spaces of Schwartz distributions. In general measure spaces, we have no derivatives, no Schwartz space, no  $C^{\infty}$ , etc. For our purposes, however, we will fix  $M = N = \mathbb{R}^n$ . Further, since in Stein-Weiss analytic interpolation we need the transformations  $T_z$ to work on dense subspaces of our distribution spaces, we will replace D(M), D(N)by

$$C_c^{\infty}(\mathbb{R}^n) = \{ f \in C^{\infty} : f \text{ has compact support} \}.$$

In this context, we can replace equation (2.2) by

$$h(z) = \int_{\mathbb{R}^n} T_z(f)(x)g(x) \ d\varpi(x), \qquad (2.4)$$

where  $d\varpi(x)$  is some measure on  $\mathbb{R}^n$ . We will use the following corollary of Theorem F:

**Corollary 2.2.** Let  $\{T_z\}$  be an admissible family and z = x + iy. If

$$\left\|T_{iy}(f)\right\|_{L^{q_0,\infty}(\mathbb{R}^n,d\varpi)} \leq M_0(y) \left\|f\right\|_{L^{p_0}(\mathbb{R}^n,d\varpi)}$$

$$\left\|T_{1+iy}(f)\right\|_{L^{q_1,\infty}(\mathbb{R}^n,d\varpi)} \le M_1(y) \left\|f\right\|_{L^{p_1}(\mathbb{R}^n,d\varpi)}$$

for all  $f \in C_c^{\infty}(\mathbb{R}^n)$ , and there exists  $b < \pi$  for which

$$\sup_{y \in \mathbb{R}} e^{-b|y|} \log M_j(y) < \infty, \quad j \in \{1, 2\}$$

then for  $t \in (0, 1)$ , and  $p_t, q_t$  be defined by

$$\frac{1}{p_t} = \frac{1-t}{p_0} + \frac{t}{p_1}, \quad \frac{1}{q_t} = \frac{1-t}{q_0} + \frac{t}{q_1}$$

we have

$$\left\|T_t(f)\right\|_{L^{q_t}(\mathbb{R}^n, d\varpi)} \leq M_t \left\|f\right\|_{L^{p_t}(\mathbb{R}^n, d\varpi)}$$

for all  $f \in C_c^{\infty}(\mathbb{R}^n)$ , where  $M_t$  is a bounded constant, and  $d\varpi$  is some measure on  $\mathbb{R}^n$ .

#### 2.4.2 Marcinkiewisz Interpolation

T is called a quasilinear operator if there exists a constant C > 0 such that T satisfies

$$|T(f+g)(x)| \leq C\left(|Tf(x)| + |Tg(x)|\right)$$

for almost every x. An operator T (possibly quasilinear) satisfying an estimate of the form

$$\left\|Tf\right\|_{L^{q,\infty}} \leqslant C \left\|f\right\|_{L^{p}}$$

is said to be of weak type (p,q). An operator is simply of type (p,q) if T is a bounded transformation from  $L^p$  to  $L^q$ :

$$\left\|Tf\right\|_{L^{q}} \leqslant C \left\|f\right\|_{L^{p}}.$$

Now we are ready to recall the Marcinkiewicz interpolation theorem (see [27] for more details).

**Lemma 2.3** (Marcinkiewicz Interpolation). If T is a quasilinear operator of weak type  $(p_0, q_0)$  and of weak type  $(p_1, q_1)$  where  $q_0 \neq q_1$ , then for each  $\theta \in (0, 1)$ , T is of type (p, q), for p and q with  $p \leq q$  of the form

$$\frac{1}{p} = \frac{1-\theta}{p_0} + \frac{\theta}{p_1}, \qquad \frac{1}{q} = \frac{1-\theta}{q_0} + \frac{\theta}{q_1}.$$

Many of the results we would like to obtain for higher dimensional Hardy spaces can be more easily obtained on a subspace, the so-called Herz-type Hardy space. Herz type spaces are important function spaces in harmonic analysis. Lu and Yang have made tremendous contributions on these spaces. Their book [20] (joint with Hu) is the unique research book on this topic.

For each  $k \in \mathbb{Z}$ , define

$$B_k = \left\{ x \in \mathbb{R}^n : |x| \leq 2^k \right\}, \quad E_k = B_k \setminus B_{k-1}$$

and let  $\chi_k$  denote the characteristic function of  $E_k$ .

**Definition 2.4** (Homogeneous Herz Space). Let  $\alpha \in \mathbb{R}$ ,  $0 < p, q < \infty$ . The homogeneous Herz space  $\dot{K}^{\alpha,p}_q(\mathbb{R}^n)$  is defined by

$$\dot{K}_{q}^{\alpha,p} = \left\{ f \in L^{q}_{\text{loc}}(\mathbb{R}^{n} \setminus \{0\}) : \|f\|_{\dot{K}_{q}^{\alpha,p}} < \infty \right\},\$$

where

$$\|f\|_{\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} = \left(\sum_{k=-\infty}^{\infty} 2^{k\alpha p} \|f\chi_{k}\|^{p}_{L^{q}(\mathbb{R}^{n})}\right)^{\frac{1}{p}}.$$

**Definition 2.5** (Herz-type Hardy Space). Let  $\alpha \in \mathbb{R}$ ,  $0 , <math>1 < q < \infty$ . The homogeneous Herz-type Hardy space  $H\dot{K}_q^{\alpha,p}(\mathbb{R}^n)$  is defined by

$$H\dot{K}^{\alpha,p}_q(\mathbb{R}^n) = \left\{ f \in S'(\mathbb{R}^n) : Gf \in \dot{K}^{\alpha,p}_q(\mathbb{R}^n) \right\},\,$$

where Gf is the grand maximal function of f and

$$\|f\|_{H\dot{K}^{\alpha,p}_q(\mathbb{R}^n)} = \|Gf\|_{\dot{K}^{\alpha,p}_q(\mathbb{R}^n)}.$$

Similar to the Hardy spaces, the space  $H\dot{K}_q^{\alpha,p}$  can be decomposed into atoms.

**Definition 2.6** (Central Atom). Suppose  $1 < q < \infty$ ,  $n(1 - \frac{1}{q}) \leq \alpha < \infty$ , and  $s \geq [\alpha + n(\frac{1}{q} - 1)]$ . A function a(x) on  $\mathbb{R}^n$  is said to be a central  $(\alpha, q)$  atom if

(i) supp  $a \subset B(0,\rho)$ ,

(ii) 
$$||a||_{L^q(\mathbb{R}^n)} \leq |B(0,\rho)|^{-\alpha/n}$$
,  
(iii)  $\int_{\mathbb{R}^n} a(x) x^{\beta} dx = 0$  for any multi-index  $\beta$  with  $|\beta| \leq s$ .

It is known that, for  $0 , <math>1 < q < \infty$ , and  $n(1-\frac{1}{q}) \leq \alpha < \infty$ ,  $f \in H\dot{K}_q^{\alpha,p}(\mathbb{R}^n)$ if and only if there exist a sequence of numbers  $\{\lambda_k\} \in \ell^p$  and a sequence of central (a,q) atoms  $\{a_k\}$  with the support in  $B_k$  such that

$$f = \sum_{k=-\infty}^{\infty} \lambda_k a_k$$

in S'. Moreover,

$$\|f\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} \cong \inf\left\{ \left(\sum_{k} |\lambda_{k}|^{p}\right)^{\frac{1}{p}} : f = \sum_{k} \lambda_{k} a_{k} \text{ is an atomic decomposition of } f \right\}$$

By the definition, it is not difficult to see

$$\dot{K}_p^{\alpha,p} = L^p$$
 if  $\alpha = 0$  and  $p \ge 1$ .

When  $0 , and <math>\frac{n}{p} = \alpha + \frac{n}{q}$ ,  $H\dot{K}_{q}^{\alpha,p}$  is a subspace of  $H^{p}$ .

#### CHAPTER 3

#### ONE DIMENSIONAL HAUSDORFF OPERATORS: THE CASE 0

In this chapter, we obtain a boundedness result for Hausdorff operators

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(t)}{t} f\left(\frac{x}{t}\right) dt$$

on one-dimensional Hardy spaces  $H^p$ , 0 , which requires a relatively simple $smoothness condition on the function <math>\Phi$ . First, we recall the following result, due to Liflyand and Miyachi:

**Theorem B** ([14]). Let 0 and <math>M = [1/p - 1/2] + 1. If  $\Phi \in C^M$  and its support is a compact set in  $(0, \infty)$ , then the Hausdorff operator  $h_{\Phi}$  is bounded on  $H^p(\mathbb{R})$ .

Liflyand and Miyachi obtained further generalizations of Theorem B (Theorems C and D, see page 8), which imply the boundedness of the Cesàro operator  $C_{\alpha}$  for all  $\alpha, p > 0$ . However, the main purpose of this chapter is to use a different method from those in [9] and [14] to obtain a new sufficient condition on  $\Phi$  to ensure the  $H^p(\mathbb{R})$ boundedness of  $h_{\Phi}$ . We will establish the following result.

**Theorem 3.1.** Let  $\Phi$  be a Lebesgue integrable function. Denote  $\phi(t) = \frac{\Phi(\frac{1}{t})}{t}$  for t > 0, and  $\phi(t) = 0$  for  $t \leq 0$ . Assume  $0 and <math>\alpha = \frac{1}{p} - 1$ . If  $\phi \in \Lambda_{\alpha}$  then

$$\|h_{\Phi}(f)\|_{H^{p,\infty}(\mathbb{R})} \leq \|f\|_{H^{p}(\mathbb{R})}.$$

If there exists a small  $\epsilon > 0$ , such that  $\alpha - \epsilon > 0$  and  $\phi \in \Lambda_{\alpha + \epsilon} \cap \Lambda_{\alpha - \epsilon}$ , then

$$||h_{\Phi}(f)||_{H^{p}(\mathbb{R})} \leq ||f||_{H^{p}(\mathbb{R})}.$$

Clearly, Theorem 3.1 is an improvement of Theorem B. But we should point out that the theorem is mutually independent to Theorems C and D. First, Theorems C and D do not imply Theorem 3.1, since Theorem 3.1 assumes less smoothness condition. On the other hand, although in Theorems C and D,  $\Phi$  is assumed more smoothness on the set  $(0, a) \cup (a, \infty)$ ,  $\Phi$  is allowed to have very little smoothness at a single point a. With this advantage, one easily sees that Theorems C and D imply the  $H^p(\mathbb{R})$  boundedness of the Cesàro operator  $C_{\alpha}$  for any a > 0, while the  $H^p(\mathbb{R})$ boundedness of the Cesàro operator can not be deduced from Theorem 3.1 if  $\alpha < 1/p$ .

Proof of Theorem 3.1. The Hardy space  $H^p$  is a distribution space when 0 .However, it suffices to show the theorem for functions <math>f in the space  $H^p \cap L^2$ , since this space is dense in  $H^p$ . Recall that the  $H^p$  space can be characterized by the Hilbert transform in the sense of

$$\|f\|_{H^p} \cong \|\mathcal{R}f\|_{L^p} + \|f\|_{L^p}$$

for all  $f \in H^p \cap L^2$ . By Theorem 3 in [17],

$$\mathcal{R} \circ h_{\Phi}(f)(x) = h_{\Phi}(\mathcal{R}f)(x) = \int_{-\infty}^{\infty} \frac{\Phi(t)}{t} \left(\mathcal{R}f\right)\left(\frac{x}{t}\right) dt.$$

Changing variables  $\frac{1}{t} = v$ , by the definition of  $\phi$ , we have

$$\mathcal{R} \circ h_{\Phi}(f)(x) | = \left| \int_{-\infty}^{\infty} \frac{\Phi(\frac{1}{v})}{v} (\mathcal{R}f) (xv) dv \right|$$
$$= \left| \int_{-\infty}^{\infty} \phi(v) (\mathcal{R}f) (xv) dv \right|.$$

Thus, by duality and scaling,

$$\begin{aligned} |\mathcal{R} \circ h_{\Phi}(f)(x)| &\leq \|\phi\|_{\Lambda_{\alpha}} \|\mathcal{R}f(x \cdot)\|_{H^{p}} \\ &= |x|^{-1/p} \|\phi\|_{\Lambda_{\alpha}} \|\mathcal{R}f\|_{H^{p}} \\ &\leq |x|^{-1/p} \|\phi\|_{\Lambda_{\alpha}} \|f\|_{H^{p}} \,, \end{aligned}$$

where the last inequality is true because the Hilbert transform is bounded on  $H^p$  for any p > 0. Now, for any  $\lambda > 0$ , it is easy to see that

$$\begin{aligned} |\{x : |\mathcal{R} \circ h_{\Phi}(f)(x)| > \lambda\}| \\ \leq \left| \{x \in \mathbb{R} : |x|^{-1/p} \|\phi\|_{\Lambda_{\alpha}} \|f\|_{H^{p}} > \lambda\} \right| \end{aligned}$$

$$= \left| \left\{ x \in \mathbb{R} : |x|^{1/p} < \frac{\|\phi\|_{\Lambda_{\alpha}} \|f\|_{H^p}}{\lambda} \right\} \right|$$
  
$$\simeq \left| \left\{ x \in \mathbb{R} : |x| < \left\{ \frac{\|\phi\|_{\Lambda_{\alpha}} \|f\|_{H^p}}{\lambda} \right\}^p \right\} \right|.$$

This shows that for  $\alpha = \frac{1}{p} - 1$ ,

$$\left\|\mathcal{R} \circ h_{\Phi}(f)\right\|_{L^{p,\infty}} \leq \left\|f\right\|_{H^{p}}.$$

Similarly we can show that

$$\|h_{\Phi}(f)\|_{L^{p,\infty}} \le \|f\|_{H^p}$$

if  $\alpha = \frac{1}{p} - 1$ . These two inequalities imply, by the Hilbert transform characterization of  $H^{p,\infty}$ , that

$$||h_{\Phi}(f)||_{H^{p,\infty}(\mathbb{R})} \le ||f||_{H^{p}(\mathbb{R})}.$$

Put

$$p_1 = \frac{1}{\alpha + \epsilon + 1}, \quad p_2 = \frac{1}{\alpha - \epsilon + 1}.$$

By the above weak  $H^p$  estimate, we have that for j = 1, 2,

$$\begin{aligned} \|\mathcal{R} \circ h_{\Phi}(f)\|_{L^{p_{j},\infty}(\mathbb{R})} &\leq \|f\|_{H^{p_{j}}(\mathbb{R})}, \\ \|h_{\Phi}(f)\|_{L^{p_{j},\infty}(\mathbb{R})} &\leq \|f\|_{H^{p_{j}}(\mathbb{R})}. \end{aligned}$$

Then it follows from the Marcinkiewicz interpolation theorem that,

$$\begin{aligned} \|\mathcal{R} \circ h_{\Phi}(f)\|_{L^{p}(\mathbb{R})} &\leq \|f\|_{H^{p}(\mathbb{R})} \\ \|h_{\Phi}(f)\|_{L^{p}(\mathbb{R})} &\leq \|f\|_{H^{p}(\mathbb{R})} \,. \end{aligned}$$

Therefore, we have

$$\|h_{\Phi}(f)\|_{H^{p}(\mathbb{R})} \simeq \|\mathcal{R} \circ h_{\Phi}(f)\|_{L^{p}(\mathbb{R})} + \|h_{\Phi}(f)\|_{L^{p}(\mathbb{R})} \le \|f\|_{H^{p}(\mathbb{R})}.$$

This completes the proof of Theorem 3.1.

We have the following corollary:

**Corollary 3.2.** Suppose  $\alpha > \frac{1}{p} - 1$  and  $\alpha$  is a positive integer. Assume that  $\Phi \in C^{\alpha} \cap L^{1}$  and satisfies

$$\begin{aligned} \chi_{(1,\infty)}(t) \left| \Phi^{(k)}(t) \right| &\leq t^{-2k-1-\alpha} \text{ for all } k = 0, 1, 2, ..., \alpha, \\ \chi_{(0,1)}(t) \left| \Phi^{(k)}(t) \right| &\leq t^{-k-1} \text{ for all } k = 0, 1, 2, ..., \alpha. \end{aligned}$$

Then

$$||h_{\Phi}(f)||_{H^{p}(\mathbb{R})} \leq ||f||_{H^{p}(\mathbb{R})}.$$

*Proof.* It is easy to see that if  $t \leq 1$ ,

$$\left|\frac{d^{\alpha}}{dt^{\alpha}}\left(\frac{\Phi(\frac{1}{t})}{t}\right)\right| \leq \sum_{k=0}^{\alpha} \left|\frac{\Phi^{k}(\frac{1}{t})}{t^{1+2k}}\right| \leq 1;$$

and if t > 1

$$\left|\frac{d^{\alpha}}{dt^{\alpha}}\left(\frac{\Phi(\frac{1}{t})}{t}\right)\right| \leq \sum_{k=0}^{\alpha} \left|\frac{\Phi^{k}(\frac{1}{t})}{t^{\alpha+1}}\right| \leq 1.$$

Hence, the function  $\Phi$  satisfies the conditions of Theorem 3.1.

In the capacity of  $\Phi$  in Theorem 3.1, one may take a number of usual Fourier multipliers, for example,  $e^{-|t|}$ ,  $e^{-t^2/2}$ , which correspond to the Poisson and Gaussian kernels, respectively. It is also worthwhile to investigate the Riesz multiplier

$$\Phi(t) = (1 - t^2)^{\delta}_+,$$

where  $f(t)_+$  is the function that is equal to 0 if  $f(t) \leq 0$  and is equal to f(t) if f(t) > 0. It is easy to check that  $(1 - t^2)^{\delta}_+$  satisfies the conditions in Theorem 1 if  $\delta > \frac{1}{p} - 1$ . We observe that Stein, Taibleson and Weiss in [26] proved that the Bochner-Riesz operator  $B_{\delta} * f$  is bounded on  $H^p(\mathbb{R}^n)$  if  $\delta > \frac{n}{p} - \frac{n+1}{2}$ , where

$$(\widehat{B_{\delta} * f})(\xi) = (1 - |\xi|^2)^{\delta}_{+} \widehat{f}(\xi).$$

Therefore, when n = 1, our result matches the critical index obtained by Stein,

Taibleson and Weiss. With this observation, it will be very interesting to extend Theorem 3.1 to the higher dimensional Hausdorff operator. This generalization is the subject of Chapter 5.

#### CHAPTER 4

## ONE DIMENSIONAL HAUSDORFF OPERATORS: THE CASE p = 1

#### 4.1 INTRODUCTION

This chapter will show that, for  $\Phi(t) \ge 0$ , the Hausdorff operator  $h_{\Phi}$  is bounded on the Hardy space  $H^1(\mathbb{R})$  if and only if  $\Phi$  is a Lebesgue integrable function; and  $h_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R})$  if and only if the function  $\Phi(t)\chi_{(0,1)}(t)\log(\frac{1}{t})$ is Lebesgue integrable. We also establish a weak type  $H^1(\mathbb{R})$  boundedness theorem for  $h_{\Phi}$ . As an application, we conclude that the Hardy operator  $\mathcal{H}$  is not bounded on either  $H^1(\mathbb{R})$  or  $h^1(\mathbb{R})$ , but it is bounded from  $H^1(\mathbb{R})$  to the weak space  $H^{1,\infty}(\mathbb{R})$ . We also study the boundedness property for the *k*th order Hardy operator  $\mathcal{H}_{(k)}$  and fractional Hardy operator  $\mathcal{H}_{(k),\alpha}$  on spaces  $H^p(\mathbb{R})$  for  $k \ge \frac{1}{p} - 1 \ge 0, \ 0 \le \alpha < 1$ .

One operator in particular will serve as a model of Hausdorff operators acting on  $H^1(\mathbb{R})$ . We begin with a discussion of the Hardy operator.

4.1.1 The Hardy Operator and its Generalizations

The Hardy operator  $\mathcal{H}$  is defined by

$$\mathcal{H}f(x) = \frac{1}{x} \int_0^x f(t) \, dt.$$

It is known that the operator  $\mathcal{H}$  is bounded on  $L^p(\mathbb{R})$ , for all 1 with the $best bound constant <math>\frac{p}{p-1}$  (or 1 for  $p = \infty$ ), and  $\mathcal{H}$  is not bounded on  $L^1(\mathbb{R})$ (see [2], also [7]). It is also known that some important operators are bounded on  $H^1(\mathbb{R})$  but not on  $L^1(\mathbb{R})$ . Thus a natural question is whether the Hardy operator is bounded on  $H^1(\mathbb{R})$ .

After a changing of variables, we have

$$\mathcal{H}f(x) = \int_0^\infty \frac{\chi_{(1,\infty)}(t) t^{-1}}{t} f(\frac{x}{t}) dt.$$

Thus,  $\mathcal{H}$  is a Hausdorff operator with  $\Phi = \frac{\chi_{(1,\infty)}(t)}{t}$ .

Two generalizations of the Hardy operator will also be of interest: the kth Hardy operator  $\mathcal{H}_{(k)}$  (k = 0, 1, 2, ...)

$$\mathcal{H}_{(k)}(f)(x) = \frac{1}{x^{k+1}} \int_0^x t^k f(t) dt$$
$$= \int_1^\infty \frac{1}{t^{k+2}} f(\frac{x}{t}) dt$$

and the kth fractional Hardy operator  $\mathcal{H}_{(k),\alpha}$   $(k = 0, 1, 2, ..., 0 \leq \alpha < 1)$ 

$$\mathcal{H}_{(k),\alpha}(f)(x) = \frac{1}{x^{k+1-\alpha}} \int_0^x t^k f(t) \, dt.$$

It is easy to see that  $\mathcal{H}_{(0)} = \mathcal{H}$  and  $\mathcal{H}_{(k),0} = \mathcal{H}_{(k)}$ .

### 4.1.2 A Lemma in Local Hardy Spaces

Recall from that the local Hardy space  $h^1$  is characterized by decomposition into atoms and big blocks. In order to demonstrate a bound on Hausdorff operators on  $h^1$ , it is useful to establish bounds on the  $h^1$  norm of both big and small blocks.

A simple computation shows (or see [6])

$$\|B\|_{h^p} \leqslant 1$$

uniformly for all big (p,q) blocks B. For small blocks, we have the following estimate.

**Lemma 4.1.** For a small (1,q) block b with support in  $(x_0 - r, x_0 + r)$ , we have

$$\|b\|_{h^1} \leqslant 1 + \log \frac{1}{r}$$

uniformly on r and  $x_0$ .

*Proof.* By a change of variables, we may assume that the support of b is in the interval (-r, r). Let B(u) = rb(ru). It is easy to see that supp  $B(u) \subseteq (-1, 1)$  and (after a simple computation)  $||B||_{L^q} \leq 1$ ; so B(u) is a big (1, q)-block. Choose  $\Psi(y) = e^{-|y|^2}$ 

in the definition of  $h^1(\mathbb{R})$ . We have

$$\begin{split} \|b\|_{h^{1}(\mathbb{R})} &= \int_{\mathbb{R}} \sup_{0 < s \leq 1} \left| \int_{\mathbb{R}} \Psi_{s}(x-u)b(u) \ du \right| \ dx \\ &= \int_{\mathbb{R}} \sup_{0 < s \leq 1} \left| \int_{\mathbb{R}} \Psi_{s}(x-ru)rb(ru) \ du \right| \ dx \\ &= \int_{\mathbb{R}} \sup_{0 < s \leq 1} r \left| \int_{\mathbb{R}} \Psi_{s}(r(x-u))B(u) \ du \right| \ dx \\ &= \int_{\mathbb{R}} \sup_{0 < s \leq \frac{1}{r}} \left| \int_{\mathbb{R}} \frac{1}{(s/r)}\Psi\left(\frac{x-u}{s/r}\right)B(u) \ du \right| \ dx \\ &= \int_{\mathbb{R}} \sup_{0 < s \leq \frac{1}{r}} \left| \int_{\mathbb{R}} \Psi_{s}(x-u)B(u) \ du \right| \ dx \end{split}$$

Let

$$N = \left[\log_2 \frac{1}{r}\right] + 1,$$

then (recalling that  $\frac{1}{r} > 1$ )

$$\begin{split} \sup_{0 < s \leq \frac{1}{r}} |\Psi_s * B(x)| &\leq \sup_{0 < s \leq 1} |\Psi_s * B(x)| + \sum_{k=0}^N \sup_{2^k < s \leq 2^{k+1}} |\Psi_s * B(x)| \\ &\leq \sup_{0 < s \leq 1} |\Psi_s * B(x)| + \sum_{k=0}^N \sup_{2^k < s \leq 2^{k+1}} |\Psi_s| * |B(x)|, \\ &\leq \sup_{0 < s \leq 1} |\Psi_s * B(x)| + \sum_{k=0}^N (\tilde{\Psi}_k * |B|)(x) \end{split}$$

where

$$\tilde{\Psi}_k(u) = \frac{1}{2^k} \Psi(\frac{u}{2^{k+1}}).$$

This gives

$$\int_{\mathbb{R}} \sup_{0 < s \leq \frac{1}{r}} |\Psi_s * B(x)| \ dx \leq \left( \left\| \sup_{0 < s < 1} |\Psi_s * B| \right\|_{L^1(\mathbb{R})} + \sum_{k=0}^N \left\| \tilde{\Psi}_k * |B| \right\|_{L^1(\mathbb{R})} \right).$$

It is easy to check that there is a constant  ${\cal C}$  independent of k such that

$$\left\|\tilde{\Psi}_k * |B|\right\|_{L^1(\mathbb{R})} \leqslant C \left\|B\right\|_{L^1(\mathbb{R})} \leqslant C.$$

$$\|b\|_{h^1(\mathbb{R})} \leq N \|B\|_{L^1} + \|B\|_{h^1(\mathbb{R})} \leq 1 + \log_2 \frac{1}{r}.$$

4.2 Necessary and Sufficient Conditions for  $H^1$  Boundedness

In [15, 18], Liflyand and Móricz proved that the Hausdorff operator has the same behavior on the Hardy space  $H^1(\mathbb{R})$  as that in the Lebesgue space  $L^1(\mathbb{R})$ , in the following sense:

**Theorem G** ([15]). If  $\Phi \in L^1(0, \infty)$ , then  $h_{\Phi}$  is bounded on  $H^1(\mathbb{R})$ .

Motivated by the known  $L^p$  results, our first aim is to show that  $\Phi \in L^1(0, \infty)$  is also a necessity condition in Theorem G, if  $\Phi$  is nonnegative valued. Precisely, we establish the following result.

**Theorem 4.2.** Let  $\Phi$  be a nonnegative valued locally integrable function.

- (1)  $h_{\Phi}$  is bounded on  $H^1(\mathbb{R})$  if and only if  $\Phi \in L^1(0, \infty)$ .
- (2)  $h_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R})$  if and only if

$$\int_{1}^{\infty} \Phi(t) dt + \int_{0}^{1} \Phi(t) \left(1 + \log(\frac{1}{t})\right) dt < \infty.$$

Proof of Part 1. To prove the first part of the theorem, by Theorem G, it suffices to show the only if part. Suppose  $\Phi \notin L^1(0, \infty)$ , we use the atomic characterization of the space  $H^1$ . Let b be a  $C^{\infty}$  odd function with support on [-1, 1] which satisfies

$$b(x) = \frac{1}{2}$$
 for  $x \in [\frac{1}{4}, \frac{1}{2}]$ , and  $\frac{1}{2} \ge b(x) \ge 0$  for  $x \in [0, 1]$ .

Then b is a  $(1, \infty, 0)$  atom. Thus,  $b \in H^1$ . Let  $a = -\mathcal{R}(b)$ . Recalling that  $-\mathcal{R}^2 = \mathrm{Id}$ and that the Hilbert transform is bounded on  $H^1$ , we know  $a \in H^1$  and  $b = \mathcal{R}(a)$ . Now

$$\begin{split} \|h_{\Phi}(a)\|_{H^{1}(\mathbb{R})} &\geq \|\mathcal{R}h_{\Phi}(a)\|_{L^{1}(\mathbb{R})} \\ &= \left\|\int_{0}^{\infty} \frac{\Phi(t)}{t} \mathcal{R}a(\frac{x}{t}) dt\right\|_{L^{1}(\mathbb{R}, dx)} \\ &= \left\|\int_{0}^{\infty} \frac{\Phi(t)}{t} b(\frac{x}{t}) dt\right\|_{L^{1}(\mathbb{R}, dx)}. \end{split}$$

The last equality holds because the Hilbert transform commutes with the dilation  $\frac{1}{t}.$  So,

$$\begin{aligned} \|h_{\Phi}(a)\|_{H^{1}(\mathbb{R})} &\geq \int_{0}^{\infty} \left| \int_{0}^{\infty} \frac{\Phi(t)}{t} b(\frac{x}{t}) dt \right| dx \\ &= \int_{0}^{\infty} \Phi(t) dt \int_{0}^{\infty} b(x) dx = \infty. \end{aligned}$$

If  $h_{\Phi}$  were bounded on  $H^1$ , it could be

$$\|h_{\Phi}(a)\|_{H^{1}(\mathbb{R})} \leq \|a\|_{H^{1}} \leq 1,$$

which leads to a contradiction.

Proof of Part 2. To show the second part, we first show sufficiency. For any  $f \in h^1(\mathbb{R})$ , we may write

$$f(x) = \sum_{j} \lambda_j a_j(x) + \sum_{j} \mu_j B_j(x)$$

where each  $a_j$  is a  $(1, \infty, 0)$  atom and each  $B_j$  is a big  $(1, \infty)$  block, and

$$\sum_{j} \left( |\lambda_j| + |\mu_j| \right) \cong \|f\|_{h^1(\mathbb{R})} \,.$$

By the Minkowski inequality,

$$\|h_{\Phi}(f)\|_{h^{1}(\mathbb{R})} \leq \sum_{j} |\lambda_{j}| \|h_{\Phi}(a_{j})\|_{h^{1}(\mathbb{R})} + \sum_{j} |\mu_{j}| \|h_{\Phi}(B_{j})\|_{h^{1}(\mathbb{R})}.$$

Since

$$\frac{a_j(\frac{x}{t})}{t} = A_j(x)$$

is again a  $(1, \infty, 0)$  atom, by the Minkowski inequality

$$\|h_{\Phi}(a_j)\|_{h^1(\mathbb{R})} \leq \int_0^\infty \Phi(t) \|A_j\|_{h^1(\mathbb{R})} dt \leq \int_0^\infty \Phi(t) dt.$$

Similarly,

$$\|h_{\Phi}(B_{j})\|_{h^{1}(\mathbb{R})} \leq \int_{0}^{1} \Phi(t) \left\|\frac{1}{t}B_{j}(\frac{\cdot}{t})\right\|_{h^{1}(\mathbb{R})} dt + \int_{1}^{\infty} \Phi(t) \left\|\frac{1}{t}B_{j}(\frac{\cdot}{t})\right\|_{h^{1}(\mathbb{R})} dt.$$

Note that  $\frac{1}{t}B_j(\frac{x}{t})$  is again a big  $(1, \infty)$  block if  $t \ge 1$ , and it may become a small  $(1, \infty)$  block when t < 1. Thus, by Lemma 4.1 we have

$$\|h_{\Phi}(B_j)\|_{h^1(\mathbb{R})} \leqslant \int_0^1 \Phi(t)(1+\log_2(\frac{1}{t})) dt + \int_1^\infty \Phi(t) dt.$$

Combining all estimates, we obtain

$$\|h_{\Phi}(f)\|_{h^{1}(\mathbb{R})} \leq C \|f\|_{h^{1}(\mathbb{R})},$$

where

$$C = \int_0^1 \Phi(t) (1 + \log_2(\frac{1}{t})) \, dt + \int_1^\infty \Phi(t) \, dt.$$

Conversely, suppose

$$\int_0^1 \Phi(t) (1 + \log_2(\frac{1}{t})) \, dt = \infty.$$

Let  $\Psi_s(x) = \frac{1}{s}e^{\frac{-x^2}{s^2}}$  and  $B(y) = \chi_{[0,\frac{1}{2}]}(y)$ . Then B(y) is a big  $(1,\infty)$  block. We have

$$\int_{\mathbb{R}} \sup_{0 < s \leq 1} |\Psi_s * h_{\Phi}(B)(x)| \, dx \ge \int_0^1 \sup_{0 < s \leq 1} |\Psi_s * h_{\Phi}(B)(x)| \, dx \ge \int_0^1 |\Psi_x * h_{\Phi}(B)(x)| \, dx.$$

Here

$$\begin{aligned} |\Psi_x * h_{\Phi}(B)(x)| &= \int_0^\infty \frac{\Phi(t)}{t} \int_{\mathbb{R}} \frac{1}{x} e^{-(\frac{x-y}{x})^2} B(\frac{y}{t}) \, dy \, dt \\ &\ge \int_0^1 \frac{\Phi(t)}{t} \int_0^{\frac{t}{2}} \frac{1}{x} e^{-(\frac{x-y}{x})^2} \, dy \, dt. \end{aligned}$$

$$\begin{split} \int_{\mathbb{R}} \sup_{0 < s \leq 1} |\Psi_s * h_{\Phi}(B)(x)| \, dx &\geq \int_0^1 \int_0^1 \frac{\Phi(t)}{t} \int_0^{\frac{t}{2}} \frac{1}{x} e^{-(\frac{x-y}{x})^2} \, dy \, dt \, dx \\ &\geq \int_0^1 \frac{\Phi(t)}{t} \left( \int_t^1 \int_0^{\frac{t}{4}} \frac{1}{x} e^{-(\frac{x-y}{x})^2} \, dy \, dx \right) \, dt \\ &\geq \frac{e^{-4}}{4} \int_0^1 \Phi(t) \left( \int_t^1 \frac{1}{x} \, dx \right) \, dt \\ &\cong -\frac{e^{-4}}{4} \int_0^1 \Phi(t) \log_2(t) \, dt = \infty. \end{split}$$

On the other hand, if  $h_{\Phi}(f)$  were bounded on  $h^1(\mathbb{R})$ , it would be

$$\int_{\mathbb{R}} \sup_{0 < s \leq 1} |\Psi_s * h_{\Phi}(B)(x)| \, dx = \|h_{\Phi}(B)\|_{h^1(\mathbb{R})} \leq \|B\|_{h^1(\mathbb{R})} \leq 1.$$

which leads to a contradiction. Next, assume

$$\int_{1}^{\infty} \Phi(t) \ dt = \infty,$$

and let B be as before. We have

$$\int_{\mathbb{R}} \sup_{0 < s \leq 1} |\Psi_s * h_{\Phi}(B)(x)| \, dx \ge \int_{\mathbb{R}} |\Psi_1 * h_{\Phi}(B)(x)| \, dx$$
$$\ge \int_1^\infty \frac{\Phi(t)}{t} \int_0^{\frac{t}{2}} \left( \int_{\mathbb{R}} e^{-(x-y)^2} \, dx \right) \, dy \, dt$$
$$\ge \int_1^\infty \Phi(t) \, dt = \infty,$$

so again  $h_{\Phi}$  is unbounded.

Our second aim is to establish conditions for  $H^1(\mathbb{R}) \to H^{1,\infty}(\mathbb{R})$  boundedness. To do this, we use an approach similar to that in Chapter 3. We rewrite the Hausdorff operator as

$$h_{\Phi}f(x) = \int_{-\infty}^{\infty} \frac{\Phi_0(t)}{t} f\left(\frac{x}{t}\right) dt,$$

where  $\Phi_0(t)$  is a function that is equal to  $\Phi$  on  $(0, \infty)$  and is equal to 0 on  $(-\infty, 0)$ .

Denote

$$\phi(t) = \frac{\Phi_0(\frac{1}{t})}{t},$$

Then we have the following result:

**Theorem 4.3.** Let  $\phi(t) = t^{-1}\Phi_0(t)$ . Suppose  $\phi \in BMO(\mathbb{R})$ . Then  $h_{\Phi}$  extends to a bounded operator from  $H^1(\mathbb{R})$  to  $H^{1,\infty}(\mathbb{R})$ .

*Proof.* It suffices to show the theorem for functions f in the space  $H^1 \cap L^2$ , since this space is dense in  $H^1$ .

By Theorem 3 in [17],

$$\mathcal{R} \circ h_{\Phi}(f)(x) = h_{\Phi}(\mathcal{R}f)(x) = \int_{-\infty}^{\infty} \frac{\Phi_0(t)}{t} (\mathcal{R}f)(\frac{x}{t}) dt.$$

Changing variables  $\frac{1}{t} = v$ , we have

$$\begin{aligned} |\mathcal{R} \circ h_{\Phi}(f)(x)| &= \left| \int_{-\infty}^{\infty} \frac{\Phi_0(\frac{1}{v})}{v} (\mathcal{R}f)(xv) \, dv \right| \\ &= \left| \int_{-\infty}^{\infty} \phi(v) (\mathcal{R}f)(xv) \, dv \right|. \end{aligned}$$

Thus, by duality and scaling,

$$\begin{aligned} |\mathcal{R} \circ h_{\Phi}(f)(x)| &\leq \|\phi\|_{\text{BMO}} \, \|(\mathcal{R}f)(x \cdot)\|_{H^{1}} \\ &= |x|^{-1} \, \|\phi\|_{\text{BMO}} \, \|(\mathcal{R}f)\|_{H^{1}} \\ &\leq |x|^{-1} \, \|\phi\|_{\text{BMO}} \, \|f\|_{H^{1}} \, . \end{aligned}$$

The last inequality is because that the Hilbert transform is bounded on  $H^1$ . Note, for any  $\lambda > 0$ , it is easy to see that

$$\begin{aligned} &|\{x \in \mathbb{R} : |\mathcal{R} \circ h_{\Phi}(f)(x)| > \lambda\}| \\ &\leq |x \in \mathbb{R} : |x|^{-1} \|\phi\|_{\text{BMO}} \|f\|_{H^{1}} > \lambda| \\ &= \left|\left\{x \in \mathbb{R} : |x| \leq \frac{\|\phi\|_{\text{BMO}} \|f\|_{H^{1}}}{\lambda}\right\}\right| \\ &\cong \frac{\|\phi\|_{\text{BMO}} \|f\|_{H^{1}}}{\lambda}. \end{aligned}$$

So by the definition,

$$\left\|\mathcal{R} \circ h_{\Phi}(f)\right\|_{L^{1,\infty}} \leqslant \left\|\phi\right\|_{\mathrm{BMO}} \left\|f\right\|_{H^{1}}$$

Similarly, we can show

$$\|h_{\Phi}(f)\|_{L^{1,\infty}} \leq \|\phi\|_{BMO} \|f\|_{H^1}.$$

These two inequalities imply

$$\|h_{\Phi}(f)\|_{H^{1,\infty}} \leq \|f\|_{H^{1}(\mathbb{R})}. \qquad \Box$$

# 4.3 Boundedness of the Hardy Operator Near $H^1$

So far we have considered Hardy spaces defined on the whole real line  $\mathbb{R}$ . If we wish to consider a Hardy space only on the half line  $\mathbb{R}_+ = (0, \infty)$ , we might use the atomic Hardy space  $H^1_A(\mathbb{R}_+)$  studied by Coifman and Weiss (see [4]).

The space  $\mathbb{R}_+ = (0, \infty)$  is a space of homogeneous type. Based on the study by Coifman and Weiss [4], we consider the atomic Hardy space  $H^1_A(\mathbb{R}_+)$ . Applying the definition of a  $(1, \infty, 0)$  atom a(x) (or 1-atom, for the sake of brevity) to the current case, we obtain

Support Condition

supp  $(a) \subset (\alpha, \beta) \subset (0, \infty)$ .

Cancellation Condition

$$\int_{\mathbb{R}_+} a(y) \, dy = 0.$$

Size Condition

$$\|a\|_{L^{\infty}} \leq (\beta - \alpha)^{-1}.$$

The space  $H^1_A(\mathbb{R}_+)$  is the collection of all  $f \in L^1(\mathbb{R}_+)$  such that

$$f(x) = \sum \lambda_j a_j(x),$$

where each  $a_j$  is a 1-atom and  $\{\lambda_j\} \in \ell^1$ . We define

$$\|f\|_{H^1_A(\mathbb{R}_+)} = \inf\left\{\sum |\lambda_j| : f(x) = \sum \lambda_j a_j(x)\right\}.$$

It is a little surprising to note that the Hardy operator  $\mathcal{H}$  is bounded on  $H^1_A(\mathbb{R}_+)$ .

**Theorem 4.4.** For the Hardy operator  $\mathcal{H}$ 

$$\|\mathcal{H}(f)\|_{H^1_A(\mathbb{R}_+)} \leqslant \|f\|_{H^1_A(\mathbb{R}_+)}.$$

*Proof.* We need only show that, for any 1-atom a,  $\mathcal{H}(a)$  is also a 1-atom. We proceed by showing that  $\mathcal{H}(a)$  satisfies the three conditions.

Support Condition

Suppose a has support in  $(\alpha, \beta)$ . We show that if  $x \notin (\alpha, \beta)$ , then  $\mathcal{H}(a)(x) = 0$ , so supp  $(\mathcal{H}(a)) \subseteq (\alpha, \beta)$ . Recall

$$\mathcal{H}(a)(x) = \frac{1}{x} \int_0^x a(t) \, dt.$$

Now, if  $x \leq \alpha$ , then a(t) = 0 for all  $t \in (0, x)$ , so  $\mathcal{H}(a)(x) = \frac{1}{x} \int_0^x a(t) dt = 0$ . On the other hand, of  $x \geq \beta$ , then a(t) = 0 for t > x, and so

$$\mathcal{H}(a)(x) = \frac{1}{x} \int_0^x a(t) dt$$
$$= \frac{1}{x} \int_0^\infty a(t) dt =$$

0

with the last equality holding by the cancellation condition.

Size Condition

This is easily seen by a computation.

$$\|\mathcal{H}(a)\|_{L^{\infty}} \leq \frac{1}{x} \int_{0}^{x} \|a\|_{L^{\infty}} dy = \|a\|_{L^{\infty}} \leq \frac{1}{\beta - \alpha}.$$

Cancellation Condition

An easy computation with the support condition on  $\mathcal{H}(a)$  gives

$$\int_0^\infty \mathcal{H}(a)(x) \, dx = \int_\alpha^\beta \int_1^\infty \frac{1}{t^2} a\left(\frac{x}{t}\right) \, dt \, dx.$$

Since the above double integral is well defined, by the Fubini theorem we obtain

$$\int_0^\infty \mathcal{H}(a)(x) \, dx = \int_1^\infty \frac{1}{t^2} \int_0^\infty a(\frac{x}{t}) \, dx \, dt$$
$$= \lim_{N \to \infty} \int_1^N \frac{1}{t^2} \int_0^\infty a(\frac{x}{t}) \, dx \, dt$$
$$= \lim_{N \to \infty} \int_1^N \frac{1}{t} \left( \int_0^\infty a(x) \, dx \right) \, dt = 0.$$

Thus  $\mathcal{H}(a)$  is a 1-atom, and so for  $f \in H^1_A(0,\infty)$ ,

$$\mathcal{H}(f) = \mathcal{H}\left(\sum \lambda_j a_j(x)\right) = \sum \lambda_j \mathcal{H}(a_j)(x)$$

expresses  $\mathcal{H}(f)$  as a sum of atoms, and so

$$\|\mathcal{H}(f)\|_{H^{1}_{A}(0,\infty)} \leq \sum |\lambda_{j}| = \|f\|_{H^{1}_{A}(0,\infty)}$$

which is the desired bound.

In [5], García-Cuerva, and Rubio De Francia discuss a connection between  $H^1_A(\mathbb{R}_+)$ and even elements of  $H^1(\mathbb{R})$  (see [5, lemmas 7.39 and 7.40] especially.) This suggests that the following result is a corollary to Theorem 4.4, and indeed, this is what we see.

Corollary 4.5. Let

$$f_e(x) = \frac{1}{2} \{ f(x) + f(-x) \}$$

be the even part of f. If  $f \in H^1$ , then we have

$$\|\mathcal{H}(f_e)\|_{H^1} \leq \|f\|_{H^1}$$
.

Particularly, we have

$$\|\mathcal{H}(f)\|_{H^1} \leqslant \|f\|_{H^1}$$

for all even functions  $f \in H^1$ .

*Proof.* We write

$$f(x) = \sum \lambda_j a_j(x)$$

with each  $a_j$  being a  $(1, \infty, 0)$  atom and

$$\sum |\lambda_j| \cong \|f\|_{H^1}.$$

As in the proof of Lemma 7.39 on page 353 in [5], we may write

$$f_e(x) = \sum \lambda_j A_j(x)$$

where

$$A_j(x) = \frac{a_j(x) + a_j(-x)}{2}$$

If  $a_j(x)$  has support in  $[0, \infty)$ , then  $a_j^-(x) = a_j(-x)$  has support in  $(-\infty, 0]$ . Following the proof of Theorem 4.4, we know both  $\mathcal{H}(a_j)$  and  $\mathcal{H}(a_j^-)$  are  $(1, \infty, 0)$  atoms so that

$$\|\mathcal{H}(A_j)\|_{H^1} \leq 1.$$

If 0 is interior to the interval-support  $(\alpha_j, \beta_j)$  of  $a_j$ , then without loss of generality, we assume  $\beta_j > |\alpha_j|$ . Thus  $A_j$  is supported in the interval  $(-\beta_j, \beta_j)$ , and

$$\frac{\beta_j - \alpha_j}{2} \leqslant \beta_j \leqslant \beta_j - \alpha_j.$$

On the other hand, since  $A_j$  is an even function, with the cancellation condition we have

$$0 = \int_{-\beta_j}^{\beta_j} A_j(x) \, dx = 2 \int_0^{\beta_j} A_j(x) \, dx = 2 \int_{-\beta_j}^0 A_j(x) \, dx$$

This indicates that, without loss of generality, both  $\chi_{(0,\infty)}(x)A_j(x)$  and  $\chi_{(-\infty,0)}(x)A_j(x)$ are  $(1, \infty, 0)$  atoms. Thus, again we have

$$\|\mathcal{H}(A_j)\|_{H^1} \leq \|\mathcal{H}(\chi_{(-\infty,0)}A_j)\|_{H^1} + \|\mathcal{H}(\chi_{(0,\infty)}A_j)\|_{H^1} \leq 1.$$

#### **Corollary 4.6.** The Hardy operator $\mathcal{H}$ has the following properties

- (1)  $\mathcal{H}$  is not bounded on  $H^1(\mathbb{R})$ .
- (2)  $\mathcal{H}$  is not bounded on  $h^1(\mathbb{R})$ .
- (3)  $\mathcal{H}$  is bounded from  $H^1(\mathbb{R})$  to  $H^{1,\infty}(\mathbb{R})$ .
- (4)  $\mathcal{H}$  is bounded on  $H^1_A(0,\infty)$ .

Proof. Recalling that  $\mathcal{H} = h_{\Phi}$  with  $\Phi = \frac{\chi_{(1,\infty)}(t)}{t}$ , and noting that  $\Phi \notin L^1(0,\infty)$ , we see parts (1) and (2) from Theorem 4.2. However, noting that in this case  $\phi(t) = \chi_{(0,1)}(t) \in BMO$ , Theorem 4.3 gives part (3). Part (4) is the result in Theorem 4.4, above.

4.4 Generalizations of the Hardy Operator (0

We now turn to the kth order Hardy operator

$$\mathcal{H}_{(k)}(f)(x) = \frac{1}{x^{k+1}} \int_0^x t^k f(t) dt,$$

**Theorem 4.7.** Let 0 .

- (1)  $\mathcal{H}_{(k)}$  is bounded from  $H^p(\mathbb{R})$  to  $H^{p,\infty}(\mathbb{R})$  if  $k = \frac{1}{p} 1$ .
- (2)  $\mathcal{H}_{(k)}$  is bounded on  $H^p(\mathbb{R})$  if  $k > \frac{1}{p} 1$ .

*Proof.* We begin by showing the first part. Then, with the easy fact that  $\mathcal{H}_{(k)}$  is weakly bounded on  $H^1$  (by Theorem 4.3), we complete the proof of the theorem by using an interpolation argument.

Choose p such that  $\frac{1}{p} - 1 = k$ . It suffices to show that  $\|\mathcal{H}_{(k)}(f)\|_{H^{p,\infty}} \leq \|f\|_{H^p}$  for  $f \in H^p \cap L^2$ , since  $H^p \cap L^2$  is dense in  $H^p$ . Using the Hilbert transform, we can show this by showing

$$\left\|\mathcal{R} \circ \mathcal{H}_{(k)}(f)\right\|_{L^{p,\infty}} + \left\|\mathcal{H}_{(k)}(f)\right\|_{L^{p,\infty}} \leq \left\|f\right\|_{H^{p}}.$$

We show  $\|\mathcal{R} \circ \mathcal{H}_{(k)}(f)\|_{L^{p,\infty}} \leq \|f\|_{H^p}$ ; the proof that  $\|\mathcal{H}_{(k)}(f)\|_{L^{p,\infty}} \leq \|f\|_{H^p}$  is the same.

As in the proof of Theorem 4.3, the Hilbert Transform satisfies

$$\mathcal{R} \circ \mathcal{H}_{(k)}(f)(x) = \mathcal{H}_{(k)}(\mathcal{R}f)(x) = \int_{1}^{\infty} \frac{1}{t^{k+2}}(\mathcal{R}f)(\frac{x}{t}) dt.$$

For any  $f \in H^p(\mathbb{R})$ ,  $\mathcal{R}f \in H^p$  since the Hilbert transform is bounded on  $H^p$ . Thus, we have an atomic decomposition

$$\mathcal{R}f = \sum_{j} \lambda_j a_j,$$

where each  $a_j$  is a  $(p, \infty, k)$  atom and

$$\sum |\lambda_j|^p \cong \|\mathcal{R}f\|_{H^p}^p.$$

Hence

$$\mathcal{R} \circ \mathcal{H}_{(k)}(f) = \sum_{j \in \mathbb{Z}} \lambda_j \int_1^\infty \frac{1}{t^{k+2}} a_j(\dot{t}) dt = \sum \lambda_j \mathcal{H}_{(k)}(a_j),$$

and therefore,

$$\left\|\mathcal{R}\circ\mathcal{H}_{(k)}(f)\right\|_{L^{p,\infty}}^{p}=\left\|\sum\lambda_{j}\mathcal{H}_{(k)}(a_{j})\right\|_{L^{p,\infty}}^{p}$$

By [26, Lemma 1.8], showing that

$$\left\|\sum \lambda_j \mathcal{H}_{(k)}(a_j)\right\|_{L^{p,\infty}}^p \leqslant \sum |\lambda_j|^p$$

is equivalent to show that

$$\left\|\mathcal{H}_{(k)}(a_j)\right\|_{L^{p,\infty}}^p \leqslant 1$$

for any  $(p, \infty, k)$ -atom  $a_j$ . Since  $\sum |\lambda_j|^p \cong ||\mathcal{R}f||_{H^p}^p \leq ||f||_{H^p}^p$ , this will give the desired result.

If  $a_j$  is a  $(p, \infty, k)$  atom with support in  $(\alpha_j, \beta_j) \subset (0, \infty)$  or  $(\alpha_j, \beta_j) \subset (-\infty, 0)$ , then by a method similar to that in the proof of Theorem 4.4, it easy to see that  $\mathcal{H}_{(k)}(a_j)$  is again a  $(p, \infty, k)$  atom with support in  $(\alpha_j, \beta_j)$ . Thus, we have

$$\left\|\mathcal{H}_{(k)}(a_j)\right\|_{H^{p,\infty}} \leqslant \left\|\mathcal{H}_{(k)}(a_j)\right\|_{H^p} \leqslant 1$$

uniformly on  $j \in \mathbb{Z}$ .

If the support of a  $(p, \infty, k)$  atom  $a_j$  contains the origin, without loss of generality, we may assume

$$\operatorname{supp}(a_j) \subset (-r_j, r_j)$$

and

$$\|a_j\|_{L^{\infty}} \leqslant r_j^{-\frac{1}{p}}.$$

For each fixed  $x \neq 0$ , the support of  $a_j(tx)$  is in  $\left(-\frac{r_j}{|x|}, \frac{r_j}{|x|}\right)$ . Write

$$\mathcal{H}_{(k)}(a_j)(x) = \frac{1}{|x|^{\frac{1}{p}}} \int_{-\infty}^{\infty} \chi_{(0,1)}(t) \left( |x|^{\frac{1}{p}} t^k \chi_{(-\frac{r_j}{|x|}, \frac{r_j}{|x|})}(t) a_j(tx) \right) dt.$$

We now claim that, for  $x \neq 0$ , the function

$$A_{j,x}(t) = |x|^{\frac{1}{p}} t^k \chi_{(-\frac{r_j}{|x|}, \frac{r_j}{|x|})}(t) a_j(tx)$$

is a  $(1, \infty, 0)$  atom supported in the interval  $\left(-\frac{r_j}{|x|}, \frac{r_j}{|x|}\right)$ . In fact, the support condition is obvious from the definition. Also,

$$\int_{-\infty}^{\infty} A_{j,x}(t) dt \cong \int_{-r_j}^{r_j} t^k a_j(t) dt = 0,$$

by the cancellation condition on  $a_j$ , and

$$||A_{j,x}||_{L^{\infty}} \leq ||a||_{L^{\infty}} \left|\frac{r_j}{x}\right|^k |x|^{\frac{1}{p}} \leq \left|\frac{x}{r_j}\right|^k$$

This shows that  $A_{j,x}$  also satisfies the cancellation and size conditions. Thus, by the duality,

$$\left|\mathcal{H}_{(k)}(a_j)(x)\right| \leq \frac{1}{|x|^{\frac{1}{p}}} \left\|\chi_{(0,1)}\right\|_{\text{BMO}} \left\|A_{j,x}\right\|_{H^1} \leq \frac{1}{|x|^{\frac{1}{p}}}$$

uniformly on j and  $x \in \mathbb{R} \setminus \{0\}$ . Following the proof of Theorem 4.3 we obtain that, for any  $\lambda > 0$ ,

$$\left|\left\{x: \left|\mathcal{H}_{(k)}(a_j)(x)\right| > \lambda\right\}\right| \leq \frac{1}{\lambda^p}.$$

Thus, by the same proof as in Theorem 4.3, we obtain that

$$\left\|\mathcal{H}_{(k)}(a_j)(x)\right\|_{L^{p,\infty}} \leq 1.$$

as desired.

Now we show part (2). Given k,  $0 such that <math>k > \frac{1}{p} - 1$ , we fix  $p_0$  such that  $k = \frac{1}{p_0} - 1$ . Then  $p_0 . The above argument shows that <math>\mathcal{H}_k$  is weakly bounded on  $H^{p_0}$ , and Theorem 4.3 shows that  $\mathcal{H}_k$  is weakly bounded on  $H^1$ . Thus by the Marcinkiewicz interpolation, we obtain the result in part (2).

*Remark* 4.8. The first part of Theorem 4.7 can also be proved by using the result in [14].

This result can be further generalized to the kth order fractional Hardy operators

$$\mathcal{H}_{(k),\alpha}(f)(x) = \frac{1}{x^{k+1-\alpha}} \int_0^x t^k f(t) \, dt, \quad 0 \le \alpha < 1.$$

We first establish an easy  $L^p \to L^q$  estimate.

**Lemma 4.9.** Choose q so that  $\frac{1}{p} - \alpha = \frac{1}{q}$ .

- (1) For  $1 \leq p < \infty$ ,  $\left\| \mathcal{H}_{(k),\alpha}(f) \right\|_{L^{q,\infty}} \leq \|f\|_{L^p}$ .
- (2) For  $1 , <math>\|\mathcal{H}_{(k),\alpha}(f)\|_{L^q} \leq \|f\|_{L^p}$ .

*Proof.* Clearly, we only need to show the weak boundedness. The strong  $L^p \to L^q$  boundedness then follows by an interpolation. By Hölder's inequality,

$$\left|\mathcal{H}_{(k),\alpha}(f)(x)\right| = |x|^{\alpha} \left| \int_{0}^{1} t^{k} f(xt) \, dt \right| \leq \frac{1}{|x|^{\frac{1}{q}}} \|f\|_{L^{p}}.$$

Hence, for any  $\lambda > 0$ ,

$$\left|\left\{x \in \mathbb{R} : \left|\mathcal{H}_{(k),\alpha}(f)(x)\right| \ge \lambda\right\}\right| \le \left(\frac{\|f\|_{L^p}}{\lambda}\right)^q$$

This proves the weak  $L^p \to L^{q,\infty}$  boundedness.

**Theorem 4.10.** Let  $0 , and <math>\frac{1}{p} - \alpha = \frac{1}{q}$ .

- (1)  $\mathcal{H}_{(k),\alpha}$  is bounded from  $H^p(\mathbb{R})$  to  $L^{q,\infty}(\mathbb{R})$  if  $k = \frac{1}{p} 1$ .
- (2)  $\mathcal{H}_{(k),\alpha}$  is bounded from  $H^p(\mathbb{R})$  to  $L^q(\mathbb{R})$  if  $k > \frac{1}{p} 1$ .

*Proof.* With Lemma 4.9 and interpolation, it suffices to show the first part of the theorem. For  $\frac{1}{p} = k + 1$  and any  $f \in H^p(\mathbb{R})$ , as in the proof of Theorem 4.7, we write an atomic decomposition

$$f = \sum_{j} \lambda_j a_j,$$

where each  $a_j$  is a  $(p, \infty, k)$  atom. Thus,

$$\mathcal{H}_{(k),\alpha}(f) = \sum_{j \in \mathbb{Z}} \lambda_j \mathcal{H}_{(k),\alpha}(a_j) \,.$$

Similar to the proof of Theorem 4.7, we only need to show

$$\left\|\mathcal{H}_{(k),\alpha}\left(a_{j}\right)\right\|_{L^{q,\infty}} \leq 1,$$

uniformly for all  $(p, \infty, k)$  atoms  $a_j$ .

If  $a_j$  is  $(p, \infty, k)$  atom with support in  $(\alpha_j, \beta_j) \subset (0, \infty)$  or  $(\alpha_j, \beta_j) \subset (-\infty, 0)$ ,  $\mathcal{H}_{(k),\alpha}(a_j)$  is also supported in  $(\alpha_j, \beta_j) = I_j$ . Choose p', q' > 1 so that  $\frac{1}{p'} - \alpha = \frac{1}{q'}$ . Then we have the following bound uniformly on j.

$$\begin{aligned} \left\| \mathcal{H}_{(k),\alpha}(a_j) \right\|_{L^q}^q &= \left\| \mathcal{H}_{(k),\alpha}(a_j)^q \cdot \chi_{I_j} \right\|_{L^1} \\ &\leqslant \left\| \mathcal{H}_{(k),\alpha}(a_j)^q \right\|_{L^{\frac{q'}{q}}} \cdot \left\| \chi_{I_j} \right\|_{L^{\frac{q'}{q'-q}}} \\ &= \left( \int \left| \mathcal{H}_{(k),\alpha}(a_j)(x) \right|^{q \cdot \frac{q'}{q}} dx \right)^{\frac{1}{q'} \cdot q} |I_j|^{1-\frac{q}{q'}} \end{aligned}$$

$$= \left\| \mathcal{H}_{(k),\alpha}(a_j) \right\|_{L^{q'}}^q \left| I_j \right|^{1-\frac{q}{q'}} \\ \leqslant \left\| a_j \right\|_{L^{p'}}^q \left| I_j \right|^{1-\frac{q}{q'}} \leqslant 1.$$

If the support of a  $(p, \infty, k)$  atom  $a_j$  contains the origin, with an easy modification of the proof for Theorem 4.7, we obtain

$$\left\|\mathcal{H}_{(k),\alpha}\left(a_{j}\right)\right\|_{L^{q,\infty}} \leq 1,$$

uniformly for all  $(p, \infty, k)$  atom  $a_j$ . This completes the proof of part (1).

To show part (2), take  $p_1 = 1$ ,  $p_2$  such that  $p_2 = \frac{1}{1+k}$ , and  $q_1, q_2$  such that  $\frac{1}{q_i} = \frac{1}{p_i} - \alpha \ (i = 1, 2)$ . Then by part (1),  $\|\mathcal{H}_{(k),\alpha}(f)\|_{L^{q_2,\infty}} \leq \|f\|_{H^{p_2}}$ , and by Lemma 4.9,  $\|\mathcal{H}_{(k),\alpha}(f)\|_{L^{q_1,\infty}} \leq \|f\|_{H^1}$ . Applying the Marcinkiewicz interpolation in this case gives  $\|\mathcal{H}_{(k),\alpha}(f)\|_{L^q} \leq \|f\|_{H^p}$ , as desired.

Remark 4.11. We do not proceed to show  $H^p \to H^q$  boundedness because  $\mathcal{H}_{(k),\alpha}$  fails to commute with the Hilbert transform.

#### CHAPTER 5

## HIGH DIMENSIONAL HAUSDORFF OPERATORS

In this chapter, we study two extensions of the Hausdorff operator in  $\mathbb{R}^n$ . For one, we obtain a sufficient and necessary condition for its boundedness on the real Hardy space  $H^1(\mathbb{R}^n)$ . For the other, we study its boundedness on the real Hardy space  $H^p(\mathbb{R}^n)$  for 0 .

Recall the definition of the one dimensional Hausdorff operator: Let  $\Phi$  be a locally integrable function on the positive real line. The one dimensional Hausdorff operator  $h_{\Phi}$  with the generating function  $\Phi$  is defined in the integral form by

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(t)}{t} f(\frac{x}{t}) dt,$$

where, for simplicity, we initially assume that the operator  $h_{\Phi}$  is defined on the class of all Schwartz functions f. For positive values x, a change of variables gives an equivalent form of  $h_{\Phi}$  by

$$h_{\Phi}(f)(x) = \int_0^\infty \frac{\Phi(\frac{x}{t})}{t} f(t) dt.$$

This suggests two different extensions of the Hausdorff operator on high dimensional space,

$$\tilde{H}_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^{n-\beta}} f(\frac{x}{|y|}) \, dy, \quad n > \beta \ge 0$$

and

$$H_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(\frac{x}{|y|})}{|y|^{n-\beta}} f(y) \, dy, \quad n > \beta \ge 0,$$

where  $\Phi$  is a locally integrable function. Each of these definitions gives a so-called fractional Hausdorff operator. We denote

$$\tilde{H}_{\Phi,0} = \tilde{H}_{\Phi}, \qquad H_{\Phi,0} = H_{\Phi}.$$

When  $\Phi(x) = \Phi(|x|)$ , and  $\Phi(t) = \chi_{(1,\infty)}(t)t^{-n+\beta}$ ,  $H_{\Phi,\beta}$  is the fractional Hardy operator if  $0 < \beta < n$ , and  $H_{\Phi,\beta}$  is the Hardy operator if  $\beta = 0$  (see [2]).

By the Minkowski inequality and a scaling argument, it is easy to see that  $H_{\Phi}$ is bounded on the Lebesgue space  $L^p(\mathbb{R}^n)$  for all  $1 \leq p \leq \infty$ , if  $\Phi$  satisfies the size condition

$$\int_{\mathbb{R}^n} |\Phi(y)| \, |y|^{-n+\frac{n}{p}} \, dy < \infty.$$

Similarly,  $\tilde{H}_{\Phi}$  is bounded on the real Hardy space  $H^1(\mathbb{R}^n)$  if  $\Phi$  is Lebesgue integrable (see [10, 12, 15, 18]). In [13], Liflyand posed an open question (among others) to establish the sharpness of this condition on  $\Phi$  to assure the  $H^1(\mathbb{R}^n)$  boundedness for  $\tilde{H}_{\Phi}$ . Motivated by his question, the first aim of this chapter is to solve the problem by showing that, for  $\Phi \ge 0$ ,  $\tilde{H}_{\Phi}$  is bounded on the real Hardy space  $H^1(\mathbb{R}^n)$  if and only if  $\Phi$  is a Lebesgue integrable function.

In Chapter 3, we established some results which used duality to provide a smoothness condition on  $\Phi$  to ensure boundedness of  $h_{\Phi}$  on  $H^p$ . The second aim of this chapter is to apply this technique to give an  $H^p(\mathbb{R}^n) \to L^q(\mathbb{R}^n)$  boundedness result for  $H_{\Phi,\beta}$  in the case where  $\Phi$  is a radial function. We also establish boundedness on  $L^p(\mathbb{R}^n, |x|^{\gamma} dx) \to L^q(\mathbb{R}^n, |x|^{\gamma} dx)$ . With these results, we naturally expect to establish an  $H^p(\mathbb{R}^n) \to H^q(\mathbb{R}^n)$  boundedness theorem for the operator  $H_{\Phi,\beta}(f)$ . However, by checking the proofs of the theorems in the earlier chapter for the one dimensional Hausdorff operator, we find that all methods fail to establish a sufficient condition for  $H^p(\mathbb{R}^n) \to H^q(\mathbb{R}^n)$  boundedness for the operator  $H_{\Phi,\beta}$ . Also, in the high dimensional case, we do not find in the literature any high dimensional  $H^p(\mathbb{R}^n) \to H^q(\mathbb{R}^n), 0 , boundedness results for the (fractional) Hausdorff$  $operator, although the <math>H^1(\mathbb{R}^n) \to H^1(\mathbb{R}^n)$  boundedness has been studied extensively (see [10, 12, 16, 18, 17, 16, 24, 30]). Hence, as the third purpose of this chapter, we will establish a sufficient condition for the  $H^p(\mathbb{R}^n) \to H^q(\mathbb{R}^n)$  boundedness for  $H_{\Phi,\beta}$ . 5.1  $H^1(\mathbb{R}^n) \to H^1(\mathbb{R}^n)$  Boundedness of  $\tilde{H}_{\Phi}$ 

**Theorem 5.1.** Let  $\Phi$  be a nonnegative valued locally integrable function.

- (1)  $\tilde{H}_{\Phi}$  is bounded on  $H^1(\mathbb{R}^n)$  if and only if  $\Phi \in L^1(\mathbb{R}^n)$ .
- (2)  $\tilde{H}_{\Phi}$  is bounded on the local Hardy space  $h^1(\mathbb{R}^n)$  if and only if

$$\int_{|y|\ge 1} \Phi(y) \, dy + \int_{|y|\le 1} \Phi(y) \left(1 + \log\left(\frac{1}{|y|}\right)\right) \, dy < \infty.$$

*Proof.* The "if" part of (1) and (2) was proved in [24] and [1], respectively. We need only show the "only if" part. Let a be a function with support on  $[-1,1]^n$  which satisfies.

(i)  $a(x) = \frac{1}{2} \text{ for } x \in [\frac{1}{4}, \frac{1}{2}]^n.$ 

(ii) 
$$\frac{1}{2} \ge a(x) \ge 0$$
 for  $\sum_{i=1}^{n} x_i \ge 0$ .

(iii) 
$$a(x) = -a(-x)$$
 if  $\sum_{i=1}^{n} x_i \leq 0$ .

Clearly,

$$\int_{\mathbb{R}^n} a(x) \ dx = 0.$$

Thus, a is a  $(1, \infty, 0)$  atom (see [28] or [26]). By the same easy computation as in the one-dimensional case we know

$$\|a\|_{H^1(\mathbb{R}^n)} \leqslant 1.$$

Suppose  $\Phi \notin L^1$ . If  $\tilde{H}_{\Phi}$  were bounded on  $H^1$ , we have,

$$\left\|\tilde{H}_{\Phi}(a)\right\|_{H^{1}} \leqslant \|a\|_{H^{1}} \leqslant 1.$$

On the other hand, by the Riesz transform characterization of  $H^1$ , we have

$$\left\|\tilde{H}_{\Phi}(a)\right\|_{H^{1}} \geq \left\|\tilde{H}_{\Phi}(a)\right\|_{L^{1}}$$

$$\geq \int_{\sum x_i \ge 0} \left| \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^n} a(\frac{x}{|y|}) \, dy \right| \, dx$$
  
$$= \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^n} \left( \int_{\sum x_i \ge 0} a(\frac{x}{|y|}) \, dx \right) \, dy$$
  
$$= \int_{\mathbb{R}^n} \Phi(y) \left( \int_{\sum x_i \ge 0} a(x) \, dx \right) \, dy = \infty.$$

This leads to a contradiction.

Next, we show the "only if" part of (2). Suppose

$$\int_{|y|<1} \Phi(y)(1-\log_2|y|) \, dy = \infty.$$

For notational simplicity, use the  $\ell^{\infty}$  norm on vectors, i.e.  $|x| = \max\{|x_i|\}$ . Let  $\Psi_s = \frac{1}{s^n} e^{-\frac{|x|^2}{s^2}}$  and  $B(x) = \chi_{[0,\frac{1}{2}]^n}(x)$ . An easy computation similar to that for one dimension shows

$$\|B\|_{h^1(\mathbb{R}^n)} \leqslant 1.$$

On the other hand,

$$\begin{split} \left\| \tilde{H}_{\Phi}(B) \right\|_{h^{1}(\mathbb{R}^{n})} &= \int_{\mathbb{R}^{n}} \sup_{0 < s \leq 1} \left| \Psi_{s} * \tilde{H}_{\Phi}(B)(x) \right| dx \\ &\geq \int_{[0,1]^{n}} \sup_{0 < s \leq 1} \left| \Psi_{s} * \tilde{H}_{\Phi}(B)(x) \right| dx \\ &\geq \int_{[0,1]^{n}} \left| \Psi_{|x|} * \tilde{H}_{\Phi}(B)(x) \right| dx. \end{split}$$

Here

$$\begin{aligned} \left| \Psi_{|x|} * \tilde{H}_{\Phi}(B)(x) \right| &= \int_{\mathbb{R}^n} \frac{\Phi(y)}{|y|^n} \int_{\mathbb{R}^n} \frac{1}{|x|^n} e^{-(\frac{|x-z|}{|x|})^2} B\left(\frac{z}{|y|}\right) \, dz \, dy \\ &\geqslant \int_{|y|\leqslant 1} \frac{\Phi(y)}{|y|^n} \int_{|z|\leqslant \frac{|y|}{4}} \frac{1}{|x|^n} e^{-(\frac{|x-z|}{|x|})^2} \, dz \, dy. \end{aligned}$$

Hence,

$$\int_{\mathbb{R}^n} \sup_{0 < s \leq 1} \left| \Psi_s * \tilde{H}_{\Phi}(B)(x) \right| dx$$
  
$$\geq \int_{|y| \leq 1} \frac{\Phi(y)}{|y|^n} \int_{|z| \leq \frac{|y|}{4}} \int_{|y| \leq |x| \leq 1} \frac{1}{|x|^n} e^{-(\frac{|x-z|}{|x|})^2} dx dz dy$$

$$\geq -\int_{|y|\leq 1} \Phi(y) \log_2 |y| \, dy = \infty.$$

This leads to a contradiction.

Next, suppose

$$\int_{|y| \ge 1} \Phi(y) \, dy = \infty.$$

We have

$$\begin{split} \int_{\mathbb{R}^n} \sup_{0 < s \leq 1} \left| \Psi_s * \tilde{H}_{\Phi}(B)(x) \right| \, dx &\geq \int_{\mathbb{R}^n} \left| \Psi_1 * \tilde{H}_{\Phi}(B)(x) \right| \, dx \\ &\geq \int_{|y| \geq 1} \frac{\Phi(y)}{|y|^n} \int_{|z| \leq \frac{|y|}{4}} \left( \int_{\mathbb{R}^n} e^{-(|x-z|)^2} \, dx \right) \, dz \, dy \\ &\geq \int_{|y| \geq 1} \Phi(y) \, dy = \infty. \end{split}$$

## 5.2 Boundedness on Lebesgue spaces

The duality techniques we have been using to show boundedness of Hausdorff operators on the Hardy spaces also yield the following nice results on power weight Lebesgue spaces.

**Theorem 5.2.** Let  $1 \leq p, q < \infty$ ,  $0 < \beta < n$ ,  $\gamma > \beta p - n$  and

$$\frac{1}{p} - \frac{\beta}{n+\gamma} = \frac{1}{q}.$$

Finally, let

$$C_p = \int_0^\infty |\Phi(t)|^{\frac{p}{p-1}} t^{\frac{\gamma-\beta p+n-p+1}{p-1}} dt \text{ for } p > 1, \qquad C_1 = \||\cdot|^{n-\beta+\gamma} \Phi(\cdot)\|_{L^\infty}.$$

For any  $p \ge 1$ , if  $C_p < \infty$  then

$$\|H_{\Phi,\beta}(f)(x)\|_{L^{q,\infty}(|x|^{\gamma}dx)} \le \|f\|_{L^{p}(|x|^{\gamma}dx)}.$$

*Proof.* We write

$$H_{\Phi,\beta}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi(\frac{|x|}{|y|})}{|y|^{n-\beta+\frac{\gamma}{p}}} f(y)|y|^{\frac{\gamma}{p}} dy.$$

By Hölder's inequality, we have

$$|H_{\Phi,\beta}(f)(x)| \leq \left\{ \int_{\mathbb{R}^n} |\frac{\Phi(\frac{|x|}{|y|})}{|y|^{n-\beta+\frac{\gamma}{p}}}|^{p'} dy \right\}^{\frac{1}{p'}} \|f\|_{L^p(|y|^{\gamma}dy)}.$$

Using polar coordinates and changing variables, we have

$$\left\{ \int_{\mathbb{R}^n} \left| \frac{\Phi(\frac{|x|}{|y|})}{|y|^{n-\beta+\frac{\gamma}{p}}} \right|^{p'} dy \right\}^{1/p'} = |S^{n-1}|^{\frac{1}{p'}} \left( \int_0^\infty \left| \frac{\Phi(t)}{t^{\beta-n-\gamma/p}} \right|^{p'} t^{-n-1} dt \right)^{\frac{1}{p'}} |x|^{n/p'+\beta-n-\gamma/p} dt = 1,$$
 If  $p = 1$ ,

$$|H_{\Phi,\beta}(f)(x)| \leq ||f||_{L^{1}(|x|^{\gamma}dx)} ||\frac{\Phi(\frac{|x|}{|\cdot|})}{|\cdot|^{n-\beta+\gamma}}||_{L^{\infty}}$$
  
=  $||f||_{L^{1}(|x|^{\gamma}dx)} |x|^{-n+\beta-\gamma} |||\cdot|^{n-\beta+\gamma} \Phi(\cdot)||_{L^{\infty}}.$ 

Let

$$K_p = \left| S^{n-1} \right|^{\frac{1}{p'}} C_p^{\frac{1}{p'}} \text{ for } p > 1 \qquad K_1 = C_1$$

Then for  $p \ge 1$  and any  $\lambda > 0$ , we have

$$\begin{split} &|\{x \in \mathbb{R}^{n}: |H_{\Phi,\beta}(f)(x)| > \lambda\}| \\ &\leq |\{x \in \mathbb{R}^{n}: K_{p}|x|^{n/p'+\beta-n-\gamma/p} \|f\|_{L^{p}(|x|^{\gamma}dx)} > \lambda\}| \\ &\leq |\{x \in \mathbb{R}^{n}: |x|^{n+\gamma/p-n/p'-\beta} \leq \frac{K_{p}\|f\|_{L^{p}(|x|^{\gamma}dx)}}{\lambda}\}| \\ &\leq |\{x \in \mathbb{R}^{n}: (|x|^{\frac{1}{p}-\frac{\beta}{n+\gamma}})^{n+\gamma} \leq \frac{K_{p}\|f\|_{L^{p}(|x|^{\gamma}dx)}}{\lambda}\}| \\ &= |\{x \in \mathbb{R}^{n}: |x|^{n+\gamma} \leq \frac{K_{p}^{q}\|f\|_{L^{p}(|x|^{\gamma}dx)}}{\lambda^{q}}\}|. \end{split}$$

Note that in the last equality, we have used the condition

$$\frac{1}{p} - \frac{\beta}{n+\gamma} = \frac{1}{q}.$$

This shows

$$\|H_{\Phi,\beta}(f)\|_{L^{q,\infty}(|x|^{\gamma}dx)} \le \|f\|_{L^{p}(|x|^{\gamma}dx)}$$

as desired.

Recall that if  $\Phi(t) = \chi_{(1,\infty)}(t)t^{-n+\beta}$ , then  $H_{\Phi,\beta}$  is the fractional Hardy operator. Let  $\mathcal{H}_{\beta}$  denote the fractional Hardy operator. Applying Theorem 5.2, we have the following corollary:

**Corollary 5.3.** Let  $1 < p, q < \infty$ ,  $0 < \beta < n$ ,  $np > \gamma > \beta p - n$  and

$$\frac{1}{p} - \frac{\beta}{n+\gamma} = \frac{1}{q}.$$

We have

$$\|\mathcal{H}_{\beta}(f)(x)\|_{L^{q,\infty}(|x|^{\gamma}dx)} \leq \|f\|_{L^{p}(|x|^{\gamma}dx)}.$$

*Proof.* If  $\gamma < np$ , then  $\Phi = \chi_{(1,\infty)}(t)t^{-n+\beta}$  satisfies the conditions of Theorem 5.2.

**Theorem 5.4.** Let  $1 \leq p, q < \infty, \ 0 < \beta < n, \ \gamma > \beta p - n$  and

$$\frac{1}{p} - \frac{\beta}{n+\gamma} = \frac{1}{q}.$$

In addition, let

$$C_{p,\epsilon} = \int_0^\infty |\Phi(t)|^{\frac{p}{p-1}} t^{\frac{\gamma-\beta p+n-p+1}{p-1}+\epsilon} dt \text{ for } p > 1, \qquad C_{1,\epsilon} = \||\cdot|^{n-\beta+\gamma+\epsilon} \Phi(\cdot)\|_{L^\infty}.$$

For any  $p \ge 1$ , if, for arbitrarily small positive  $\epsilon$ ,  $C_{p,\pm\epsilon} < \infty$ , then

$$\|H_{\Phi,\beta}(f)(x)\|_{L^q(|x|^{\gamma}dx)} \le \|f\|_{L^p(|x|^{\gamma}dx)}.$$

*Proof.* For p > 1, this result follows from Theorem 5.2 and the Marcinkiewicz Interpolation Theorem. To prove the result for p = 1, we define an analytic family:

$$H_{\Phi,z}(f)(x) = \int \frac{\Phi(\frac{|x|}{|y|})}{|y|^{n-z}} f(y) dy.$$

Following the proof of Theorem 5.2, it is easy to check that there exist  $\beta_1$ ,  $\beta_2$  satisfying  $0 < \beta_1 < \beta < \beta_2 < n$ , such that

$$\|H_{\Phi,z_1}(f)\|_{L^{q_1,\infty}(|x|^{\gamma}dx)} \le \|f\|_{L^1(|x|^{\gamma}dx)}$$

and

$$\|H_{\Phi,z_2}(f)(x)\|_{L^{q_2,\infty}(|x|^{\gamma}dx)} \le \|f\|_{L^1(|x|^{\gamma}dx)}$$

where Re  $z_1 = \beta_1$ , Re  $z_2 = \beta_2$  and  $1 < q_1 < q < q_2 < \infty$ . Thus we obtain the result by using the Stein-Weiss analytic Interpolation Theorem (see Corollary 2.2).

5.3 BOUNDEDNESS ON  $H^p(\mathbb{R}^n)$ 

5.3.1  $H^p \to L^q$  Boundedness of  $H_{\Phi,\beta}$ 

In this section, we suppose that  $\Phi$  is radial, that is, if  $|x_1| = |x_2|$ , then  $\Phi(x_1) = \Phi(x_2)$ . Thus we can think of  $\Phi$  as having domain  $[0, \infty)$ , where  $\Phi(t) = \Phi(x)$  when |x| = t.

**Lemma 5.5.** Let  $0 \leq \beta < n$  and

$$\psi(y) = \frac{\Phi(1/|y|)}{|y|^{n-\beta}}.$$

Assume  $0 and <math>\alpha = n\left(\frac{1}{p} - 1\right)$ . If  $\psi \in \Lambda_{\alpha}$ , then

$$\|H_{\Phi,\beta}(f)\|_{L^{q,\infty}(\mathbb{R}^n)} \le \|f\|_{H^p(\mathbb{R}^n)}$$

where q satisfies

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}$$

*Proof.* Changing variables, we compute

$$\frac{\Phi(|x|/|y|)}{|y|^{n-\beta}} = |x|^{-n+\beta} \psi(\frac{y}{|x|}).$$

By duality and scaling,

$$|H_{\Phi}(f)(x)| \leq |x|^{-n+\beta} \|f\|_{H^{p}(\mathbb{R}^{n})} \left\|\psi\left(\frac{\cdot}{|x|}\right)\right\|_{\Lambda_{\alpha}} = |x|^{-n/p+\beta} \|f\|_{H^{p}(\mathbb{R}^{n})} \|\psi\|_{\Lambda_{\alpha}}.$$

This shows that for all  $\lambda > 0$ ,

$$\left| \left\{ x \in \mathbb{R}^{n} : |H_{\Phi}(f)(x)| > \lambda \right\} \right|$$

$$\leq \left| \left\{ x \in \mathbb{R}^{n} : |x|^{\frac{n}{p} - \beta} < \|f\|_{H^{p}(\mathbb{R}^{n})} \|\psi\|_{\Lambda_{\alpha}} \lambda^{-1} \right\} \right|$$

$$\leq \left| \left\{ x \in \mathbb{R}^{n} : |x|^{n} \leq \left\{ \|f\|_{H^{p}(\mathbb{R}^{n})} \|\psi\|_{\Lambda_{\alpha}} \lambda^{-1} \right\}^{q} \right\} \right|.$$

The last inequality holds because

$$\frac{n}{\frac{n}{p}-\beta} = q.$$

Thus, we obtain

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^{q,\infty}(\mathbb{R}^n)} \le \left\|f\right\|_{H^p(\mathbb{R}^n)}$$

as desired.

**Theorem 5.6.** Let  $\beta$ , p,  $\alpha$ , q,  $\psi$  be as in Lemma 5.5. If for some  $\epsilon > 0$  small enough that  $\alpha - \epsilon > 0$ ,  $\psi \in \Lambda_{\alpha + \epsilon} \cap \Lambda_{\alpha - \epsilon}$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^q(\mathbb{R}^n)} \le \|f\|_{H^p(\mathbb{R}^n)}.$$

*Proof.* Choose  $p_1, p_2$ , s.t.  $\alpha + \epsilon = n(\frac{1}{p_1} - 1), \ \alpha - \epsilon = n(\frac{1}{p_2} - 1)$ , and choose  $q_1, q_2$  satisfying

$$\frac{1}{p_i} = \frac{1}{q_i} + \frac{\beta}{n}, \quad i = 1, 2.$$

An easy computation shows that  $p_1, p_2, p, q_1, q_2, q$  satisfy the requirement for the Marcinkiewisz interpolation, and by the Lemma 5.5,  $||H_{\Phi,\beta}(f)||_{L^{q_i,\infty}(\mathbb{R}^n)} \leq ||f||_{H^{p_i}(\mathbb{R}^n)}$ . So by Marcinkiewisz Interpolation,  $||H_{\Phi,\beta}(f)||_{L^{q}(\mathbb{R}^n)} \leq ||f||_{H^{p}(\mathbb{R}^n)}$ .

Unfortunately, we cannot continue in the manner shown in Theorem 3.1, since  $\mathcal{R}_J \circ H_{\Phi,\beta}(f) \neq H_{\Phi,\beta}(\mathcal{R}_J f)$ , as was in the case in one dimension. Instead, we see the following.

**Lemma 5.7.** Suppose  $\Phi$ ,  $\widehat{\Phi} \in L^1(\mathbb{R}^n)$ , and  $0 < \beta < n$ . Then for any  $J = \{j_1, \ldots, j_L\} \in \{0, 1, 2, \ldots, n\}^L$ ,  $f \in S$ , we have

$$\mathcal{R}_J H_{\Phi,\beta}(f) = H_{\mathcal{R}_J \Phi,\beta}(f).$$

*Proof.* Without loss of generality, we suppose L = 1. (L > 1 may be shown similarly.) Fix J = j. We show the desired equality by taking Fourier transform and inverse Fourier transform. By definition,

$$\widehat{H_{\Phi,\beta}(f)}(\xi) = \int_{\mathbb{R}^n} \left( \int_{\mathbb{R}^n} \frac{f(y)}{|y|^{n-\beta}} \Phi\left(\frac{x}{|y|}\right) \, dy \right) e^{-i\langle x,\xi \rangle} \, dx$$

Noting that  $\Phi \in L^1$ , f is a Schwartz function, and  $\beta > 0$ , we may use Fubini's theorem to obtain

$$\widehat{H_{\Phi,\beta}(f)}(\xi) = \int_{\mathbb{R}^n} \left( \int_{\mathbb{R}^n} \frac{f(y)}{|y|^{n-\beta}} \Phi(\frac{x}{|y|}) e^{-i\langle x,\xi \rangle} \, dx \right) \, dy$$
$$= \int_{\mathbb{R}^n} \left( \int_{\mathbb{R}^n} |y|^\beta f(y) \Phi(u) e^{-i\langle u, |y|\xi \rangle} \, du \right) \, dy$$
$$= \int_{\mathbb{R}^n} |y|^\beta f(y) \hat{\Phi}(|y|\xi) \, dy.$$

Thus, we see that

$$\widehat{\mathcal{R}_{j}H_{\Phi,\beta}}(f)(\xi) \cong \frac{\xi_{j}}{|\xi|}\widehat{H_{\Phi,\beta}}(\xi) = \frac{\xi_{j}}{|\xi|}\int_{\mathbb{R}^{n}} |y|^{\beta} f(y)\widehat{\Phi}(|y|\xi) \, dy$$

Next, we let  $\mathcal{F}^{-1}$  denote the inverse Fourier transform, and compute

$$\mathcal{F}^{-1}\left(\mathcal{R}_{j}\widehat{H_{\Phi,\beta}}(f)\right)(x) = \int_{\mathbb{R}^{n}} \left(\frac{\xi_{j}}{|\xi|} \int_{\mathbb{R}^{n}} |y|^{\beta} f(y)\hat{\Phi}(|y||\xi) \, dy\right) e^{i\langle x,\xi\rangle} \, d\xi.$$

Noting that  $\hat{\Phi} \in L^1$  and again that  $\beta > 0$ , we again use Fubini's theorem to obtain

$$\mathcal{F}^{-1}\left(\widehat{\mathcal{R}_{j}H_{\Phi,\beta}(f)}\right)(x) = \int_{\mathbb{R}^{n}} |y|^{\beta} f(y)\left(\int_{\mathbb{R}^{n}} \frac{\xi_{j}}{|\xi|} \widehat{\Phi}(|y|\xi)e^{i\langle x,\xi\rangle} d\xi\right) dy$$

$$= \int_{\mathbb{R}^n} |y|^{\beta-n} f(y) \mathcal{R}_j \Phi\left(\frac{x}{|y|}\right) dy$$
$$= H_{\mathcal{R}_j \Phi, \beta}$$

as desired.

# 5.3.2 $H^p \to H^q$ Boundedness of $H_{\Phi,\beta}$

The condition of Lemma 5.5, that  $\psi \in \Lambda_{\alpha}$ , can be rewritten, when  $\Phi$  is radial, as

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}\Phi(\frac{|x|}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})} \leq |x|^{-\alpha}.$$

In this form, we can omit the condition that  $\Phi$  is radial, obtaining

**Lemma 5.8.** Let  $0 , <math>0 \leq \beta < n$ , and  $\alpha = n(\frac{1}{p} - 1)$ . If  $\Phi$  satisfies, for all  $x \in \mathbb{R}^n$ ,

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}\Phi(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})} \leq |x|^{-\alpha}$$

then

$$\|H_{\Phi}(f)\|_{L^{q,\infty}(\mathbb{R}^n)} \leqslant \|f\|_{H^p(\mathbb{R}^n)}$$

with

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

*Proof.* Noting the pairing inequality (equation (2.1)) relating the Lipschitz space  $\Lambda_{\alpha}$ and the Hardy space  $H^p$ , and multiplying and dividing by  $|x|^{n-\beta}$  in the definition of  $H_{\Phi,\beta}$ , we obtain

$$|H_{\Phi,\beta}(f)(x)| \leq ||f||_{H^p(\mathbb{R}^n)} |x|^{-n+\beta} \left\| \frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}} \Phi(\frac{x}{|\cdot|}) \right\|_{\Lambda_{\alpha}(\mathbb{R}^n)}$$
$$\leq |x|^{\frac{-n}{p}+\beta} ||f||_{H^p(\mathbb{R}^n)}.$$

This shows that for all  $\lambda > 0$ ,

$$\begin{split} &|\{x \in \mathbb{R}^{n} : |H_{\Phi,\beta}(f)(x)| > \lambda\}| \\ &\leq \left|\left\{x \in \mathbb{R}^{n} : |x|^{\frac{-n}{p} + \beta} \|f\|_{H^{p}(\mathbb{R}^{n})} \geqslant \lambda\right\}\right| \\ &= \left|\left\{x \in \mathbb{R}^{n} : |x| \leq \left(\frac{\|f\|_{H^{p}(\mathbb{R}^{n})}}{\lambda}\right)^{\frac{q}{n}}\right\}\right| \\ &\leq \left(\frac{\|f\|_{H^{p}(\mathbb{R}^{n})}}{\lambda}\right)^{q}. \end{split}$$

Thus,  $||H_{\Phi,\beta}(f)||_{L^{q,\infty}} \leq ||f||_{H^p}$ , as desired.

In many important cases, even when  $\Phi$  is radial, the condition of Lemma 5.8 is easier to check than the one give in Lemma 5.5. For example, we see this in the proof of the following corollary.

**Corollary 5.9.** Let  $\Phi(y)$  be the Gaussian function  $e^{-|y|^2}$  or the Poisson function  $e^{-|y|}$ . The Hausdorff operator  $H_{\Phi,\beta}$  is bounded from  $H^p(\mathbb{R}^n)$  to  $L^q(\mathbb{R}^n)$  for all 0 , $<math>0 \leq \beta < n$ , and  $\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}$ .

*Proof.* We will show the corollary for the Poisson function  $e^{-|y|}$ , since the proof for the Gaussian function is similar. By Lemma 5.8, we need to check

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}e^{-\frac{|x|}{|\cdot|}}\right\|_{\Lambda_{\alpha}} \leqslant |x|^{-\alpha}$$

for all integer  $\alpha = 0, 1, 2, ...$ , where by convention, we denote  $\Lambda_0 = L^{\infty}$ . First, when  $\alpha = 0$ ,

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}e^{-\frac{|x|}{|\cdot|}}\right\|_{L^{\infty}} = \sup_{|y|>0} \left|\frac{|x|^{n-\beta}}{|y|^{n-\beta}}e^{-\frac{|x|}{|y|}}\right|$$
$$= \sup_{|y|>0} \left||y|^{n-\beta}e^{-|y|}\right| \leqslant 1.$$

For integers  $\alpha > 0$ , we recall the definition of the  $\Lambda_{\alpha}$  norm, namely

$$\|f\|_{\Lambda_{\alpha}} = \sup_{|I|=\alpha} \left\{ \left\| \partial^{I} f \right\|_{L^{\infty}} \right\}$$

where  $I = (i_1, i_2, \cdots, i_n) \in \mathbb{N}^n$  is a multi-index and  $\hat{c}^I$  denotes the derivative

$$\left(\frac{\partial}{\partial y_1}\right)^{i_1} \left(\frac{\partial}{\partial y_2}\right)^{i_2} \cdots \left(\frac{\partial}{\partial y_n}\right)^{i_n}.$$

So we must show that, for each multi-index I such that  $|I| = \alpha$ ,

$$\left\| \widehat{c}^{I} \left( \frac{|x|^{n-\beta}}{|y|^{n-\beta}} e^{-\frac{|y|}{|x|}} \right) \right\|_{L^{\infty}} \leq |x|^{-\alpha}.$$

This can easily be shown by computation.

We can also see how the condition of Lemma 5.8 might be extended to allow us to obtain a boundedness result on Hardy spaces.

**Theorem 5.10.** Let  $n \ge 2$ ,  $\alpha = n(\frac{1}{p} - 1)$ ,  $0 < \beta < n$ , and  $\Phi, \widehat{\Phi} \in L^1$ . If, for some L large enough that  $p > \frac{n-1}{n-1+L}$ , all generalized Riesz transforms  $\mathcal{R}_J(\Phi) = \mathcal{R}_{j_1} \cdots \mathcal{R}_{j_L} \Phi$  satisfy

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}(\mathcal{R}_{J}\Phi)(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})} \leqslant |x|^{-\alpha}$$

then

$$\|H_{\Phi,\beta}(f)\|_{H^{q,\infty}(\mathbb{R}^n)} \leqslant \|f\|_{H^p(\mathbb{R}^n)}$$

with

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

*Proof.* Applying Lemma 5.8 to each  $\mathcal{R}_J \Phi$ , we obtain for each  $\mathcal{R}_J$ ,

$$\left\|H_{\mathcal{R}_J(\Phi),\beta}\right\|_{L^{q,\infty}} \leqslant \|f\|_{H^p}.$$

Lemma 5.7 tells us that for each  $\mathcal{R}_J$ ,  $\mathcal{R}_J H_{\Phi,\beta} = H_{\mathcal{R}_J(\Phi),\beta}(f)$ . Applying the Reisz transform characterization of  $H^{q,\infty}$ ,

$$\|H_{\Phi,\beta}(f)\|_{H^{q,\infty}} \cong \sum_{J} \|\mathcal{R}_{J}H_{\Phi,\beta}(f)\|_{L^{q,\infty}} \leqslant \|f\|_{H^{p}}$$

as desired.

The condition of Theorem 5.10 is somewhat laborious to check. The main result of this chapter is to provide a sufficient (but not necessary) condition on  $\Phi$  to meet the requirement of Theorem 5.10.

**Theorem 5.11.** Suppose  $0 , <math>0 < \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform of  $\Phi$  and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M = n(\frac{1}{p}-1)$ , suppose that  $\widehat{\Phi}$  is a function in  $C^{2M+n}(\mathbb{R}^n)$  with compact support in the set  $\mathbb{R}^n \setminus \{0\}$ . Then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^{q,\infty}(\mathbb{R}^n)} \leq \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

*Proof.* Let  $M = n(\frac{1}{p} - 1)$ . By Theorem 5.10, we need to check

$$\left\|\frac{|x|^{n-\beta}}{\left|\cdot\right|^{n-\beta}}(\mathcal{R}_{J}\Phi)(\frac{x}{\left|\cdot\right|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})} \leqslant |x|^{-\alpha},$$

with  $\alpha = M$ . Taking Fourier transforms, we may write

$$\frac{|x|^{n-\beta}}{|y|^{n-\beta}} (\mathcal{R}_J \Phi)(\frac{x}{|y|})$$
  
=  $|x|^{n-\beta} \int_{\mathbb{R}^n} \frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^L} \widehat{\Phi(\xi)} |y|^{-(n-\beta)} e^{i\frac{1}{|y|}\langle\xi,x\rangle} d\xi.$ 

For a multi-index  $I = (i_1, i_2, \cdots, i_n) \in \mathbb{N}^n$ , we will use the notation

$$\partial^{I} = \left(\frac{\partial}{\partial y_{1}}\right)^{i_{1}} \left(\frac{\partial}{\partial y_{2}}\right)^{i_{2}} \cdots \left(\frac{\partial}{\partial y_{n}}\right)^{i_{n}},$$

and

$$|I|=i_1+i_2+\cdots+i_n.$$

From easy computations

$$\frac{\partial}{\partial y_j} e^{i\frac{1}{|y|}\langle \xi, x \rangle} \cong \langle \xi, x \rangle \frac{y_j}{|y|^3} e^{i\frac{1}{|y|}\langle \xi, x \rangle}$$

and

$$\frac{\partial^2}{\partial y_j^2} e^{i\frac{1}{|y|}\langle\xi,x\rangle} = c_1 \langle\xi,x\rangle^2 \frac{y_j^2}{|y|^6} e^{i\frac{1}{|y|}\langle\xi,x\rangle} + c_2 \langle\xi,x\rangle \frac{1}{|y|^3} e^{i\frac{1}{|y|}\langle\xi,x\rangle} + c_3 \langle\xi,x\rangle \frac{y_j^2}{|y|^5} e^{i\frac{1}{|y|}\langle\xi,x\rangle},$$

where  $c_1, c_2, c_3$  are constants, in general, it is easy to see that for any multi-index I,

$$\hat{c}^{I} e^{i \frac{1}{|y|} \langle \xi, x \rangle} = e^{i \frac{1}{|y|} \langle \xi, x \rangle} \sum_{\substack{|I| \leq l_1 \leq 2|I| \\ l_1 - l_2 = |I|}} Q_{l_1}(y) P_{l_2}(\langle \xi, x \rangle),$$

where  $Q_{l_1}(y)$  is a  $C^{\infty}(\mathbb{R}^n \setminus \{0\})$  function satisfying

$$|Q_{l_1}| \leqslant \frac{1}{|y|^{l_1}}$$

and  $P_{l_2}$  is a homogeneous polynomial of degree  $l_2$ .

Also, it is easy to check

$$\partial^{I}\left(\left|y\right|^{-n+\beta}\right) = \frac{S_{I}(y)}{\left|y\right|^{n-\beta+2\left|I\right|}},$$

where  $S_I(y)$  is a  $C^{\infty}(\mathbb{R}^n \setminus \{0\})$  function satisfying

$$|S_I(y)| \leq |y|^{|I|}.$$

So, by the generalized Leibniz rule, for any multi-index I with  $|I| \ge 1$ , we have

$$\partial^{I} \left( \frac{e^{i \frac{1}{|y|} \langle \xi, x \rangle}}{|y|^{n-\beta}} \right) = \sum_{K_{1}+K_{2}=I} C_{K_{1}}^{I} \partial^{K_{1}} (|y|^{-n+\beta}) \partial^{K_{2}} (e^{i \frac{1}{|y|} \langle \xi, x \rangle})$$
$$= \sum_{K_{1}+K_{2}=I} C_{K_{1}}^{I} \frac{S_{K_{1}}(y)}{|y|^{n-\beta+2|K_{1}|}} \left( \sum_{\substack{|K_{2}| \leqslant l_{1} \leqslant 2|K_{2}|\\l_{1}-l_{2}=|K_{2}|}} Q_{l_{1}}(y) e^{i \frac{1}{|y|} \langle \xi, x \rangle} P_{l_{2}} (\langle \xi, x \rangle) \right)$$

where  $C_{K_1}^I$  are constants.

It is easy to see that

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}(\mathcal{R}_{J}\Phi)(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})}$$

$$= \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left| \partial^I \left( |x|^{n-\beta} \int_{\mathbb{R}^n} \frac{\xi_{j_1} \xi_{j_2} \cdots \xi_{j_L}}{|\xi|^L} \widehat{\Phi}(\xi) |y|^{-n+\beta} e^{i\frac{1}{|y|} \langle \xi, x \rangle} d\xi \right) \right| \right\}$$
$$= \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} |x|^{n-\beta} \left| \int_{\mathbb{R}^n} \frac{\xi_{j_1} \xi_{j_2} \cdots \xi_{j_L}}{|\xi|^L} \widehat{\Phi}(\xi) \partial^I \left( |y|^{-n+\beta} e^{i\frac{1}{|y|} \langle \xi, x \rangle} \right) d\xi \right| \right\}.$$

So, let us consider the value of

$$\begin{split} &|x|^{n-\beta} \left| \int_{\mathbb{R}^{n}} \frac{\xi_{j_{1}}\xi_{j_{2}}\cdots\xi_{j_{L}}}{|\xi|^{L}} \widehat{\Phi}(\xi) \partial^{I} \left( |y|^{-n+\beta} e^{i\frac{1}{|y|}\langle\xi,x\rangle} \right) d\xi \right| \\ &= |x|^{n-\beta} \left| \int_{\mathbb{R}^{n}} \frac{\xi_{j_{1}}\xi_{j_{2}}\cdots\xi_{j_{L}}}{|\xi|^{L}} \widehat{\Phi}(\xi) \sum_{K_{1}+K_{2}=I} C_{K_{1}}^{I} \frac{S_{K_{1}}(y)}{|y|^{n-\beta+2|K_{1}|}} \left( \sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\ l_{1}-l_{2}=|K_{2}|}} Q_{l_{1}}(y) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_{2}}(\langle\xi,x\rangle) \right) d\xi \right| \\ &= |x|^{n-\beta} \left| \sum_{K_{1}+K_{2}=I} C_{K_{1}}^{I} \frac{S_{K_{1}}(y)}{|y|^{n-\beta+2|K_{1}|}} \left( \sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\ l_{1}-l_{2}=|K_{2}|}} Q_{l_{1}}(y) \int_{\mathbb{R}^{n}} \frac{\xi_{j_{1}}\xi_{j_{2}}\cdots\xi_{j_{L}}}{|\xi|^{L}} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_{2}}(\langle\xi,x\rangle) d\xi \right) \right| \\ &\leqslant |x|^{n-\beta} \sum_{K_{1}+K_{2}=I} \frac{|S_{K_{1}}(y)|}{|y|^{n-\beta+2|K_{1}|}} \left( \sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\ l_{1}-l_{2}=|K_{2}|}} |Q_{l_{1}}(y)| \left| \int_{\mathbb{R}^{n}} \frac{\xi_{j_{1}}\xi_{j_{2}}\cdots\xi_{j_{L}}}{|\xi|^{L}} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_{2}}(\langle\xi,x\rangle) d\xi \right| \right) \\ &\leqslant |x|^{n-\beta} \sum_{K_{1}+K_{2}=I} \frac{|y|^{|K_{1}|}}{|y|^{n-\beta+2|K_{1}|}} \left( \sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\ l_{1}-l_{2}=|K_{2}|}} \frac{1}{|y|^{l}}} \left| \int_{\mathbb{R}^{n}} \frac{\xi_{j_{1}}\xi_{j_{2}}\cdots\xi_{j_{L}}}{\xi|^{L}} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_{2}}(\langle\xi,x\rangle) d\xi \right| \right) \\ &= \left( \frac{|x|}{|y|} \right)^{n-\beta} \sum_{K_{1}+K_{2}=I} \frac{1}{|y|^{|K_{1}|}} \left( \sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\ l_{1}-l_{2}=|K_{2}|}} \frac{1}{|\xi|^{L}}} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_{2}}(\langle\xi,x\rangle) d\xi \right| \right). \end{split}$$

We now consider two possible cases. If

$$\frac{|x|}{|y|} \leqslant 1,$$

then

$$\sup_{y\in\mathbb{R}^n\setminus\{0\}}\left\{\sum_{|I|=\alpha}\left|\partial^I\left(|x|^{n-\beta}\int_{\mathbb{R}^n}\frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^L}\widehat{\Phi}(\xi)\,|y|^{-(n-\beta)}\,e^{i\frac{1}{|y|}\langle\xi,x\rangle}\,d\xi\right)\right|\right\}$$

$$\begin{split} &\leqslant \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \sum_{K_1+K_2=I} \frac{1}{|y|^{|K_1|}} \left( \sum_{\substack{|K_2| \leqslant l_1 \leqslant 2|K_2| \\ l_1-l_2 = |K_2|}} \frac{1}{|\xi|^{L}} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_2}(\langle\xi,x\rangle) \ d\xi \right| \right) \right\} \\ &\leqslant \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \sum_{K_1+K_2=I} \frac{1}{|y|^{|K_1|}} \left( \sum_{\substack{|K_2| \leqslant l_1 \leqslant 2|K_2| \\ l_1-l_2 = |K_2|}} \frac{1}{|y|^{l_2}} \int_{\mathbb{R}^n} \left| \frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^{L-l_2}} \widehat{\Phi}(\xi) \right| \ d\xi \right) \right\} \\ &\leqslant \frac{1}{|x|^{\alpha}} \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{\substack{|I|=\alpha} K_1+K_2=I} \sum_{\substack{|K_2| \leqslant l_1 \leqslant 2|K_2| \\ l_1-l_2 = |K_2|}} \left( \sum_{\substack{|K_2| \leqslant l_1 \leqslant 2|K_2| \\ l_1-l_2 = |K_2|}} \frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^{L-l_2}} \widehat{\Phi}(\xi) \right| \ d\xi \right) \right\} \\ &\leqslant |x|^{-\alpha} \,. \end{split}$$

On the other hand, if

$$\frac{|x|}{|y|} > 1,$$

for fixed x, without loss of generality, we assume  $|x_1| \ge \frac{|x|}{n}$ . For each integral

$$\frac{1}{|y|^{l_1}} \left| \int_{\mathbb{R}^n} \frac{\xi_{j_1} \xi_{j_2} \cdots \xi_{j_L}}{|\xi|^L} \widehat{\Phi}(\xi) e^{i \frac{1}{|y|} \langle \xi, x \rangle} P_{l_2}(\langle \xi, x \rangle) \, d\xi \right|,$$

using integration by parts on the  $\xi_1$  variable k times, where

$$k = 2M + n \ge |I| + l_2 + n - \beta.$$

Recalling  $l_1 - l_2 = |K_2|$ , and using the fact that  $\widehat{\Phi} \in C^{2M+n}(\mathbb{R}^n)$  with compact support in  $\mathbb{R}^n \setminus \{0\}$ , we obtain

$$\frac{1}{|y|^{l_1}} \left| \int_{\mathbb{R}^n} \frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^L} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} P_{l_2}(\langle\xi,x\rangle) d\xi \right| \\
\leqslant \frac{1}{|y|^{l_1-l_2}} \left( \frac{|y|}{|x|} \right)^k \frac{|x|^{l_2}}{|y|^{l_2}} \\
\leqslant \frac{1}{|y|^{|K_2|}} \left( \frac{|y|}{|x|} \right)^{|I|+n-\beta}.$$

This shows that

$$\sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left| \partial^I \left( |x|^{n-\beta} \int_{\mathbb{R}^n} \frac{\xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}}{|\xi|^L} \widehat{\Phi}(\xi) |y|^{-n+\beta} e^{i\frac{1}{|y|}\langle\xi,x\rangle} d\xi \right) \right| \right\}$$
  
$$\leqslant \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left( \frac{|x|}{|y|} \right)^{n-\beta} \sum_{K_1+K_2=I} \frac{1}{|y|^{|K_1|}} \frac{1}{|y|^{|K_2|}} \left( \frac{|y|}{|x|} \right)^{|I|+n-\beta} \right\}$$
  
$$\approx \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left( \frac{|x|}{|y|} \right)^{n-\beta} \frac{1}{|y|^{|I|}} \left( \frac{|y|}{|x|} \right)^{|I|+n-\beta} \right\} \leqslant |x|^{-\alpha}.$$

This proves that for each J, when  $M = n(\frac{1}{p} - 1)$ ,

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}(\mathcal{R}_{J}\Phi)(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})} \leqslant |x|^{-\alpha},$$

and thus by Theorem 5.10,  $\|H_{\Phi,\beta}(f)\|_{H^{q,\infty}(\mathbb{R}^n)} \leq \|f\|_{H^p(\mathbb{R}^n)}$ .

We show the following  $H^1(\mathbb{R}^n)$ -boundedness result in order to provide another endpoint for interpolation. Note that the results obtained in Section 5.1 were about  $\widetilde{H}_{\Phi,\beta}$ , and so unrelated to this one.

**Lemma 5.12.** Suppose  $0 < \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform of  $\Phi$  and

$$1 = \frac{1}{q} + \frac{\beta}{n}$$

Suppose  $\widehat{\Phi}$  is a function in  $C^n(\mathbb{R}^n)$  with compact support. Then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^{q,\infty}} \leqslant \left\|f\right\|_{H^1}.$$

*Proof.* Define the analytic family of operators

$$H_{\Phi,z}(f)(x) = \int_{\mathbb{R}^n} \frac{\Phi\left(\frac{x}{|y|}\right)}{|y|^{n-z}} f(y) \ dy$$

Fix  $z_1, z_2$  such that Re  $z_1 = \beta_1$ , Re  $z_2 = \beta_2$ , with  $0 < \beta_1 < \beta < \beta_2 < n$ . Using Hölder's inequality, for i = 1, 2, we have

$$|H_{\Phi,z_i}(f)(x)| = \left| \int_{\mathbb{R}^n} \frac{\Phi(\frac{x}{|y|})}{|y|^{n-z_i}} f(y) \ dy \right|$$

$$\leqslant \|f\|_{L^{1}} \sup_{|y|>0} \left| \frac{\Phi(\frac{x}{|y|})}{|y|^{n-\beta_{i}}} \right|$$

Clearly,  $\Phi$  is a bounded function, by assumption. Thus, if  $\frac{|x|}{|y|} \leq 1$ , we can write,

$$|H_{\Phi,z_i}(f)(x)| \leq ||f||_{L^1} |x|^{-n+\beta_i}.$$

If  $\frac{|x|}{|y|} > 1$ , we use integration by parts n times, using  $\widehat{\Phi} \in C^n$  with compact support to obtain

$$\left|\frac{\Phi(\frac{x}{|y|})}{|y|^{n-\beta_i}}\right| = \left|\frac{1}{|y|^{n-\beta_i}} \int_{\mathbb{R}^n} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle\xi,x\rangle} d\xi\right| \leqslant |x|^{-n+\beta_i}.$$

This gives

$$\begin{aligned} |\{x: |H_{\Phi,z_i}(f)(x)| > \lambda\}| &\leqslant \left| \left\{ x: ||f||_{L^1} |x|^{-n+\beta_i} \geqslant \lambda \right\} \right| \\ &\simeq \left| \left\{ x: |x| \leqslant \left( \frac{||f||_{L^1}}{\lambda} \right)^{\frac{1}{n-\beta_i}} \right\} \right| \\ &\simeq \left( \frac{||f||_{L^1}}{\lambda} \right)^{\frac{n}{n-\beta_i}} \\ &= \left( \frac{||f||_{L^1}}{\lambda} \right)^{q_i}. \end{aligned}$$

where  $q_1, q_2$  are such that

$$\frac{1}{q_i} + \frac{\beta_i}{n} = 1 \qquad i = 1, 2.$$

By definition of the weak type spaces, then, the last inequality implies

$$||H_{\Phi,z_i}(f)||_{L^{q_i,\infty}(\mathbb{R}^n)} \leq ||f||_{L^1(\mathbb{R}^n)}.$$

Thus, by Stein-Weiss analytic interpolation, we obtain, for any  $q_0$  such that  $q_1 < q_0 < q_2$ ,

$$\|H_{\Phi,\beta}(f)\|_{L^{q_0}(\mathbb{R}^n)} \leq \|f\|_{L^1(\mathbb{R}^n)} \leq \|f\|_{H^1(\mathbb{R}^n)}.$$

In particular, this holds when  $q_0 = q$ , so

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^q(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^1(\mathbb{R}^n)}.$$

But q > 1, so  $H^q = L^q$ , and thus

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^q(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^1(\mathbb{R}^n)}.$$

Further, we know  $\left\|\cdot\right\|_{H^{q,\infty}}\leqslant\left\|\cdot\right\|_{H^{q}},$  so

$$\|H_{\Phi,\beta}(f)\|_{H^{q,\infty}(\mathbb{R}^n)} \leqslant \|f\|_{H^1(\mathbb{R}^n)}$$

as desired.

**Theorem 5.13.** Suppose  $0 , <math>0 < \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform of  $\Phi$  and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M > n(\frac{1}{p}-1)$ , suppose that  $\widehat{\Phi}$  is a function in  $C^{2M+n}(\mathbb{R}^n)$  with compact support in the set  $\mathbb{R}^n \setminus \{0\}$ . Then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^q(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

*Proof.* Since  $M > n(\frac{1}{p} - 1)$  we have a  $p_1 < p$  satisfying

$$M = n\left(\frac{1}{p_1} - 1\right).$$

By Theorem 5.11, we have

$$||H_{\Phi,\beta}(f)||_{H^{q_1,\infty}(\mathbb{R}^n)} \leq ||f||_{H^{p_1}(\mathbb{R}^n)},$$

where

$$\frac{1}{p_1} = \frac{1}{q_1} + \frac{\beta}{n}.$$

Further, by Lemma 5.12,

$$\|H_{\Phi,\beta}(f)\|_{H^{q_2,\infty}(\mathbb{R}^n)} \leqslant \|f\|_{H^1(\mathbb{R}^n)}$$

where

$$1 = \frac{1}{q_2} + \frac{\beta}{n}$$

Thus, by an interpolation argument, we obtain

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^q(\mathbb{R}^n)} \leq \left\|f\right\|_{H^p(\mathbb{R}^n)}$$

as desired.

The techniques used to prove Theorems 5.11 and 5.13 can also be used to show the following result. The specific  $\Phi$  considered here does not meet the conditions on Theorems 5.11 and 5.13, which may suggest in general a method of relaxing these conditions.

Corollary 5.14. Suppose  $0 , <math>0 < \beta < n$ . Let  $\Psi \in S(\mathbb{R}^n)$ , and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M \ge n(\frac{1}{p}-1)$ , let

$$\Phi = \begin{cases} |\Delta|^{M+\frac{n}{2}}\Psi & \text{if } n \text{ is even} \\ |\Delta|^{M+\frac{n+1}{2}}\Psi & \text{if } n \text{ is odd} \end{cases}$$

where  $\Delta$  is the Laplacian and  $|\Delta|^{\gamma}$  is the fractional Laplacian defined for any function F by

$$(\widehat{|\Delta|^{\gamma} F})(\xi) = |\xi|^{2\gamma} \widehat{F}(\xi).$$

If  $M = n(\frac{1}{p} - 1)$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^{q,\infty}(\mathbb{R}^n)} \leq \left\|f\right\|_{H^p(\mathbb{R}^n)}$$

If  $M > n(\frac{1}{p} - 1)$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{H^q(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

Proof. Recall that according to Theorem 5.10, we must show that

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}(\mathcal{R}_J\Phi)(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^n)} \leqslant |x|^{-\alpha}$$

for for all  $J \in \{0, ..., n\}^L$ , where L is large enough that  $p > \frac{n-1}{n-1+L}$ . Note that we can choose L even such that 2M + n > L.

In the proof of Theorem 5.11, we have shown that

$$\left\|\frac{\left|x\right|^{n-\beta}}{\left|\cdot\right|^{n-\beta}}(\mathcal{R}_{J}\Phi)(\frac{x}{\left|\cdot\right|})\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})}$$

$$\leqslant \sup_{y\in\mathbb{R}^{n}\setminus\{0\}}\left\{\sum_{|I|=\alpha}\left(\frac{\left|x\right|}{\left|y\right|}\right)^{n-\beta}\sum_{K_{1}+K_{2}=I}\frac{1}{\left|y\right|^{\left|K_{1}\right|}}\left(\sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\l_{1}-l_{2}=\left|K_{2}\right|}}\frac{1}{\left|\xi\right|^{L}}\widehat{\Phi}(\xi)e^{i\frac{1}{\left|y\right|}\langle\xi,x\rangle}P_{l_{2}}(\langle\xi,x\rangle) d\xi\right|\right)\right\},$$

where  $P_{l_2}$  is a homogeneous polynomial of degree  $l_2$ . In this case, from the definition of  $\Phi$ , we can rewrite the above as

$$\left\|\frac{\left|x\right|^{n-\beta}}{\left|\cdot\right|^{n-\beta}}(\mathcal{R}_{J}\Phi)\left(\frac{x}{\left|\cdot\right|}\right)\right\|_{\Lambda_{\alpha}(\mathbb{R}^{n})}$$

$$\leqslant \sup_{y\in\mathbb{R}^{n}\setminus\{0\}}\left\{\sum_{|I|=\alpha}\left(\frac{\left|x\right|}{\left|y\right|}\right)^{n-\beta}\sum_{K_{1}+K_{2}=I}\frac{1}{\left|y\right|^{\left|K_{1}\right|}}\left(\sum_{\substack{|K_{2}|\leqslant l_{1}\leqslant 2|K_{2}|\\l_{1}-l_{2}=\left|K_{2}\right|}}\frac{1}{\left|\int_{\mathbb{R}^{n}}P(\xi)\widehat{\Psi}(\xi)e^{i\frac{1}{\left|y\right|}\langle\xi,x\rangle}P_{l_{2}}(\langle\xi,x\rangle)\ d\xi\right|\right)\right\},$$

where  $P(\xi)$  is a function of  $\xi$  with polynomial growth, namely,

$$P(\xi) = \begin{cases} \xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}|\xi|^{2M+n-L} & \text{if } n \text{ is even} \\ \xi_{j_1}\xi_{j_2}\cdots\xi_{j_L}|\xi|^{2M+n+1-L} & \text{if } n \text{ is odd} \end{cases}$$

An integration by parts, as used in the proof of Theorem 5.11, works as expected.  $\Box$ 

We can replace the support condition in Theorems 5.11 and 5.13 by the following condition, to show  $H^p \to L^q$  boundedness.

**Corollary 5.15.** Suppose  $0 , <math>0 \leq \beta < n$ . Let  $\widehat{\Phi}$  denote the Fourier transform of  $\Phi$  and

$$\frac{1}{p} = \frac{1}{q} + \frac{\beta}{n}.$$

For an integer  $M \ge n(\frac{1}{p}-1)$ , suppose  $\widehat{\Phi}$  is a function in  $C^{2M+n}(\mathbb{R}^n)$  and satisfies, for all I with  $|I| \le 2M + n$ ,

$$\partial^{I}(\widehat{\Phi})(\xi) |\xi|^{k} \in L^{1}(\mathbb{R}^{n}), \text{ for } k = 0, 1, 2, \dots |I|.$$

Assume also

$$\lim_{\xi \to \infty} \left| \partial^{I}(\widehat{\Phi})(\xi) \right| |\xi|^{M} = 0, \text{ for all } I \text{ satisfying } |I| \leq 2M + n$$

If  $M = n(\frac{1}{p} - 1)$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^{q,\infty}(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

If  $M > n(\frac{1}{p} - 1)$ , then

$$\left\|H_{\Phi,\beta}(f)\right\|_{L^q(\mathbb{R}^n)} \leqslant \left\|f\right\|_{H^p(\mathbb{R}^n)}.$$

*Proof.* The proof follows the same argument as the proof of Theorem 5.11. We prove the case  $M = n(\frac{1}{p} - 1)$ . To prove the first inequality in the corollary, by Lemma 5.8, we need to check

$$\left\|\frac{|x|^{n-\beta}}{|\cdot|^{n-\beta}}\Phi(\frac{x}{|\cdot|})\right\|_{\Lambda_{\alpha}} \leq |x|^{-\alpha},$$

where  $\alpha = M$ . Here, by Fourier transforms, we may write

$$\frac{|x|^{n-\beta}}{|y|^{n-\beta}}\Phi(\frac{x}{|\cdot|}) = |x|^{n-\beta} \int_{\mathbb{R}^n} \widehat{\Phi}(\xi) |y|^{-n+\beta} e^{i\frac{1}{|y|}\langle\xi,x\rangle} d\xi.$$

Thus, as in the proof of Theorem 5.11,

$$\sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left| \partial^I \left( |x|^{n-\beta} \int_{\mathbb{R}^n} \widehat{\Phi}(\xi) |y|^{-(n-\beta)} e^{i\frac{1}{|y|} \langle \xi, x \rangle} d\xi \right) \right| \right\}$$

$$\ll \sup_{y \in \mathbb{R}^n \setminus \{0\}} \left\{ \sum_{|I|=\alpha} \left(\frac{|x|}{|y|}\right)^{n-\beta} \sum_{K_1+K_2=I} \frac{1}{|y|^{|K_1|}} \left( \sum_{\substack{|K_2| \leqslant l_1 \leqslant 2|K_2| \\ l_1-l_2=|K_2|}} \frac{1}{|f_1|^{l_1}} \left| \int_{\mathbb{R}^n} \widehat{\Phi}(\xi) e^{i\frac{1}{|y|}\langle \xi, x \rangle} P_{l_2}(\langle \xi, x \rangle) \ d\xi \right| \right) \right\}.$$

The conditions in the theorem are sufficient to allow the integration by parts in Theorem 5.11 to proceed, which allows us to prove the first inequality. An argument similar to the one used to prove Lemma 5.12 shows that the condition is also sufficient to give the conclusion to that lemma. Thus, we obtain the second inequality by an interpolation argument.  $\hfill \Box$ 

# CHAPTER 6 ADDITIONAL SPACES AND OPERATORS

#### 6.1 MOTIVATION

In studying the  $H^p$  boundedness of the Hausdorff operator, we have so far obtained only results which require some smoothness condition on  $\Phi$ .

**Question 4.** If we don't assume any smoothness condition on  $\Phi$ , and only assume  $\Phi$  satisfies some size condition, can we obtain the  $H^p$  boundedness of  $\Phi$ ?

To gain insight into this question, let us review the argument showing that  $h_{\Phi}$  is bounded on  $H^1(\mathbb{R})$  when  $\Phi \in L^1$ . When p = 1,  $H^1$  is a normed space, and so we may use the Minkowski inequality.

Recall that for  $f \in H^p$ , 0 , we may write f as an atomic decomposition,

$$f = \sum \lambda_j a_j$$

with

$$\sum \left|\lambda_j\right|^p \cong \|f\|_{H^p}^p,$$

where each  $a_j(x)$  is a  $(p, \infty, s)$  atom, with s an integer such that  $s \ge \left[n\left(\frac{1}{p}-1\right)\right]$ . We study the  $H^p$  boundedness of the Hausdorff operator.

So, for  $f \in H^1$ , take an atomic decomposition  $f = \sum \lambda_j a_j$ . Using the Minkowski inequality, we have

$$\|h_{\Phi}(f)\|_{H^1} \leq \sum |\lambda_j| \|h_{\Phi}(a_j)\|_{H^1}.$$

where  $\sum \lambda_j a_j$  is an atomic decomposition of f. In the above inequality, by the Minkowski inequality again,

$$\|h_{\Phi}(a_j)\|_{H^1} = \left\|\int_0^\infty \frac{\Phi(t)}{t} a_j(\frac{x}{t}) dt\right\|_{H^1}$$

$$\leq \int_0^\infty \frac{|\Phi(t)|}{t} \left\| a_j(\frac{\cdot}{t}) \right\|_{H^1} dt$$
$$= \int_0^\infty |\Phi(t)| \left\| \frac{1}{t} a_j(\frac{\cdot}{t}) \right\|_{H^1} dt$$

Since, for fixed t > 0,  $\frac{1}{t}a_j(\frac{x}{t})$  is also a  $(1, \infty, s)$  atom, we have

$$\left\|\frac{1}{t}a_j(\frac{\cdot}{t})\right\|_{H^1} \leqslant 1$$

uniformly for t and j. Thus we obtain that

$$\|h_{\Phi}(f)\|_{H^1} \leq \sum |\lambda_j| \int_0^\infty |\Phi(t)| \ dt \simeq \|f\|_{H^1} \int_0^\infty |\Phi(t)| \ dt.$$

Therefore, we obtain that  $h_{\Phi}$  is bounded on  $H^1$  provided

$$\int_0^\infty |\Phi(t)| \ dt < \infty.$$

Now we try to extend the above argument to the  $H^p$  boundedness for 0 .First we have

$$\|h_{\Phi}(f)\|_{H^{p}}^{p} \leq \sum |\lambda_{j}|^{p} \|h_{\Phi}(a_{j})\|_{H^{p}}^{p}.$$
(6.1)

If we can show, by assuming certain size condition on  $\Phi$ , that

$$\|h_{\Phi}(a_j)\|_{H^p}^p \leqslant 1 \tag{6.2}$$

uniformly on j, then from equation (6.1) we obtain the  $H^p$  boundedness of  $h_{\Phi}$ . So, showing the  $H^p$  boundedness of  $h_{\Phi}$  is reduced to showing equation (6.2). Unfortunately, when p < 1,  $H^p$  is not a normed space, so we cannot apply the Minkowski inequality to obtain, for an atom a,

$$\|h_{\Phi}(a)\|_{H^p} \leq \int_0^\infty \frac{|\Phi(t)|}{t} \|a(\frac{\cdot}{t})\|_{H^p} dt$$

as we did for  $H^1$  norm. To overcome this difficulty, we write

$$h_{\Phi}(a)(x) = \int_0^\infty \frac{\Phi(t)}{t} a(\frac{x}{t}) \, dt = \sum_{k=-\infty}^\infty \int_{2^k}^{2^{k+1}} \frac{\Phi(t)}{t} a(\frac{x}{t}) \, dt,$$

$$\int_{2^k}^{2^{k+1}} \frac{\Phi(t)}{t} a(\frac{x}{t}) dt.$$

We introduce a normalizing factor, and define a  $(p, \infty, 0)$  atom  $A_k$  by

$$A_k(x) = \left(2^{\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \int_{2^k}^{2^{k+1}} \frac{\Phi(t)}{t} a(\frac{x}{t}) dt.$$

This is possible if supp a contains the origin. To show this fact, without loss of generality, we may assume the support of a is  $(-\rho, \rho)$ . As t runs over the interval  $(2^k, 2^{k+1})$ , we can view that the support of  $a(\frac{x}{t})$  is contained in the interval  $\{x : |x| \leq \rho 2^{k+1}\}$ . Thus  $A_k(x)$  is supported in  $(-\rho 2^{k+1}, \rho 2^{k+1})$ .

Next, by applying Fubini's theorem, we can show the cancellation condition, as follows. Fix  $u \leq s$ . Then

$$\int_{\mathbb{R}} x^{u} A_{k}(x) = \left(2^{\frac{1}{p}} \int_{2^{k}}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \int_{2^{k}}^{2^{k+1}} \frac{\Phi(t)}{t} \int_{\mathbb{R}} x^{u} a(\frac{x}{t}) dx dt$$
$$= \left(2^{\frac{1}{p}} \int_{2^{k}}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \int_{2^{k}}^{2^{k+1}} \Phi(t) t^{u} \int_{\mathbb{R}} x^{u} a(x) dx dt = 0.$$

Also, we have the size condition with radius  $\rho 2^{k+1}$ 

$$\begin{split} \|A_k\|_{L^{\infty}} &\leqslant \left(2^{\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \int_{2^k}^{2^{k+1}} \frac{|\Phi(t)|}{t} \|a\|_{L^{\infty}} dt \\ &\leqslant \rho^{-\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{|\Phi(t)|}{t} dt \left(2^{\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \\ &= 2^{\frac{1}{p}} (2^{k+1} \rho)^{-\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{2^{\frac{k}{p}} |\Phi(t)|}{t} dt \left(2^{\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{-1} \\ &\cong (2^{k+1} \rho)^{-\frac{1}{p}}. \end{split}$$

Thus, we have proved that each  $A_k$  is a  $(p, \infty, s)$  atom. We write

$$\int_0^\infty \frac{\Phi(t)}{t} a(\frac{x}{t}) \, dt = \sum_{k=-\infty}^\infty \left( 2^{\frac{1}{p}} \int_{2^k}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} \, dt \right) A_k(x).$$

Now, if we assume that for some  $\sigma > \frac{1-p}{p}$ ,

$$\int_{\mathbb{R}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} (1 + \left|\log_2 |t|\right|)^{\sigma} dt \leq C$$

then by an easy computation, we obtain

$$\|h_{\Phi}(a)\|_{H^{p}}^{p} \leqslant \sum_{k=-\infty}^{\infty} \left(2^{\frac{1}{p}} \int_{2^{k}}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{p} \|A_{k}\|_{H^{p}}^{p}$$
$$\leqslant \sum_{k=-\infty}^{\infty} \left(2^{\frac{1}{p}} \int_{2^{k}}^{2^{k+1}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} dt\right)^{p}$$
$$\leqslant \left(\int_{\mathbb{R}} \frac{t^{\frac{1}{p}} |\Phi(t)|}{t} (1 + |\log_{2}|t||)^{\sigma} dt\right)^{p} \leqslant 1.$$

Thus equation (6.2) is proved, if  $a_j$  is supported in an interval containing 0.

However, we will see that the above argument fails if the support of a does not contain the origin. If supp  $(a) \subset (\alpha, \beta)$ , and  $\beta > \alpha > 0$ , then

$$\int_{2^k}^{2^{k+1}} \frac{\Phi(t)}{t} a(\frac{x}{t}) dt$$

is supported in  $(2^k \alpha, 2^{k+1}\beta)$ . Now the length of  $(2^k \alpha, 2^{k+1}\beta)$  is

$$2^{k+1}\beta - 2^k\alpha = 2^{k+1}(\beta - \alpha) + 2^k\alpha.$$

This length can not be compared to  $2^{k+1}(\beta - \alpha)$ , since the number  $2^k \alpha$  may be arbitrarily large. Thus, we can not choose a normalizing factor  $F_k$  so that  $F_k \int_{2^k}^{2^{k+1}} \frac{\Phi(t)}{t} a(\frac{x}{t}) dt$ satisfies the size condition in general.

Based on these observations, we wish to restrict our discussion to only those atoms whose support includes the origin. In other words, we wish to consider a space whose elements f have the central atomic decompositions, that is, each atom  $a_j$  in a decomposition of f is supported in  $(-\rho_j, \rho_j)$ . Such spaces are called Herz-type Hardy spaces  $H\dot{K}_q^{\alpha,p}(\mathbb{R}^n)$ , introduced in section 2.5. 6.2 Boundedness of Hausdorff Operator on  $H\dot{K}^{\alpha,p}_q(\mathbb{R}^n)$ 

Recall from page 26 the definitions of the homogeneous Herz space and the Herz-type Hardy space.

**Definition 2.4** (Homogeneous Herz Space). Let  $\alpha \in \mathbb{R}$ ,  $0 < p, q < \infty$ . The homogeneous Herz space  $\dot{K}_q^{\alpha,p}(\mathbb{R}^n)$  is defined by

$$\dot{K}_{q}^{\alpha,p} = \left\{ f \in L_{\text{loc}}^{q}(\mathbb{R}^{n} \setminus \{0\}) : \|f\|_{\dot{K}_{q}^{\alpha,p}} < \infty \right\},\$$

where

$$\|f\|_{\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} = \left(\sum_{k=-\infty}^{\infty} 2^{k\alpha p} \|f\chi_{k}\|^{p}_{L^{q}(\mathbb{R}^{n})}\right)^{\frac{1}{p}}.$$

**Definition 2.5** (Herz-type Hardy Space). Let  $\alpha \in \mathbb{R}$ ,  $0 , <math>1 < q < \infty$ . The homogeneous Herz-type Hardy space  $H\dot{K}_q^{\alpha,p}(\mathbb{R}^n)$  is defined by

$$H\dot{K}_{q}^{\alpha,p}(\mathbb{R}^{n}) = \left\{ f \in S'(\mathbb{R}^{n}) : Gf \in \dot{K}_{q}^{\alpha,p}(\mathbb{R}^{n}) \right\},\$$

where Gf is the grand maximal function of f and

$$\|f\|_{H\dot{K}^{\alpha,p}_q(\mathbb{R}^n)} = \|Gf\|_{\dot{K}^{\alpha,p}_q(\mathbb{R}^n)}.$$

Recall also from page 26 that the Herz-type Hardy space can be decomposed into central atoms.

**Definition 2.6** (Central Atom). Suppose  $1 < q < \infty$ ,  $n(1 - \frac{1}{q}) \leq \alpha < \infty$ , and  $s \geq [\alpha + n(\frac{1}{q} - 1)]$ . A function a(x) on  $\mathbb{R}^n$  is said to be a central  $(\alpha, q)$  atom if

- (i) supp  $a \subset B(0,\rho)$ ,
- (*ii*)  $||a||_{L^q(\mathbb{R}^n)} \leq |B(0,\rho)|^{-\alpha/n}$ ,
- (iii)  $\int_{\mathbb{R}^n} a(x) x^{\beta} dx = 0$  for any multi-index  $\beta$  with  $|\beta| \leq s$ .

Following logic similar to that in Section 6.1, we have the following result for Herz-type Hardy spaces. **Theorem 6.1.** Let  $0 , and <math>n(1 - \frac{1}{q}) \le \alpha < \infty$ .

(1) For 0 , let $<math display="block">C_{p,\sigma} = \int_{\mathbb{R}^n} \frac{|\Phi(y)|}{|y|^n} |y|^\alpha |y|^{\frac{n}{q}} (1 + \log|y|)^\sigma \, dy.$ If, for some  $\sigma > \frac{1-p}{p}$ ,  $C_p := C_{p,\sigma} < \infty$ , then  $\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,p}(\mathbb{R}^n)} \leq \|f\|_{H\dot{K}^{\alpha,p}(\mathbb{R}^n)}.$ 

$$\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} \leq \|f\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})}.$$

(2) For p = 1, let

$$C_{1} = \int_{\mathbb{R}^{n}} \frac{|\Phi(y)|}{|y|^{n}} |y|^{\alpha} |y|^{\frac{n}{q}} dy.$$

If  $C_1 < \infty$ , then

$$\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,1}_q(\mathbb{R}^n)} \leqslant \|f\|_{H\dot{K}^{\alpha,1}_q(\mathbb{R}^n)}.$$

*Proof.* We prove part (1) only; the argument for part (2) is identical. For  $f \in H\dot{K}_q^{\alpha,p}$ , take a central atomic decomposition.

$$f = \sum_{k} \lambda_k a_k$$

where

$$\sum_{k \in \mathbb{Z}} |\lambda_k|^p \cong \|f\|_{H\dot{K}^{\alpha,p}_q}^p$$

and each  $a_k$  is an  $(\alpha, q)$  central atom supported in  $B(0, \rho_k)$ . Now we have

$$H_{\Phi}(f) = \sum_{k \in \mathbb{Z}} \lambda_k H_{\Phi}(a_k).$$

To prove the theorem, it suffices to show that

$$H_{\Phi}(a_k) = \sum_{j \in \mathbb{Z}} c_{k,j} a_{k,j},$$

$$\sum_{j\in\mathbb{Z}} |c_{k,j}|^p \leqslant 1$$

uniformly on  $k \in \mathbb{Z}$ .

where each  $a_{k,j}$  again is a central  $(\alpha, q)$  atom and

We write

$$b_{k,j}(x) = \int_{2^{-j} \le |y| \le 2^{-j+1}} \frac{\Phi(y)}{|y|^n} a_k\left(\frac{x}{|y|}\right) dy, \quad j \in \mathbb{Z}.$$

So

$$H_{\Phi}(a_k)(x) = \sum_{j \in \mathbb{Z}} b_{k,j}(x).$$

It is easy to check that each  $b_{k,j}$  satisfies the same cancellation condition as  $a_k$ . By the Minkowski inequality, the size of  $b_{k,j}$  is

$$\begin{aligned} \|b_{k,j}\|_{L^{q}} &\leq \int_{2^{-j} \leq |y| \leq 2^{-j+1}} \frac{|\Phi(y)|}{|y|^{n}} \left\|a_{k}\left(\frac{\cdot}{|y|}\right)\right\|_{L^{q}} dy \\ &\leq \rho^{-\alpha} \int_{2^{-j} \leq |y| \leq 2^{-j+1}} \frac{|\Phi(y)|}{|y|^{n}} |y|^{\frac{n}{q}} dy. \end{aligned}$$

If  $|x| > 2^{-j+1}\rho_k$  then

$$\frac{|x|}{|y|} \ge 2^{j-1} |x| > \rho_k,$$

which means  $a_k(\frac{x}{|y|}) = 0$  for all  $2^{-j} \leq |y| \leq 2^{-j+1}$ . This tells us that

$$\operatorname{supp}(b_{k,j}) \subset B(0, 2^{-j+1}\rho_k).$$

Now we write

$$H_{\Phi}(a_k)(x) = \sum_{j \in \mathbb{Z}} c_{k,j} a_{k,j}$$

with

$$c_{k,j} = 2^{(-j+1)\alpha} \int_{2^{-j} \le |y| \le 2^{-j+1}} \frac{|\Phi(y)|}{|y|^n} |y|^{\frac{n}{q}} dy,$$

and

$$a_{k,j} = \frac{1}{c_{k,j}} b_{k,j}.$$

It is easy to check that  $a_{k,j}$  is a central  $(\alpha, q)$  atom and

$$\sum_{j\in\mathbb{Z}} |c_{k,j}|^p \leq \sum_{j\in\mathbb{Z}} \left( \int_{2^{-j} \leq |y| \leq 2^{-j+1}} \frac{|\Phi(y)|}{|y|^n} |y|^{\alpha + \frac{n}{q}} dy \right)^p$$
$$\leq \left( \int_{\mathbb{R}^n} \frac{|\Phi(y)|}{|y|^n} |y|^{\alpha + \frac{n}{q}} (1 + |\log|y||)^\sigma dy \right)^p = C_p.$$

This shows

$$H_{\Phi}(f) = \sum_{k \in \mathbb{Z}} \lambda_k H_{\Phi}(a_k)$$
$$= \sum_{k \in \mathbb{Z}} \sum_{j \in \mathbb{Z}} \lambda_k c_{k,j} a_{k,j}.$$

By the atomic decomposition, we obtain

$$\|H_{\Phi}(f)\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} \leq \left(\sum_{k\in\mathbb{Z}}\sum_{j\in\mathbb{Z}}|\lambda_{k}c_{k,j}|^{p}\right)^{\frac{1}{p}}$$
$$\leq \left(\sum_{k\in\mathbb{Z}}|\lambda_{k}|^{p}\right)^{\frac{1}{p}}$$
$$\leq \|f\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})},$$

as desired.

We now return to the example of the Cesàro operator  $\mathcal{C}_{\beta}$ .

**Corollary 6.2.** Let  $0 , and <math>n(1 - \frac{1}{q}) \le \alpha < \infty$ . For any  $\beta > 0$ , we have

$$\left\|\mathcal{C}_{\beta}(f)\right\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})} \leqslant \left\|f\right\|_{H\dot{K}^{\alpha,p}_{q}(\mathbb{R}^{n})}.$$

*Proof.* Note  $C_{\beta} = H_{\Phi}$ , where  $\Phi = \chi_{|y|<1}(y)(1-|y|)^{\beta-1}$ . It is easy to check that this  $\Phi$  meets the conditions of Theorem 6.1.

# 6.3 $H\dot{K}^{\alpha,p}_q(\mathbb{R}) \to H^r(\mathbb{R})$ Boundedness

In this section, we study the  $H\dot{K}^{\alpha,p}_q(\mathbb{R}) \to H^{r,\infty}(\mathbb{R})$  boundedness for the one dimensional modified Hausdorff operator

$$\widetilde{h}_{\Phi}(f)(x) = \int_{\mathbb{R}} \frac{\Phi(t)}{t} f(\frac{x}{t}) dt.$$

Note that here  $\Phi : \mathbb{R} \to \mathbb{R}$  is a locally integrable function. As before, we denote

$$\phi(v) = \begin{cases} \frac{\Phi(\frac{1}{v})}{v} & \text{if } v \neq 0\\ \lim_{v \to 0} \frac{\Phi(\frac{1}{v})}{v} & \text{if } v = 0. \end{cases}$$

**Theorem 6.3.** Let  $0 , and <math>\left(1 - \frac{1}{q}\right) \le \alpha < \infty$ . For

$$r = \frac{1}{\alpha + \frac{1}{q}}, \quad N = \left[\alpha + \frac{1}{q} - 1\right] = \left[\frac{1}{r} - 1\right], \quad \frac{1}{q'} + \frac{1}{q} = 1$$

Let

$$\gamma = \frac{1}{r} - 1 - N = \frac{1}{r} - 1 - \left[\frac{1}{r} - 1\right].$$

If  $\phi \in C^N$  and

$$\int_{-\rho}^{\rho} \left| \phi^{(N)}(t) - \phi^{(N)}(0) \right|^{q'} dt \leq \rho^{1 + \gamma q'}$$

uniformly for  $\rho > 0$ , then we have

$$\left\|\widetilde{h}_{\Phi}(f)\right\|_{H^{r,\infty}} \leqslant \|f\|_{H\dot{K}^{\alpha,p}_{q}}.$$

*Proof.* For any  $f \in H\dot{K}_q^{\alpha,p}$ , we write

$$f = \sum_{k} \lambda_k a_k$$

where

$$\sum_{k \in \mathbb{Z}} |\lambda_k|^p \cong \|f\|_{H\dot{K}^{\alpha,p}_q}^p,$$

$$\widetilde{h}_{\Phi}(f) = \sum_{k \in \mathbb{Z}} \lambda_k \widetilde{h}_{\Phi}(a_k).$$

and each  $a_k$  is an  $(\alpha, q)$  central atom supported in the ball  $B(0, \rho_k)$ . Now we have

Then as in the proof of Theorem 3.1, for the Hilbert transform  $\mathcal{R}$ , we can write

$$\mathcal{R} \circ \widetilde{h}_{\Phi}(f) = \sum_{k \in \mathbb{Z}} \lambda_k (\mathcal{R} \circ \widetilde{h}_{\Phi})(a_k)$$

and

$$(\mathcal{R} \circ \tilde{h}_{\Phi})(a_k)(x) = \int_{-\infty}^{\infty} \frac{\Phi(t)}{t} (\mathcal{R}a_k)(\frac{x}{t}) dt$$

Changing variables  $\frac{1}{t} = v$ , by the definition of  $\phi$ , we have

$$\begin{aligned} \left| \mathcal{R} \circ \widetilde{h}_{\Phi}(a_k)(x) \right| &= \left| \int_{-\infty}^{\infty} \frac{\Phi(\frac{1}{v})}{v} (\mathcal{R}a_k)(xv) \, dv \right| \\ &= \left| \int_{-\infty}^{\infty} \phi(v) (\mathcal{R}a_k)(xv) \, dv \right|. \end{aligned}$$

It is known [20] that the Hilbert transform is bounded on the space  $H\dot{K}_q^{\alpha,p}$ . That is,

$$\|(\mathcal{R}a_k)\|_{H\dot{K}_q^{\alpha,p}} \leq \|a_k\|_{H\dot{K}_q^{\alpha,p}} \leq 1$$

uniformly for all atoms  $a_k$ . Thus  $\mathcal{R}a_k$  is an element in  $H\dot{K}_q^{\alpha,p}$ , so we may write

$$(\mathcal{R}a_k) = \sum_j c_{k,j} a_{k,j},$$

where each  $a_{k,j}$  is again a central  $(\alpha, q)$  atom and

$$\sum_{j} \left| c_{k,j} \right|^p \leqslant 1$$

uniformly on k. This shows that

$$\left|\mathcal{R} \circ \widetilde{h}_{\Phi}(a_k)(x)\right| = \left|\sum_{j} c_{k,j} \int_{-\infty}^{\infty} \phi(v) a_{k,j}(xv) dv\right|.$$

We further note that for each fixed  $x \neq 0$ ,

$$b_{k,j}(v) = |x|^{\frac{1}{r}} a_{k,j}(xv)$$

is again an  $(\alpha, q)$  central atom. Thus,

$$\left|\mathcal{R} \circ \widetilde{h}_{\Phi}(a_k)(x)\right| = |x|^{-\frac{1}{r}} \left|\sum_{j} c_{k,j} \int_{-\infty}^{\infty} \phi(v) b_{k,j}(v) \ dv\right|.$$

For simplicity of notation, we write

$$b(v) = b_{k,j}(v),$$

and assume the support of b is  $(-\rho,\rho).$  Thus,

$$\int_{-\infty}^{\infty} \phi(v) b_{k,j}(v) \ dv = \int_{-\rho}^{\rho} \phi(v) b(v) \ dv.$$

First we assume  $N \ge 1$ . Using the Taylor expansion of  $\phi$ , we write

$$\phi(v) = \sum_{k=0}^{N-1} \frac{1}{k!} \phi^{(k)}(0) v^k + \frac{1}{(N-1)!} v^N \int_0^1 (1-s)^{N-1} \phi^{(N)}(sv) \, ds.$$

By the cancellation condition on b, we now have

$$\begin{split} \int_{-\infty}^{\infty} \phi(v)b(v) \, dv &= \frac{1}{(N-1)!} \int_{-\rho}^{\rho} v^N \left( \int_0^1 (1-s)^{N-1} \phi^{(N)}(sv) \, ds \right) b(v) \, dv \\ &\cong \int_{-\rho}^{\rho} \left\{ \int_0^1 (1-s)^{N-1} [\phi^{(N)}(sv) - \phi^{(N)}(0)] \, ds \right\} v^N b(v) \, dv \\ &= \int_0^1 (1-s)^{N-1} \int_{-\rho}^{\rho} [\phi^{(N)}(sv) - \phi^{(N)}(0)] v^N b(v) \, dv \, ds. \end{split}$$

Thus

$$\begin{split} \left| \int_{-\infty}^{\infty} \phi(v) b(v) \ dv \right| &\leqslant \rho^{N} \int_{0}^{1} (1-s)^{N-1} \int_{-\rho}^{\rho} \left| \phi^{(N)}(sv) - \phi^{(N)}(0) \right| \left| b(v) \right| \ dv \ ds \\ &\leqslant \rho^{N} \left\| b \right\|_{L^{q}} \int_{0}^{1} (1-s)^{N-1} \left( \int_{-\rho}^{\rho} \left| \phi^{(N)}(sv) - \phi^{(N)}(0) \right|^{q'} \ dv \right)^{\frac{1}{q'}} \ ds \\ &\leqslant \rho^{N} \left\| b \right\|_{L^{q}} \int_{0}^{1} (1-s)^{N-1} s^{-\frac{1}{q'}} \left( \int_{-s\rho}^{s\rho} \left| \phi^{(N)}(v) - \phi^{(N)}(0) \right|^{q'} \ dv \right)^{\frac{1}{q'}} \ ds \\ &\leqslant \rho^{N-\alpha} \rho^{\gamma+\frac{1}{q'}} \\ &= \rho^{\frac{1}{r}-1+\frac{1}{q'}-\alpha} \\ &= \rho^{\frac{1}{r}-\frac{1}{q}-\alpha} = 1 \end{split}$$

uniformly for all atoms b. On the other hand, if N = 0,

$$\begin{split} \left| \int_{-\infty}^{\infty} \phi(v) b(v) \, dv \right| &= \left| \int_{-\rho}^{\rho} (\phi(v) - \phi(0)) b(v) \, dv \right| \\ &\leq \left( \int_{-\rho}^{\rho} |\phi(v) - \phi(0)| \, dv \right)^{\frac{1}{q'}} \|b\|_{L^q} \leqslant 1 \end{split}$$

uniformly for all atoms b.

This indicates

$$\left|\mathcal{R} \circ \widetilde{h}_{\Phi}(a_k)(x)\right| = |x|^{-\frac{1}{r}} \left|\sum_{j} c_{k,j} \int_{-\infty}^{\infty} \phi(v) b_{k,j}(v) \ dv\right| \leq |x|^{-\frac{1}{r}}$$

uniformly on central atoms  $a_k$ .

For any  $\lambda > 0$ , now we have

$$\begin{split} \left| \left\{ x \neq 0 : \left| \mathcal{R} \circ \tilde{h}_{\Phi}(a_k)(x) \right| > \lambda \right\} \right| \\ &\leq \left| \left\{ x \neq 0 : |x|^{-\frac{1}{r}} \geq \lambda \right\} \right| \\ &= \left| \left\{ x \neq 0 : |x|^{\frac{1}{r}} \leq \frac{1}{\lambda} \right\} \right| \\ &\cong \left( \frac{1}{\lambda} \right)^r. \end{split}$$

We can show by a similar method

$$\left|\left\{x: \left|\widetilde{h}_{\Phi}(a_k)(x)\right| > \lambda\right\}\right| \leq \left(\frac{1}{\lambda}\right)^r.$$

This gives

$$\begin{split} \left\| \widetilde{h}_{\Phi}(f) \right\|_{H^{r,\infty}} &\cong \left\| \mathcal{R} \circ \widetilde{h}_{\Phi}(f) \right\|_{L^{r,\infty}} + \left\| \widetilde{h}_{\Phi}(f) \right\|_{L^{r,\infty}} \\ &\leqslant \left( \sum_{k} |\lambda_{k}|^{p} \right)^{\frac{1}{p}} \leqslant \| f \|_{H\dot{K}^{\alpha,p}_{q}} \end{split}$$

as desired.

Remark 6.4. An analogous strong boundedness result can be shown by interpolating on r and q.

### 6.4 BILINEAR HAUSDORFF OPERATORS

Next we study the bilinear Hausdorff operator

$$H_{\Phi,m,k}(f,g)(x) = \int_{-\infty}^{\infty} \frac{\Phi(t)}{t} f(\frac{x^m}{t}) g(\frac{x^k}{t}) dt.$$

Note this is defined for all  $x \in \mathbb{R} \setminus \{0\}$  when  $m, k \in \mathbb{Z}$ .

**Theorem 6.5.** Let  $m, k = 1, 2, \ldots$  For any  $p, p_1, p_2, r, p' \ge 1$  satisfying

$$\frac{1}{p_1} + \frac{1}{p_2} = \frac{1}{p}, \qquad \frac{1}{r} = \frac{m}{p_1} + \frac{k}{p_2}, \qquad \frac{1}{p} + \frac{1}{p'} = 1,$$

if

$$\left(\int_{-\infty}^{\infty} \left|\frac{\Phi\left(\frac{1}{t}\right)}{t}\right|^{p'} dt\right)^{\frac{1}{p'}} < \infty$$

then

$$\|H_{\Phi,m,k}(f,g)\|_{L^{r,\infty}} \leq \|f\|_{L^{p_1}} \|g\|_{L^{p_2}}$$

*Proof.* By Hölder's inequality and a scaling argument, for x > 0

$$\begin{aligned} |H_{\Phi,m,k}(f,g)(x)| &= \left| \int_{-\infty}^{\infty} \frac{\Phi(\frac{1}{t})}{t} f(tx^m) g(tx^k) \, dt \right| \\ &\leqslant \left( \int_{-\infty}^{\infty} \left| \frac{\Phi(\frac{1}{t})}{t} \right|^{p'} \, dt \right)^{\frac{1}{p'}} \left\| f(\cdot x^m) g(\cdot x^k) \right\|_{L^p} \\ &\leqslant \left( \int_{-\infty}^{\infty} \left| \frac{\Phi(\frac{1}{t})}{t} \right|^{p'} \, dt \right)^{\frac{1}{p'}} \left\| f(\cdot x^m) \right\|_{L^{p_1}} \left\| g(\cdot x^k) \right\|_{L^{p_2}} \\ &= |x|^{-\frac{m}{p_1} - \frac{k}{p_2}} \left( \int_{-\infty}^{\infty} \left| \frac{\Phi(\frac{1}{t})}{t} \right|^{p'} \, dt \right)^{\frac{1}{p'}} \| f\|_{L^{p_1}} \| g\|_{L^{p_2}} \\ &= C_{\Phi} \left| x \right|^{-\frac{m}{p_1} - \frac{k}{p_2}} \| f\|_{L^{p_1}} \| g\|_{L^{p_2}} \,, \end{aligned}$$

where

$$C_{\Phi} = \left( \int_{-\infty}^{\infty} \left| \frac{\Phi(\frac{1}{t})}{t} \right|^{p'} dt \right)^{\frac{1}{p'}}.$$

$$\begin{split} &|\{x: |H_{\Phi}(f,g)(x)| > \lambda\}| \\ &\leq \left| \left\{ x: C_{\Phi} |x|^{-\frac{m}{p_{1}} - \frac{k}{p_{2}}} \|f\|_{L^{p_{1}}} \|g\|_{L^{p_{2}}} > \lambda \right\} \right| \\ &\cong \left| \left\{ x: |x|^{\frac{m}{p_{1}} + \frac{k}{p_{2}}} \leqslant \frac{\|f\|_{L^{p_{1}}} \|g\|_{L^{p_{2}}}}{\lambda} \right\} \right| \\ &\cong \left( \frac{\|f\|_{L^{p_{1}}} \|g\|_{L^{p_{2}}}}{\lambda} \right)^{\frac{1}{p_{1}} + \frac{k}{p_{2}}}, \end{split}$$

where by assumption  $\frac{m}{p_1} + \frac{k}{p_2} < 1$ .

Recalling  $r = \frac{1}{\frac{m}{p_1} + \frac{k}{p_2}} \ge 1$ , we have

$$\begin{split} & |\{x: |H_{\Phi}(f,g)(x)|^{r} > \lambda\}| \\ &= \left| \left\{ x: |H_{\Phi}(f,g)(x)| > \lambda^{\frac{1}{r}} \right\} \right| \\ & \leq \left( \|f\|_{L^{p_{1}}} \|g\|_{L^{p_{2}}} \right)^{\frac{1}{\frac{m}{p_{1}} + \frac{k}{p_{2}}}} \lambda^{-1} \end{split}$$

This gives

$$\|H_{\Phi}(f,g)\|_{L^{r,\infty}} \leq \|f\|_{L^{p_1}} \|g\|_{L^{p_2}}$$

as desired.

If we restrict the domains of  $f, g, \Phi$  to  $[0, \infty)$ , we can extend the definition of the bilinear Hausdorff operator as: for  $f, g \in S([0, \infty)), \alpha, \beta \in \mathbb{R}$ ,

$$H_{\Phi,\alpha,\beta}(f,g)(x) = \int_0^\infty \frac{\Phi(t)}{t} f\left(\frac{x^\alpha}{t}\right) g\left(\frac{x^\beta}{t}\right) dt,$$

in which case, the method of proof used for Theorem 6.5 can also be used to show the analogous result.

#### BIBLIOGRAPHY

- [1] CHEN, J., FAN, D., AND LI, J. Hausdorff operators on function spaces. *Chinese* Annals of Mathematics-Series B 33, 4 (2012), 537–556.
- [2] CHRIST, M., AND GRAFAKOS, L. The best constants for two non-convolution inequalities. Proc. Amer. Math. Soc., 123 (1995), 1687 – 1693.
- [3] COIFMAN, R. A real-variable characterization of H<sup>p</sup>. Studia Math., 51 (1974), 269–274.
- [4] COIFMAN, R., AND WEISS, G. Extensions of Hardy spaces and their use in analysis. Bull. Amer. Math. Soc., 83 (1977), 569–645.
- [5] GARCÍA-CUERVA, J., AND RUBIO DE FRANCIA, J. Weighted norm inequalities and related topics. North Holland, 1985.
- [6] GOLDBERG, D. A local version of real Hardy spaces. Duke Mathematical Journal 46, 1 (1979), 27–42.
- [7] HARDY, G. H., LITTLEWOOD, J. E., AND POLYA, G. Inequalities. Cambridge University Press, 1959.
- [8] HOLMSTEDT, T. Interpolation of quasi-normed spaces. Math. Scand. 26 (1970), 177–199.
- [9] KANJIN, Y. The Hausdorff operators on the real Hardy spaces  $H^1(\mathbb{R})$ . Studia Math., 148 (2001), 37–45.
- [10] LERNER, A., AND LIFLYAND, E. Multidimensional Hausdorff operators on the real Hardy spaces. *Jour. Austr. Math. Soc.*, 83 (2007), 79–86.
- [11] LIFLYAND, E. Hausdorff operators. http://www.dmsa.unipd.it/res/cong/ mini2008/ProfLiflyand08lesson.pdf, 2000.
- [12] LIFLYAND, E. Boundedness of multidimensional Hausdorff operators on  $H^1(\mathbb{R}^n)$ . Acta Sci. Math. (Szeged), 74 (2008), 845–851.
- [13] LIFLYAND, E. Complex and real Hausdorff operators. http://ddd.uab.cat/ pub/prepub/2011/hdl\_2072\_182564/Pr1046.pdf, 2011.
- [14] LIFLYAND, E., AND MIYACHI, A. Boundedness of Hausdoff operators in  $H^p$  spaces, 0 . Studia Math., 194 (2009), 279–292.
- [15] LIFLYAND, E., AND MÓRICZ, F. The Hausdorff operator is bounded on the real Hardy space  $H^1(\mathbb{R})$ . Proc. Amer. Math. Soc., 128 (2000), 1391 1396.
- [16] LIFLYAND, E., AND MÓRICZ, F. The multi-parameter Hausdorff operator is bounded on the product Hardy space  $H^{11}(\mathbb{R} \times \mathbb{R})$ . Analysis, 21 (2001), 107–118.

- [17] LIFLYAND, E., AND MÓRICZ, F. Commuting relations for Hausdorff operators and Hilbert transforms on real Hardy spaces. Acta Mathematica Hungarica 97, 1 (2002), 133–143.
- [18] LIFLYAND, E., AND MÓRICZ, F. The Hausdorff operator is bounded on the real Hardy spaces. *Jour. Austr. Math. Soc.*, 83 (2007), 79–86.
- [19] LIN, X., AND SUN, L. Some estimates on Hausdorff operator. Acta Sci. Math. (Szeged) (accepted for publication).
- [20] LU, S., YANG, D., AND HU, G. *Herz type spaces and their applications*. Science Press, 2008.
- [21] LU, S. Z. Four lectures on real H<sup>p</sup> spaces. World Scientific Publishing Co. Inc., River Edge, NJ, 1995.
- [22] MIYACHI, A. Weak factorization of distributions in H<sup>p</sup> spaces. Pacific J. Math 115 (1984), 165–175.
- [23] MIYACHI, A. Boundedness of the Cesáro operator in Hardy spaces. Journal of Fourier Analysis and Applications 10, 1 (2004), 83–92.
- [24] MÓRICZ, F. Multivariate Hausdorff operators on the spaces  $H^1(\mathbb{R}^n)$  and BMO( $\mathbb{R}^n$ ). Analysis Mathematica 31, 1 (2005), 31–41.
- [25] STEIN, E. Singular integrals and differentiability properties of functions. Princeton University Press, 30 (1970).
- [26] STEIN, E., TAIBLESON, M., AND WEISS, G. Weak type estimates for maximal operators on certain H<sup>p</sup> classes. *Rend. Circ. Mat. Palermo, Suppl 1*, 2 (1981), 81–97.
- [27] STEIN, E., AND WEISS, G. Introduction to Fourier Analysis on Euclidean Spaces (PMS-32), vol. 32. Princeton University Press, 1971.
- [28] TAIBLESON, M., AND WEISS, G. The molecular characterization of certain Hardy spaces. Asterisque., 77 (1980), 67–151.
- [29] TRIEBEL, H. Interpolation theory, function spaces, differential operators, second ed. Johann Ambrosius Barth, Heidelberg, 1995.
- [30] WEISZ, F. The boundedness of the Hausdorff operator on multi-dimensional Hardy spaces. *Analysis.*, 24 (2004), 183–195.
- [31] ZYGMUND, A. Trigonometric series. Bull. Amer. Math. Soc 66 (1960), 6–12.

# XIAOYING LIN

### CURRICULUM VITAE

### Personal Data

Place of Birth	Xiamen, China
Nationality	Chinese Citizen, US Permanent Resident
Language	Native Chinese speaker, Fluent in English

### Education

- 2013 PhD in Mathematics from the University of Wisconsin—Milwaukee Thesis: Boundedness of Hausdorff Operator in Various Function Spaces Advisor: Dashan Fan
- 2009 MS in Mathematics from the University of Wisconsin—Milwaukee
- 2007 BS in Mathematics Education from the University of Wisconsin-Milwaukee
- 2005 AA in Math Teacher Education from Milwaukee Area Technical College
- **1995 AAS in Financial Accounting of Commercial Enterprise** from Beijing Social Correspondence University of China
- **1992** AAS in Sugar Refinery Technology from Jimei Light Industry School (China)

### Employment

2010 - 2013	Graduate Teaching Assistant in the Mathematical Sciences Depart-		
	ment at UW—Milwaukee		
2009 - 2010	Adjunct Faculty in the Mathematical Science Department at Mil-		
	waukee Area Technical College		
2009 - 2010	High School Teacher at Milwaukee Montessori IB High School		
2007 - 2009	Graduate Teaching Assistant in the Mathematical Sciences Depart-		
	ment at UW—Milwaukee		
2007	Math Student Teacher at Riverside University High School, Mil-		
	waukee		
2006	Math Student Teacher at I.D.E.A.L Charter School, Milwaukee		
1997 - 2001	Marketing Sales Representative, Futis Oversea Limited, Xiamen,		
	China		

1992-1997 – Assistant Engineer, Xiamen Sugar Factory, Xiamen, China

## **Teaching Experience**

### 2007 - 2013: Teaching Fellow

UWM Mathematical Sciences Department. Had various levels of responsibility for a variety of college level courses.

Fall 2012 Complex Analysis—Grader. Graduate-level course in Complex Analysis.

- Fall 2011 Advanced Calculus—Grader. Upper-level undergraduate course, which treats calculus from an abstract math viewpoint. This is a first proof-writing oriented course for many students. Grading involves evaluation of student writing as well as computation.
- Fall 2010 Calculus and Analytic Geometry I—Primary instructor. Limits, derivatives, and graphs of algebraic, trigonometric, exponential, and logarithmic functions; antiderivatives, the definite integral, and the fundamental theorem of calculus, with applications. Responsible for setting syllabus to prepare students to reach university benchmarks to proceed to Calculus II.
- Fall 2008 Survey in Calculus and Analytic Geometry—Discussion Leader. This one semester survey course covers applications for business administration, economics, and health sciences. Topics include coordinate systems, equations of curves, limits, differentiation in one and several variables, basic techniques of integration, and many applications.
- **Fall 2007** Discrete Probability and Statistics for Elementary Education Majors (Topics for K 8 teachers)—Classroom Assistant. This course covers basic concepts in probability and statistics, with a focus on hands-on activities and the understanding of mathematics as fundamentally comprehensible study. Taught in cooperation with a professor and a visiting teacher from Milwaukee Public Schools. Responsible for grading, assistance with classroom activities, and occasional lecture.

#### 2009 – 2010: Part Time Faculty

Milwaukee Area Technical College Math Department Had full responsibility for a variety of math courses.

- Summer 2010 Calculus and Analytic Geometry I—Primary Instructor. This is an intensive eight-week summer course. Students come from various other colleges to take summer credits for transfer to a primary institution. Course introduces the basic properties of limits, rate of change of functions, continuity, derivatives of algebraic, trigonometric and elementary and transcendental functions and their applications of derivatives, the indefinite integral and its applications including areas, derivatives and integrals involving logarithmic exponential, inverse trigonometric and hyperbolic functions, curve sketching, finding maxima and minima. Responsible for ensuring consistency of material covered with what is listed in transfer materials, in order to provide a comfortable transfer experience.
- **Spring 2010** Calculus and Analytic Geometry III—Primary Instructor. Taught in multiple locations via simultaneous video feed as part of school's longdistance learning program. Communication with students was through email, and written work was submitted and returned by fax. The challenges of this teaching style included communicating clearly about mathematics in plain-text emails, monitoring understanding of remote students, ensuring proper organization for materials that needed to be sent to several

campuses, and scheduling test proctors for remote locations. This course is targeted toward applied science students. Full responsibility for designing syllabus and selecting course topics. Topics include vectors, geometry of space, vector valued functions, partial derivatives, multiple integrals, and vector analysis.

Fall 2009 College Mathematics—Primary Instructor. A first course in college level algebra. Taught in a technical college setting to students from diverse backgrounds, including many returning students and students with work and family responsibilities. Many students come to this class with very limited mathematical knowledge. One central idea this course tries to teach is learning math through understanding rather than rote memorization of procedures. Topics covered include solving polynomial equations, polynomial arithmetic, introduce exponentials and logarithms.

#### 2009 – 2010: High School Teacher

Montessori IB High School, Milwaukee

Followed International Baccalaureate (IB) program guidelines to teach mathematics to high school juniors and seniors. Students came from low-income families in a majority-minority area. The school's goal was to help these students be well-prepared for college in mathematics and other areas. Working with the challenges provided by limited school resources, I worked to provide individualized instruction in a classroom setting. Material included polynomials, working with functions, inequalities, sequences and patterns, and basic trigonometry.

#### Spring 2007: Math Student Teacher

Riverside University High School, Milwaukee

Worked with senior teacher to develop hands-on lessons for a diverse population of high school students. This school's focus was on fostering student creativity and investigation, rather than rote training of individual mathematical techniques.

#### Fall 2006: Math Student Teacher

I.D.E.A.L. Charter School, Milwaukee

Worked with senior teacher to develop lessons for a diverse population of middle school students. A major focus at this level was to emphasize relevance of material to the world beyond the classroom. Worked to involve students' parents in their learning. Focused on hands-on activities and investigation.

#### 2004 – 2007: Teacher of Chinese Language and Culture

Cricket Academy, Wauwatosa

Saturday enrichment class for children aged 5-7, covering reading, writing and speaking of basic Chinese, as well as Chinese culture.

## Training and Professional Development

- 2010 2012 Progress toward Wisconsin Community/Technical College Teaching Certification, Milwaukee Area Technical College, Milwaukee, WI
  - **2010** IB (Internal Baccalaureate) Program Teacher Training, Salt Lake City, UT

## **Publications and Preprints**

L. Sun and X. Lin *Some estimates on Hausdorff Operator*, Acta Sci.Math.(Szeged), accepted for publication May 2012.

## Honors and Distinctions

- **2007 2012** Recipient of *Graduate Assistantship in Areas of National Need Fellowship.* This fellowship is awarded to students with excellent records who demonstrate financial need and plan to pursue the highest degree available in their course of study in a field designated as an area of national need.
- **2010 2012** Recipient of Advanced Opportunity Program Fellowship. This fellowship is awarded to graduate students who are members of groups underrepresented in graduate study.
  - **2007** Recipient of *Greater Milwaukee Foundation's Alice C. Helland Scholarship*, an award to female students planning to re-enter the workforce, enrolled in a science, math, engineering, or computer science curriculum.

## Service

- Spring 2005 Volunteer Teacher Assistant, Grand View High School, Milwaukee, WI Assisted in math classroom in alternative school for at-risk students. Conveyed my own learning experiences to students to encourage success.
- Fall 2006 Volunteer Math Tutor, University of Wisconsin—Milwaukee
- 2003 2004 Volunteer Math Tutor, Milwaukee Area Technical College Provided individual instruction to college students in algebra and general math. Introduced the use of graphing calculators. Focused on building self confidence and increasing motivation. Provided encouragement with immediate praise and verbal reinforcement.

## Licenses and Certifications

Teaching Certification, Wisconsin Early Adolescence to Adolescence (ages  $10\,-\,21)$  in Mathematics

# Memberships

National Council of Teachers of Mathematics Wisconsin Mathematics Council American Mathematical Society